

Development and application of holistic injury prevention strategies in pre-elite sport: Implementation in Australian netball

Erin Amanda Smyth

BPhty GradDip (Ext ScopePhty) MPhty (Sport)

Thesis submitted in fulfilment of the requirements for the award of

the degree of:

Doctor of Philosophy (Health)

Research Institute for Sport and Exercise

University of Canberra

Bruce, ACT, Australia

For Rachel Mary & Claudia Anne

Abstract

This thesis by publication is entitled 'Development and application of holistic injury prevention strategies in pre-elite sport: Implementation in Australian netball'. Pre-elite athletes sustain high rates of injury at a critical point in their careers. Translating Research into Injury Prevention Practice (TRIPP) is a well-established framework for injury prevention. It consists of six stages and has provided the foundation for this investigation into injury prevention in pre-elite athletes as it has been promoted as an approach that leads to quantifiable real-world injury prevention outcomes. To assess the usefulness of this framework in pre-elite athletes, due to the time constraints of a PhD, netball was used as an exemplar sport.

This thesis comprises five studies, of which four have been published, with one currently under review and being considered for publication. The first study applies TRIPP Stages 1 and 2 by conducting an injury surveillance study and establishing the aetiology and mechanisms of injury at the 2018 17/U & 19/U Australian National Netball Championships (ANNC). This study found an injury incidence rate of 89.4 injuries per 1000 player hours, with ankle sprains being the highest medical attention and sports incapacity injury. Foot blisters and lower back pain were also frequent medical attention injuries, while anterior cruciate ligament (ACL) rupture and sport-related concussion were the next most frequent sports incapacity injuries.

TRIPP Stage 3 is focused on developing injury prevention strategies and the first step is to obtain research evidence. Therefore, our second study was a systematic review investigating injury prevention strategies specific to pre-elite athletes competing in Olympic and professional sports. Eleven studies were identified demonstrating limited evidence supporting exercise and psychology interventions and no evidence supporting the use of equipment or nutrition strategies for preventing injury. This work established that there is a need for high-quality methodologically rigorous research studies to evaluate the efficacy of injury prevention interventions for pre-elite athletes.

The third study, demonstrating TRIPP Stage 4, consists of a randomised controlled trial to assess the effect of ankle tape and taping procedures on proprioception. Netball Australia has an ankle taping

policy whereby pre-elite netball athletes are encouraged to prophylactically tape their ankles, but there is a lack of scientific evidence supporting this policy. This study demonstrated that proprioception improves and is maintained during a netball session with either Sport and Exercise Physiotherapist (SEP) applied or self-applied ankle taping.

TRIPP Stage 5 encourages implementation planning to optimise injury prevention strategy effectiveness. Netball Australia introduced the KNEE (Knee injury prevention for Netballers to Enhance performance and Extend play) program to reduce lower limb injuries. Ankle and knee injuries were the injuries with the highest burden at the 2018 17/U & 19/U ANNC. Therefore, our fourth study investigated how the implementation of the KNEE program could be improved by using concept mapping to identify barriers to its implementation. The coaches, strength and conditioning coaches and physiotherapists working with the 17/U & 19/U State teams identified 'athlete engagement', 'supervision & correction of technique', 'time constraints', 'athlete technique', 'education' and 'support staff resourcing' as the most important and difficult challenges to implementing the KNEE program in pre-elite netball.

The fifth study fulfils TRIPP Stage 6 by assessing the effectiveness of a foot blister prevention strategy at the 2019 17/U & 19/U ANNC. Foot blisters should be easily prevented with correct footwear and hygiene but were one of the most prevalent injuries at the 2018 17/U & 19/U ANNC. The strategy employed in 2019 consisted of every athlete receiving an advice sheet and blister prevention pack (Appendix 17) six weeks prior to the ANNC commencing. A no new shoe less than four weeks before the 2019 ANNC policy was also introduced. Injury surveillance was repeated in 2019, identifying an incidence rate of 82.5 medical attention injuries/1000 player competition hours. Ankle sprains, lower back pain and foot blisters continued to be the most frequently reported medical attention injuries. There was no change in foot blister frequency from 2018 to 2019.

This thesis has identified that the TRIPP framework is useful in guiding essential steps for injury prevention but difficult to apply in the way it was intended in the 'real world'. We identified that ankle sprains, lower back pain and foot blisters were the most prevalent injuries at the 2018 & 2019 17/U & 19/U ANNC, and that ACL rupture and sports-related concussion resulted in the greatest sports

incapacity. Netball Australia had already implemented injury prevention strategies for ankle sprains and ACL rupture. Therefore, the TRIPP Stages should be used as a guide for continual improvement to add value rather than applying the TRIPP model sequentially as it was originally intended, particularly with sports with established prevention programs.

Acknowledgements

This PhD has certainly been a journey. I became passionate about improving development athlete health while working in high performance sport as a physiotherapist and managing elite athletes' injuries that stemmed from their developmental years. Over the past four years I have developed skills I can use to help sporting organisations improve the health of their development athletes.

I would like to firstly thank my panel of supervisors Dr Michael Drew, Dr Juanita Weissensteiner, Dr Phil Newman and Professor Gordon Waddington, for their unwavering encouragement, guidance, and support throughout my candidature. I am incredibly grateful for your expertise, encouragement and understanding whilst I navigated the research world.

Michael, I can't thank you enough and I will be forever grateful for the amount of time and energy you have put into my development as a researcher and as a person over the past four years. You are one of the smartest people I have had the privilege to work with. You have gone above and beyond with your supervision; the countless meetings, emails, phone calls and manuscript revisions are immense, and I hope I can make you proud as my career progresses.

Juanita, I can't tell you how much I appreciated receiving your emails filled with constant support, enthusiasm, and encouragement. To have someone with your development athlete expertise in my corner gave me an enormous amount of confidence. You were always extremely responsive to my needs, and I hope we can continue to work together in the future in our pursuit of improving development athlete health.

Phil, how do I thank you for screening 9831 studies for our systematic review? Such an arduous task and greatly appreciated. Thank you for your support and at times, challenging questions which helped crystalize the direction of my PhD. Your feedback of smiling when you read my thesis definitely gave me a big boost.

Gordon, I was so lucky to have you as my Primary Supervisor. I know how incredibly busy you are, yet you were so responsive to me. Your gentle character combined with a 'can-do' attitude meant I

could approach you with any problem and it would be resolved, resulting in a completed thesis. Thank you for your gifts of knowledge and unwavering support.

Thank you to my other co-authors Dr Laura Piromalli, Alanna Antcliff, Dr Liam Toohey, Dr Jeremy Witchalls, Steven Hughes, Assoc. Prof. Theo Niyonsenga, Ruth Fazakerley, Dr Renee Appaneal and Dr Alex Donaldson. Each of you provided expert knowledge, assistance and support which resulted in five completed studies for my PhD. I have really appreciated and valued the opportunity to collaborate with such a group of respected people in a wide range of fields.

I would also like to thank the staff and my fellow PhD students at the UC Research Institute for Sport and Exercise. You've created such a welcoming and stimulating environment in which to work. Also, I would like to thank the AIS Performance Health (formerly known as Athlete Availability) team. Thank you for including me in your team and the enormous amount of assistance you have provided me with. Additionally, I would like to thank Dr Paula Charlton for her continual encouragement and assistance.

Thank you to the University of Canberra (UC) and Australian Institute of Sport (AIS) for providing me with a scholarship (including maternity leave) which allowed me to study full-time and care for our baby. Many thanks also to the support I received from Netball Australia, particularly Dr Laura Piromalli (nee Juliff) and Alanna Antcliff. Without their unwavering support, this PhD would not have been possible.

This thesis was professionally edited by Matthew Sidebotham in accordance with the *Guidelines for editing research theses* by the Institute of Professional Editors and the Deans and Directors of Graduate Studies. I am very grateful to Murray Turner, University of Canberra Liaison Librarian, who greatly assisted with developing my Systematic Review search strategy and formatting my references.

Finally, I would like to thank my parents, Lynn and Howard, who have always supported me and provided me with a great foundation to build my career on. My father-in-law, Dr Neil Harrigan, who himself started his PhD in his 60s while holding the position of CEO of a large not-for-profit organisation, inspired me to start a PhD. His encouragement and insight have always been welcomed and appreciated. Lastly, I would like to thank my wife, Rachel for her love, support, guidance and

understanding throughout this entire process. Thank you for entertaining our toddler for many hours during lockdown, while I tapped away at my keyboard. I would not have been able to complete this thesis without you. And to our daughter Claudia, *'imagine with all your mind, believe with all your heart, achieve with all your might!'*

Table of Contents

Certificate of authorship of thesis	v
Abstract	vii
Acknowledgements	xi
List of Tables	xix
List of Figures	xxi
List of Appendices	xxiii
List of Abbreviations	XXV
Publications and Presentations	. xxvii
Papers published	xxvii
Papers submitted for publication	xxvii
Conference presentations	xxvii
Supervisors	xxix
Chapter 1 Introduction	1
Background	1
Injuries in pre-elite athletes	2
The sport of netball	4
Injury prevention	5
Importance of injury prevention in pre-elite athletes	5
Classification of injury prevention in sport	7
Models of prevention	8
Knowledge gaps	21
Thesis aim	21
Thesis structure	21
Research Questions	24
Chapter 2 Injury surveillance and establishing aetiology and mechanism of injury (TRIPP 1 & 2)	f 25
Published work	25
Co-authors' declaration	27
A prospective study of health problems at the 2018 17/U and 19/U Australian Netbal National Championships with comparison of surveillance methodology	ll 29
Unpublished data from the 2018 17/U & 19/U ANNC	35
Chapter 3 Developing preventative measure (TRIPP 3)	37

Published work	
Co-authors' declaration	
Injury prevention strategies specific to pre-elite athletes competing in Olympic a professional sports – a systematic review	nd 41
Systematic review update	55
Methods	55
Results	56
Discussion	58
Conclusion	62
Chapter 4 'Ideal conditions'/Scientific evaluation of the preventative m (TRIPP 4)	easure63
Published work	63
Co-authors' declaration	65
Does ankle tape improve proprioception acuity immediately after application and a netball session? A randomised controlled trial	1 following 67
Chapter 5 Describing intervention context to inform implementation str (TRIPP 5)	rategies
Published work	73
Co-authors' declaration	75
Identifying the challenges to implementing a neuromuscular warm-up in pre-elitousing concept mapping	e netball 77
Chapter 6 Evaluating effectiveness of preventative measures in an implementation context (TRIPP 6)	
Paper submitted for publication	
Co-authors' declaration	90
Injury surveillance at the 17/U & 19/U Australian National Netball Championshi effect of a foot blister prevention intervention	ips and the 92
Chapter 7 Discussion and conclusion	
Thesis summary	
Focus of thesis	
Summary of chapters and their key findings	
Discussion of the key thematic findings of the thesis	110
The implementation of injury prevention strategies in a pre-elite population is challen	ging 110
Prevention frameworks are sequential and do not consider measures in place already. be considered when attempting to apply the TRIPP framework in a pre-elite population established prevention programs in place.	This should on with 113
TRIPP Stage 1 (Injury surveillance)	115

TRIPP Stage 2 (Establish aetiology and mechanisms of injury)	118
TRIPP Stage 3 (Develop preventive measures)	119
TRIPP Stage 4 ('Ideal conditions'/scientific evaluation)	121
TRIPP Stage 5 (Describe intervention context to inform implementation strategies)	122
TRIPP Stage 6 (Evaluate effectiveness of preventive measures to implementation context)	124
Thesis strengths	126
Thesis limitations	128
Directions for further research	129
Conclusion	130
References1	133
Appendices1	145

List of Tables

Table 1: Sports incapacity injuries at the 2018 17/U 7 19/U ANNC 36
Table 2: Intervention – description of study participants, level of competition, incidence of injury of
included studies, injury definition and injury definitions concept framework (IDCF)59
Table 3: Characteristics of the studies including data collection period, statistical analysis, efficacy,
level of evidence
Table 4:Intervention summary, duration and frequency, specific details of intervention (i.e., reps, sets,
intensity, progression), co-intervention, reproducible, intervention reporting standard

List of Figures

Figure 1: The integrated FTEM (Foundations, Talent, Elite, Mastery) framework	2
Figure 2: Netball court dimensions	5
Figure 3: The 'sequence of prevention' of sports injuries	8
Figure 4: The TRIPP framework compared to the four-stage van Mechelen approach	9
Figure 5: Comprehensive model for injury causation	12
Figure 6: Six-step intervention development process	14
Figure 7: Schematic representation of the thesis as relates to the TRIPP approach	22
Figure 8: Menstrual cycle data collected by team physiotherapists	35
Figure 9 (Figure 4 in publication). Forest plot	55
Figure 10: PRISMA flow chart	57
Figure 11:Intervention Mapping (IM) Step 5	62
Figure 12:Risk of bias summary	

List of Appendices

Appendix 1: Human Ethics Committee approval for Chapter 2	
Appendix 2: Information to Participants for Chapter 2	147
Appendix 3: Information for Parent/Guardian of Minor for Chapter 2	149
Appendix 4: Informed Consent Form for Chapter 2	151
Appendix 5: Human Ethics Committee approval for Chapter 4	153
Appendix 6: Information to Participants for Chapter 4	155
Appendix 7: Information for Parent/Guardian of Minor for Chapter 4	157
Appendix 8: Informed Consent Form for Chapter 4	159
Appendix 9: Informed Consent Form for Minors for Chapter 4	161
Appendix 10: Human Ethics Committee approval for Chapter 5	
Appendix 11: Information to Participants for Chapter 5	165
Appendix 12: Informed Consent Form for Chapter 5	167
Appendix 13: Human Ethics Committee approval for Chapter 6	169
Appendix 14: Information to Participants for Chapter 6	171
Appendix 15: Information for Parent/Guardian of Minor for Chapter 6	173
Appendix 16: Informed Consent Form for Chapter 6	175
Appendix 17: Foot blister advice sheet and pack	

List of Abbreviations

UC	University of Canberra
AIS	Australian Institute of Sport
FTEM	Foundations, Talent, Elite and Mastery
OSTRC-H	Oslo Sports Trauma Research Centre
HPQ	Health Problems Questionnaire
IOC	International Olympic Committee
STROBE	Strengthening the Reporting of Observational Studies in Epidemiology
IDP	Intervention Development Process
ANNC	Australian National Netball Championships
ACL	Anterior Cruciate Ligament
CONSORT	Consolidated Standards of Reporting Trials
RE-AIM	Reach, Effectiveness, Adoption, Implementation and Maintenance
IM	Intervention Mapping
SSM	Sports Setting Matrix
ANNC	Australian National Netball Championships
RR	Relative Risk
IRR	Incidence Rate Ratio
ACWR	Acute:Chronic Workload Ratio
PRISMA	Preferred Reporting Items for Systematic Reviews and Meta-Analyses
IDCF	Injury Definitions Concept Framework
CERT	Consensus on Exercise Reporting Template
OCEBM	Oxford Centre for Evidence Based Medicine
CNS	Central Nervous System
SEP	Sport & Exercise Physiotherapist
KNEE	Knee injury prevention for Netballers and Enhance performance and
	Extend play
СМ	Concept Mapping
AMEDA	Active Movement Extent Discrimination Apparatus
NCAA	National Collegiate Athletic Association
NSO	National Sporting Organisation
SSO	State Sporting Organisation
OSICS	Orchard Sports Injury Classification System
KTS	Knowledge Transfer Scheme
TIP	Team-sport Injury Prevention
AMS	Athlete Management System
RED-S	Relative Energy Deficiency in Sport
ACC	Accident Compensation Corporation
PROSPERO	International Prospective Register of Systematic Reviews

Publications and Presentations

Papers published

I hereby certify that this thesis comprises five papers of which I am lead author. I have included as a part of the thesis a written statement from each co-author pertaining to my contribution to the joint publications.

- <u>Smyth E</u>, Piromalli L, Antcliff A, et al. A prospective study of health problems at the 2018 17/U and 19/U Australian National Netball Championships with comparison of surveillance methodology. *J Sci Med Sport*. 2020;23:215-21.
- Smyth E, Newman P, Waddington G, et al. Injury prevention strategies specific to pre-elite athletes competing in Olympic and professional sports a systematic review. *J Sci Med Sport*. 2019;22:887-901.
- <u>Smyth E</u>, Waddington G, Witchalls J, et al. Does ankle tape improve proprioception acuity immediately after application and following a netball session? A randomised controlled trial. *Phys Ther Sport*. 2021; 48:20-25.
- 4) <u>Smyth E</u>, Appaneal R, Drew M, et al. Identifying the challenges to implementing a neuromuscular warm-up in pre-elite netball. *Int J Sports Sci Coach*. 2021;16:913-24.

Papers submitted for publication

 <u>Smyth E</u>, Toohey L, Antcliff A, et al. Injury surveillance at the 17/U & 19/U Australian National Netball Championships and the effect of a foot blister prevention intervention. *Transl Sports Med.*

Conference presentations

 <u>Smyth E</u>, Piromalli L, Antcliff A et al. A prospective study of injury epidemiology at the 2018 17/U and 19/U Australian National Netball Championships. Paper presented at: Sports Medicine Australia Conference; 2019; Sunshine Coast, QLD, Australia.

- Smyth E, Appaneal R, Drew M et al. Identifying the challenges to implementing a netball neuromuscular warm-up using concept mapping. Paper prepared for: IOC world conference on Prevention of Injury & Illness in Sport; 2020; Monaco. [Postponed due to COVID]
- 3) <u>Smyth E,</u> Waddington G, Toohey L et al. Injury surveillance at the 2019 17/U & 19/U Australian Netball National Championships and the effect of a foot blister prevention intervention. Paper prepared for: IOC World Conference on Prevention of Injury & Illness in Sport; 2020; Monaco. [Postponed due to COVID]
- Smyth E, Appaneal R, Drew M et al. Using concept mapping to improve implementation of Netball Australia's 'KNEE' program. Paper prepared for: Australian Physiotherapy Association Conference; 2021; Brisbane, QLD, Australia. [Postponed due to COVID]
- <u>Smyth E</u>, Waddington G, Witchalls J, et al. Can ankle proprioception be improved by tape? Paper prepared for: Australian Physiotherapy Association Conference; 2021; Brisbane, QLD, Australia. [Postponed due to COVID]

Supervisors

Primary Supervisor

Prof Gordon Waddington

Director Research Institute for Sport and Exercise University of Canberra

Co-Supervisors

Dr Michael Drew

Program Manager Athlete Performance Health Australian Institute of Sport

Adjunct Associate Professor Research Institute for Sport and Exercise University of Canberra

Dr Phillip Newman

Clinical Associate Professor, Physiotherapy Faculty of Health Research Institute for Sport and Exercise University of Canberra

Dr Juanita Weissensteiner

Principal Advisor Talent Pathways Sector Performance NSW Office of Sport

Chapter 1 Introduction

Background

Sport plays an important role in the lives of Australians. More than 90% of Australian adults express some interest in sport, with 14 million Australians participating in sport each year.^{1,2} High-performance sport has become increasingly competitive; with countries investing hundreds of millions of dollars in elite programs.² Some Australian sports, however, struggle to stay competitive and rely heavily on government funding to stay afloat. With limited resources, it is more important than ever that the available money is used wisely.² Injury has been shown to cause 17% of Australian track and field athletes to retire before age 18,³ more research is required to establish attrition rates across other sports. Athletic development is multifactorial and integrative, dependent on physical growth, biological maturation, skill development and behavioural development.⁴ The Foundations, Talent, Elite and Mastery (FTEM) model is a guiding framework for athletic development (Figure 1),⁵ recognised by the International Olympic Committee (IOC)⁴ and utilised extensively within Australia and internationally. It provides a visualisation of the three major stages of athlete development – Foundational, Talent and Elite/Mastery. It establishes a continuum between participation and sports mastery with ten development phases, which allows for greater understanding of athlete development and progression. The 'Talent' stage starts with T1, marking the beginning of the high-performance pathway. This development phase involves an athlete demonstrating their potential through formal and informal talent identification processes. The next level (T2) requires the athlete's talent to be confirmed by immersing them in realistic training and competition contexts. T3 and T4 are collectively known as the pre-elite stage, defined by the FTEM framework as Talent: Practising and Achieving (T3) and Talent: Breakthrough and Reward (T4).⁵ The T3 phase contains the largest cohort of potential elite athletes, as athletes can remain at this level from 1–8 years or longer.⁵ However these athletes can be vulnerable to

injury and illness because of a lack of support at this stage of development.⁵ An important part of success is the avoidance of injury and illness as the athlete develops,⁶ yet the pre-elite population has been shown to be at high risk of injury^{3,7,8} compared to athletes at other FTEM levels.⁹ The increased risk of

injury at this stage of development can potentially result in a 'bottleneck' for development because of dropout³ and an increased prevalence of subsequent injuries,^{10,11} hence limiting Australia's talent pool.^{12,13} To date, there has been limited research focused on the pre-elite population.



Figure 1: The integrated FTEM (Foundations, Talent, Elite, Mastery) framework. Image courtesy of the Australian Sports Commission, http://www.ausport.gov.au

Injuries in pre-elite athletes

Age and FTEM level align in many sports; however, FTEM level is not defined by an age group, with many, but not all pre-elite athletes in their adolescent years. The IOC has released a consensus statement on youth athletic development outlining some of the challenges faced by adolescent athletes. These challenges include a rapid increase in training load, insufficient rest and recovery, growth and a lack of support services compared to elite athletes.^{4,14,15} The IOC consensus statement clearly states that 'no youth athlete should compete-or train or practice in a way that loads the affected injured area, interfering

with or delaying recovery-when in pain or not completely rehabilitated and recovered from an illness or injury'.⁴ A number of recent studies have reported high injury rates in pre-elite athletes.^{3,9,16-19}

A prospective controlled cohort study investigated injury rates for semi-professional football leagues during the German 2015–16 season.⁹ Data were collected by means of a self-administered questionnaire and a mid-season injury report provided by medical staff. Results showed that elite junior (i.e., preelite) footballers had the highest incidence rate of overuse injuries (7.4/1000hrs) compared to the highest division of semi-professional football (5.4/1000hrs; RR 1.37, 95%CI 0.95-1.98; p=0.04).

Pre-elite injury rates were also assessed at 15 national sports high schools in Sweden using a prospective cohort study.¹⁷ '*By definition, all adolescent athletes studying at National High Sports Schools are elite athletes. To attend these schools, athletes need to be among the top in terms of ranking in their age group for respective sporting events, on national level*'.¹⁷ A weekly self-reported questionnaire (Oslo Sports Trauma Research Centre (OSTRC) Health Problems Questionnaire (HPQ)²⁰ and additional questions developed by Jacobsson et al.²¹) was used to collect injury data across 52 weeks from 284 elite adolescent athletes competing in athletics, cross-country skiing, downhill skiing, freestyle skiing, handball, orienteering or ski orienteering. The injury incidence rate was higher during competition (23.8/1000hrs) than during training (2.8/1000hrs) and, on average, 3 of 10 (30.8%) elite adolescent athletes reported being injured every week.

A Norwegian study also used the weekly self-reported HPQ²⁰ to collect injury data on 260 youth elite athletes (15–16 years of age) across 30 sports for 26 weeks.¹⁶ An average of 43% youth elite athletes had some form of health problem and 25% had substantial health problems during this 26 week period. Substantial health problems were defined as those leading to moderate or severe reduction in training volume or performance, or complete time loss from sport.

In a study in Australian track and field, 103 youth elite athletes were asked to recall over a one to six year period, the type, site and severity of injury sustained when aged 13–17.³ Eighty-one athletes (78.6%) sustained a cumulative 200 injuries with 74% of these athletes injured more than once. Fourteen athletes (17% of injured athletes) reported retirement due to injury before reaching 18 years of age.

Young elite athletes (n=119) from the National Football Training and Development Centre for Females in France were prospectively observed over an eight-year period.¹⁸ The overall rate of injury was 6.4 per 1000 hours exposure. There aren't any published injury rates for female professional footballers in France to compare to but in the USA, an injury incidence rate of 1.93 injuries per 1000 exposure hours has been reported in women's professional football.²² Which is at least three times higher than the French young elite athletes.

The injury data of 166 male pre-elite athletes attending the Middle Eastern Youth Sports Academy were prospectively collected over a five-year (2009–14) period.¹⁹ The athletes competed in athletics, squash, table tennis, fencing, gymnastics, swimming, golf and shooting. An injury was defined as any physical complaint requiring medical attention. The overall injury incidence rate was 5.5/1000 hours of exposure, with the prevalence of overuse injuries being 50.3%.

As outlined by the IOC consensus statement on youth athletic development, adolescent athletes are a unique population and, as demonstrated,^{3,9,16-19} have a high incidence rate of injuries when compared to senior professional athletes.^{9,22} Injuries are a very influential factor that can limit the sustainability of a sport's talent and performance pathway and, in turn, future high-performance success.²³ There is an urgent need to develop a holistic strategy for reducing injuries in pre-elite athletes, with some sports taking formal steps to accomplish this. For example, in 2018 Netball Australia sought assistance from the Australian Institute of Sport (AIS) to help develop aspects of its pre-elite injury prevention strategies.

The sport of netball

Netball is a ball sport played in 77 countries, predominantly in the British Commonwealth.²⁴ It was initially developed as a game played by females, but it is now played by boys and girls, women and men. Nevertheless, only females play netball professionally and represent their countries at the Commonwealth Games. It is a game in which two teams of seven players oppose each other on a rectangular court 30.5m (100 feet) x 15.25m (50 feet) (Figure 2).²⁵ The court is divided into thirds and there are goal circles at each end and a centre circle in the middle of the court. Players, according to their designated position, have specified areas in which they can move.



Figure 2: Netball court dimensions

The aim of the game is to move the ball down the court into the goal circle, from where a goal can be scored by shooting the ball into the goal. Players are not permitted to take a step when they are in possession of the ball and they have three seconds to dispose of the ball. Opposition players must remain three feet (0.9m) away from the player in possession of the ball and contact between opposing players is not permitted (although it frequently occurs). Netball is a fast and skilful game requiring running, change of direction, rapid deceleration and acceleration, jumping, throwing, catching and shooting.^{26,27}

Injury prevention

Importance of injury prevention in pre-elite athletes

Athlete welfare

Injury prevention needs to be prioritised by sporting organisations because, first and foremost, athletes are people whose wellbeing and safety need to be at the forefront of their athletic journey. A recent Australian Human Rights Commission review into Gymnastics in Australia identified 'a "win-at-all-costs" culture that prevailed across the sport and found that this created unacceptable risk for the safety and wellbeing of very young gymnasts'.²⁸ Also, the Gymnastics New Zealand Independent Report,²⁹ outlines there was a fear of disclosing injuries to coaches and training while injured. However this contradicts the IOC consensus statement on youth athletic development, which clearly states: 'No youth athlete should compete when in pain or not completely rehabilitated and recovered from an illness or

injury.⁴ Other reviews into abuse of athletes worldwide are being conducted or have been published in countries including Japan,³⁰ and the United Kingdom.³¹ We will continue to see a culture of long-term recurring injury, burnout, mental ill-health and abuse, if physical health, including prevention of injury, is not prioritised.

Performance

Many factors contribute to athletic performance, including the athlete's psychological and physical health, the athlete–coach relationship and technical, tactical and physical attributes.³² Injuries and illnesses have a detrimental impact on team and individual athletic success.⁶ Several studies have shown that athlete availability is critical to team success in track and field³³, football³⁴ and basketball.³⁵ Pre-competition³⁶ and in-competition³⁷ injuries have both been associated with failure to achieve key performance indicators.

Researchers are beginning to investigate this area of health and performance in pre-elite athletes. One recent study explored the effect of injury on ranking position in young elite athletes in athletics, cross-country skiing and orienteering using a weekly, web-based injury questionnaire; a significant difference between top-ranked (3.6%) and middle-ranked (2.3%) athletes compared to low-ranked athletes (11.3%) for substantial injuries was reported.³⁸ Ranking position was determined by each sport according to past competition results, with the top ranked athletes being better performers. Substantial injury was defined as an injury leading to moderate or severe reductions in either training volume, performance or complete inability to participate in sport. Larruskain et al.³⁹ recently reported their findings from a prospective six-season injury surveillance study at an elite football academy. They demonstrated that injuries were negatively associated with player progression through the academy pathway towards reaching the First team. This type of research, demonstrating the negative impact injury has on performance strengthens the argument for more investment in injury prevention strategies for the pre-elite athletic population.

Classification of injury prevention in sport

Natural history of disease

In order to understand how to prevent injury it is important to understand the 'natural history of disease', which is the progress of a disease in an individual, from onset to recovery, disability or death, without intervention.⁴⁰ According to Gordis,³⁷ there are four main stages in the natural history of disease. The first is the 'susceptibility stage', when risk factors (such as playing netball) increase the chance of injury. The second stage is 'subclinical', in which the athlete may be pain-free but pathological changes begin, such as inflammation of a growth plate. In the third stage, 'clinical', signs and symptoms become apparent, such as Achilles insertional pain and stiffness. Then, in the fourth stage of 'recovery, disability or death', the athlete stops playing netball and suffers an ongoing disability. Therefore, injury prevention interventions must be targeted according to the stage of disease.

Categorisation of prevention in sport

Injury prevention intervention classification has been described in two ways:⁴¹ by observation of pathology and by risk indicators identified in athletic populations. When considering pathology, the most effective form of intervention is preventing the initial development of disease by immunisation or reducing exposure to a risk factor.⁴⁰ Classification of this type of prevention generally describes 'primary prevention'. 'Secondary prevention' is at the 'clinical' stage and involves early detection of existing disease to reduce severity and complications, such as screening for cancer; 'tertiary prevention' involves reducing the impact of the disease, such as through rehabilitation for a muscle injury.⁴⁰ When focusing on risk factors, injury prevention interventions are classified as 'universal', 'selective' or 'indicated'. Universal measures are recommended for everyone playing sport; examples include good hydration, adequate rest and use of protective equipment. Selective injury prevention is for subgroups based on sex, age, sport etc. Athletes in this subgroup do not possess any individual risk factors but are at risk because of their subgroup. An example of this is knee injury prevention interventions are for individuals with risk factors that place them at higher-than-average risk of developing an illness or

injury. These risk factors can include previous injury, hamstring weakness and cardiac irregularities, among others, and are often identified through periodic health evaluations.

Models of prevention

The van Mechelen 'sequence of prevention' model

The van Mechelen model (Figure 3) was developed to improve the understanding of sports injuries and create a scientifically based prevention strategy.⁴² Guidelines were established for injury surveillance and reporting, such as expressing injury incidence as the number of sports injuries per exposure time and the need for consistent 'sports injury' and 'sports participation' definitions. The model consists of four stages. Firstly, the extent of the sports injury problem must be identified and described, outlining incidence and severity. Secondly, contributing factors and mechanisms of injury must be determined. Based on these findings, the third step is to introduce a preventative measure; and, finally, the intervention is evaluated by repeating step one. As injury prevention research has evolved over time, the need to further develop this model has become apparent.



Figure 3: The 'sequence of prevention' of sports injuries.

Translating Research into Injury Prevention Practice (TRIPP)

Translating Research into Injury Prevention Practice (TRIPP),⁴³ now a well-established framework, consisting of six stages and is an expansion of the van Mechelen model (Figure 4).⁴² It will provide the foundation for our investigation into injury prevention in pre-elite athletes as it has been promoted as an approach that will lead to real-world injury prevention gains.



Figure 4: The TRIPP framework compared to the four-stage van Mechelen approach.

Stage One focusses on establishing valid and reliable methods of injury surveillance to allow for routine and ongoing monitoring and reporting of sports injuries. The objective of Stage 2 is to understand the aetiology of injuries to ensure injury prevention interventions target the specific cause of injury. Stage 3 was introduced by Finch⁴³ to develop potential solutions to the injury problem based on evidence-based practice. Ideally, a multidisciplinary team including the adopters (i.e., those who will decide to use the intervention) and implementers (i.e., those who will deliver the implementation) should develop the injury prevention intervention together. Stage 4 involves assessing the injury prevention intervention under 'ideal conditions' – controlled, targeted and well-resourced. Stage 5 was also introduced by Finch⁴³ to plan implementation in the real-world to encourage behavioural change. Finally, Stage 6 involves implementing the injury prevention intervention in the real world and evaluating its effectiveness. In the following section we will address each of the stages in detail, examining evidence and application to date within pre-elite level netball, the focus of our investigations.

TRIPP Stage 1 (Injury surveillance)

High-quality injury surveillance information is crucial for informing all other stages of injury prevention. Valid and reliable methodologies must be used to allow for routine, ongoing monitoring

and reporting.⁴⁴⁻⁴⁶ The recently published IOC consensus statement, *Methods for Recording and Reporting of Epidemiological Data on Injury and Illness in Sport 2020*, outlines a new Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) checklist. It provides recommendations for defining and classifying health problems, severity of health problems, capturing and reporting athlete exposure, expressing risk, burden of health problems, study population characteristics and data collection methods.⁴⁷ Applying these recommendations when fulfilling TRIPP Stages One and Two will ensure accurate injury surveillance that will guide the remaining TRIPP stages to achieve injury prevention.

Traditionally, injury surveillance studies have used time loss from sport for defining the occurrence of an injury.⁴⁸ This approach does not capture the full impact of overuse injuries, as athletes can often continue to train and compete despite ongoing symptoms.⁴⁹ Whalan et al.⁵⁰ captured self-report data from 25 semi-professional football teams using the HPQ, which identifies any health problem regardless of time loss. They demonstrated the risk of a time loss injury was 3.6 to 6.9 times higher when preceded by a 'minor' and 'moderate' non-time loss complaint. Earlier detection using the HPQ can lead to early intervention, which may reduce the burden and severity of injury.⁵¹

This thesis will investigate the use of different injury data collection methods, i.e., athlete self-report and clinical examination injuries in netball to identify both, time loss and non-time loss injuries.⁵² The athlete self-report method will provide us with prevalence measures and the clinical examination method will result in incidence rate measures. Both measures can be useful depending on the goals and circumstances of the study. Nielsen et. al.,⁵³ recommends incidence-based measures be used for assessing aetiology and prevention strategies, whereas prevalence-based measures be used for establishing treatment needs and assessing athlete availability.

Application of TRIPP Stage 1 in pre-elite netball

A recent systematic review⁵⁴ of injury surveillance in netball identified 45 articles, the earliest publication being 35 years ago, in 1986.⁵⁵ Eighty-seven per cent of these studies were conducted in Australia, New Zealand or South Africa. They consisted of prospective studies over one or more playing

seasons (n=14, 30%), prospective studies at tournaments (n=6, 13%), retrospective studies (n=9, 20%) and studies accessing hospital and insurance databases (n=17, 37%).

Four studies have reported prospective injury rates for pre-elite netball athletes during a season or at a tournament. The ankle and knee have been the most frequent sites of injury during an elite & pre-elite 2013 netball season,⁵⁶ the 2009 South African Netball Championships,⁵⁷ and the 1995 New South Wales State Netball Championships.⁵⁸ Ankle sprains were the most frequent injuries recorded by Hopper and Elliot⁵⁹ at the 1988 Australian National Championships, followed by shin/calf injuries. The three tournament studies^{57,59} defined an injury as any complaint requiring medical attention, whereas the study reporting pre-elite injuries during a netball season used a time loss injury definition. These studies indicate that injury prevention strategies should continue to focus on the ankle and knee, but it is imperative that comprehensive standardised injury surveillance⁴⁸ for the target population is always established prior to introducing an injury prevention strategy.

TRIPP Stage 2 (Establish aetiology and mechanisms of injury)

In order to prevent injury, we need to understand what causes or contributes to injury.⁴³ In 1994, Meeuwisse⁶⁰ presented a multifactorial model of athletic injury aetiology. It describes an interaction of intrinsic risk factors (i.e., age, strength, previous injury), extrinsic risk factors (i.e., equipment, weather conditions, rules) and an inciting event to cause injury. In 2005, Bahr and Krosshaug⁶¹ expanded on Meeuwisses's model to include an expanded biomechanical approach and a focus on sport-specific characteristics (Figure 5).



*Figure 5: Comprehensive model for injury causation. BMD, Body mass density; ROM, range of motion. Reprinted from Bahr & Krosshaug.*⁶¹

This model demonstrates many contributing factors to injury; however, if each risk factor is assessed separately without controlling for other risk factors, it will not result in a clear understanding of how each contributes to injury.⁶⁰ A systems-based research approach, as outlined by Hulme et al.,⁶² is required for a better understanding of causation, but discussion and critical analysis of this is beyond the scope of this thesis.

Application of TRIPP Stage 2 in pre-elite netball

All four pre-elite netball studies previously described⁵⁶⁻⁵⁹ collected aetiology and mechanism of injury data. Age,^{56,58,59} level of competition,⁵⁶⁻⁵⁹ playing position,^{56,57,59} injury history,⁵⁷⁻⁵⁹ mode of onset,⁵⁷⁻⁵⁹ time of onset,^{57,59} footwear and use of tape or a brace^{58,59} and mechanism of injury^{58,59} data were routinely collected. Pickering Rodriguez et al.⁵⁶ were particularly interested in examining the relationship between lower-body stiffness and injury incidence, so they also collected active and quasi-static stiffness data for the lower body. Adherence to an injury prevention strategy was also recorded by Hume and Steele,⁵⁸ and Hopper and Elliot⁵⁹ included a podiatric assessment and recorded the prevalence of existing injuries reported by players at the start of the championships.

These four studies have provided us with a better understanding of what contributes to pre-elite netball injuries, but there is a need to expand this knowledge to aid future injury prevention strategies. There was no relationship between lower-limb stiffness and injury in pre-elite athletes.⁵⁶ Hopper and Elliot,⁵⁹ reported that most injuries occurred in the second quarter of the game (32.7%) whereas Langeveld et al.⁵⁷ found most injuries occurred in the third quarter (26%). The most frequent mechanisms of injury were 'incorrect landing'⁵⁹ and 'contact with a player'.^{57,59} Additionally, according to Langeveld et al.⁵⁷ the position of Goal Defence sustained the most injuries (22%).

Coetzee et al.⁶³ reported additional data from the tournaments previously described by Langeveld et al.⁵⁷ They identified that a cement surface had a 1.9 times higher injury rate than a synthetic surface and reported that 52% of injured athletes did not perform any core stability sessions and 58% did not perform neuromuscular control sessions, but 64% did partake in flexibility training. Hume and Steele⁵⁸ also found a lack of awareness of injury prevention strategies, with 28% of injured athletes not knowing what a wobble or balance board was.

The mechanism of injury data presented by Hume and Steele⁵⁸ is a good example of why we need consistent definitions and methodology, as outlined by the IOC consensus statement.⁴⁷ They reported on 'netball action at the time of injury', the most frequent responses being 'other' (21%) and 'combinations of movement' (18%). Neither of these responses provide us with a better understanding of mechanism of injury. It must be acknowledged that the data for this study were collected 26 years ago and our understanding of data collection requirements has improved greatly since then.

Several studies have examined the relationship between netball biomechanics and injury, but none specifically for pre-elite athletes.^{27,64,65} Landings have been identified as a high-risk activity for Anterior Cruciate Ligament (ACL) injuries in netball.⁶⁵ Using unanticipated 180° turn agility sprints, Maulder⁶⁶ identified a dominant limb asymmetry of greater than 10% as having moderate association with injury.⁶⁶ Downs'⁵⁴ systematic review identified a lack of studies reporting the circumstance and mechanisms of injury in order to provide more meaningful data for developing prevention strategies.

TRIPP Stage 3 (Develop preventive measures)

This stage involves identification and development of potential solutions to the injury problem guided by TRIPP Stage 2, in which potential risk factors have been identified.⁴³ A multi-disciplinary approach is required to consider all possible injury prevention strategies. Following the Intervention Development Process (IDP)⁶⁷ is one way of satisfying Stage 3 of TRIPP. This involves combining research evidence and clinical experience as well as engaging end-users early to develop an injury prevention strategy. The IDP was successfully used to create a lower-limb injury prevention exercise training program ('FootyFirst') for community Australian football.⁶⁷ This process recognises that research evidence alone is insufficient to implement injury prevention strategies⁶⁸ and ensures that the best available scientific evidence is combined with practitioner expertise and end-user values.



Figure 6: Six-step intervention development process. Reprinted from Donaldson et al.⁶⁷

Application of TRIPP Stage 3 in pre-elite netball

There is no evidence of the six-step IDP being applied in pre-elite netball. Two observational studies have been identified that could be considered to partially satisfy Step 4 of the IDP. Fox et al.⁶⁹ used preelite athletes to assess whether biomechanical errors identified during a single-leg squat correlated with errors identified during a netball-specific leap landing. Landings have been identified as a high-risk activity for ACL injuries in netball. In elite-level netball 13 out of 16 (81%) cases occurred when an athlete was landing from a jump.⁶⁵ Because of this association between landing and ACL injury, they concluded the single-leg squat may be a useful movement screen for identifying netball players who may be at risk of sustaining an ACL injury. Sinclair et al.⁷⁰ investigated the effect of knee bracing on knee joint kinetics and kinematics using pre-elite netball athletes. They found no difference in joint kinetics but an increase in subjective knee stability when wearing the brace. For these two studies to be effective, researchers, clinicians and end-users need to incorporate this work into the intervention development process to create an implementable intervention.

TRIPP Stage 4 ('Ideal conditions'/scientific evaluation)

Once the intervention is developed in TRIPP Stage 3, its efficacy must be assessed in a controlled and targeted manner. This is usually in the form of a randomised controlled trial or a cohort study. The CONSORT (Consolidated Standards of Reporting Trials) statement is the gold standard for transparent reporting of design, analysis and interpretation for randomised controlled trials.⁷¹ The STROBE statement provides guidelines for reporting a cohort study.⁷² These studies, even if they are field-based, are often conducted in an artificial environment in which the investigators deliver the intervention, participating teams and players are provided with staff and resources and participants are provided with incentives and reminders.⁴³ Most often, once the research is completed the injury prevention strategy is not continued, because very few sporting associations possess the human resources and/or finances required to maintain the intervention.⁷³ To prevent this from occurring, TRIPP Stage 5 and Stage 6 were introduced to encourage a greater emphasis on implementation of an injury prevention strategy.⁴³

Application of TRIPP Stage 4 in pre-elite netball

No studies have investigated the efficacy of injury prevention strategies in pre-elite netball. Two publications were identified using university student participants to test the efficacy of two injury prevention training programs.^{74,75} One of the programs assessed the efficacy of an eight-week barefoot training programme on ankle stability, agility and speed;⁷⁴ the other assessed the efficacy of a six-week neuromuscular/safe-landing training program called 'Down to Earth' for reducing injury incidence.⁷⁵ The barefoot training resulted in a significant improvement in agility and ankle stability. The barefoot group significantly improved their performance in the 505-Agility test to the left (p = 0.01; d = 1.04)

and the right (p = 0.002; d = 1.4); As well as single leg balance as assessed on the Biodex Balance System, showing superior overall stability of the right leg (p = 0.01; d = 1.62), anterior-posterior stability of the right leg (p = 0.01; d = 1.36) and medial-lateral stability of the right leg (p = 0.04; d = 1.14).⁷⁴ The 'Down to Earth' program resulted in a significant reduction in training injuries during the six weeks following the intervention period. Ten (38%) injuries were recorded for the intervention group and 15 (75%) were recorded for the control group (RR 0.01, 95%CI 0.001-0.376; p=0.02).⁷⁵ However, caution is advised when interpreting these results, as there were 32% (n=8) fewer athletes in the intervention group and only an athlete self-report injury definition was employed. This lack of evidence for injury prevention strategies in pre-elite netball athletes must be addressed.

TRIPP Stage 5 (Describe intervention context to inform implementation strategies)

TRIPP Stage 5 is essential for determining whether efficacy research is translatable into actions that can be implemented in the real-world context of sport. Firstly, a clear understanding of what the current safety behaviours are, and whether they need to change, is required. Secondly, if safety measures are not being adopted, we need to know why. Gaining an insight into motivators/barriers to uptake and an understanding of the knowledge and attitudes of players, coaches and sports bodies towards injury and prevention of injuries is essential for planning implementation.⁴³ Sports injury interventions are often ineffective because they are not widely accepted and adopted.⁷⁶ A number of tools exist to facilitate the implementation of injury prevention interventions;⁷⁶⁻⁸² intervention mapping (Step 5), the Reach, Effectiveness, Adoption, Implementation and Maintenance (RE-AIM) framework and concept mapping are examples of implementation strategies recently used in the sports injury context. There has been recent growth in sports medicine studies utilising these processes but they need to become standardised for all injury prevention interventions to achieve a reduction in injury/illness rates.⁸³

Implementation sciences

Intervention mapping (step 5)

Intervention mapping (IM) is a protocol for systematic planning for behavioural change based on theory and evidence.⁸⁰ IM is a six-step process that starts with problem identification and ends with problem solving or reduction. There are several tasks in each of the six steps. It is possible to use IM (Step 5)

independently to complete TRIPP Stage 5 for planning the adoption, implementation and sustainability of an injury prevention intervention in the real world.⁷⁹ The tasks for this stage include:

- 1. Identification of potential adopters
- 2. Identification of potential implementers
- 3. Establish an implementation planning group with representatives from adopters and implementers
- 4. State intervention use outcomes and specify performance objectives for reach, adoption and implementation
- 5. Specify determinants of reach, adoption and implementation
- 6. Select strategies for reach, adoption and implementation that are theory-informed, evidencebased and context-specific
- 7. Design interventions for reach, adoption and implementation.

This IM protocol has been used to develop plans for the 'Mayday' neck and spinal injury prevention policy in rugby union⁸⁴ as well as the FootyFirst exercise training programme to prevent leg injuries in Australian football.⁷⁹

RE-AIM (reach, effectiveness, adoption, implementation and maintenance) framework

This framework is primarily used as an evaluation tool for intervention studies, but it can also be used as a planning tool alongside TRIPP and other implementation strategies. It was developed on the basis that behaviour change will only occur if interventions are available to the group of interest, adopted by them, used as they were intended and then sustained. There are five components:

- **Reach** is the absolute number, proportion and representativeness of individuals willing to participate in the intervention. Planning should consider the demographics of the intended population and potential barriers that will limit success in reaching the target population.
- Effectiveness is the impact the intervention will have on important outcomes, including potential negative effects. This component determines if the intervention is evidence-based or a new innovation and key stakeholders should be consulted to determine how success will be defined and measured.
- Adoption is the absolute number, proportion and representativeness of implementers and adopters that are willing to initiate the intervention. Consideration should be given to the percentage of target population that will initiate the intervention as well as to what the greatest barriers to adoption will be.
- **Implementation** refers to the consistency, cost and adaptations made during delivery. Planning needs to ensure the intervention can be delivered consistently but also has the flexibility to change mid-course if required.
- **Maintenance** is the ability of the intervention to become institutionalised or part of routine practice. Plans should be made to support initial success and prevent drop-off and to determine whether resources would assist in providing long-term support.

The RE-AIM framework has been used to successfully implement several sports injury prevention interventions, such as a safe-landings programme for junior netballers⁸⁵ and community footballers.⁸⁶

Concept mapping

Concept mapping has been used for injury prevention in public health⁸⁷ and in sport.⁸⁸ It was developed in the 1980s using qualitative and quantitative methods to enable a group of people to articulate ideas and have these ideas graphically and objectively represented.⁸⁹ The structured process involves
brainstorming relevant ideas, after which each participant is asked to sort the ideas into groups that make sense to them and then to rate each idea for how important they think it is and how difficult they think it will be to implement that idea. The Concept Systems Global MAX⁹⁰ website is an effective way of facilitating this process.

Knowledge and attitudes questionnaire

Exploring the knowledge and attitudes of players, coaches and sports organisations in relation to the frequency, causes and prevention of injuries as they occur in their sport is an important part of implementation planning. Shamlaye et al.⁹¹ used a web-based nationwide survey of 538 football coaches to discover the attitudes, beliefs and factors influencing their adherence to the FIFA 11+ injury prevention programme. One finding was that coaches with less experience may be less likely to use the 11+, meaning this population should perhaps be the target for future education workshops.

Application of TRIPP Stage 5 in pre-elite netball

There is no evidence of TRIPP Stage 5 being applied in a pre-elite netball population. In total, four netball studies focusing on implementation have been identified: four in community netball^{85,92,94} and one involving secondary school coaches and players.⁹⁵ Two of the community netball studies were conducted to assess the efficacy and implementation of injury prevention-related coach education.^{85,92} Components of the RE-AIM framework⁹² and the full RE-AIM framework⁸⁵ were applied when seeking feedback from community coaches after attending an education session. Part of the RE-AIM framework was also applied in another community netball study, reporting the implementation of the Netball KNEE program.⁹⁴ Fidelity was found to be poor, with only 12–18% of recommended activities performed as intended across three age groups. The final community netball study used a questionnaire developed from the Theory of Planned Behaviour to assess player behavioural factors associated with players' intentions to learn correct landing technique during coach-led training sessions.⁹³ The secondary school study investigated coach and player attitudes to injury.⁹⁵ Of the 226 players surveyed, 87% (n=196) reported hiding or downplaying an injury. Similarly, of the 117 coaches, 87% (n=102) had witnessed a player continue to play on when they thought they should not have. The absence of

more implementation studies such as these is indicative of why there is a lack of evidence for effective 'real-world' injury prevention strategies in netball.

TRIPP 6 (Evaluate effectiveness of preventive measures to implementation context)

This stage determines how effective the scientifically proven interventions are when applied to the realworld context. The intervention is implemented and its effectiveness is evaluated.⁴³ A good example of this was the promotion of protective eyewear for squash players.⁹⁶ The protective eyewear promotion (PEP) project, consisting of availability of eyewear and educational material, resulted in 'PEP' players being 2.4 times more likely to wear eyewear than the control group. The implementation of this successful strategy was guided by planning (i.e., TRIPP Stage 5) conducted by some of the same researchers using principles of behaviour change.⁸² When the International Sporting Federation Chief Medical Officers were surveyed, they valued TRIPP Stage 5/6 research most highly out of all TRIPP stages.⁹⁷ However there is a lack of this type of research, with a systematic review revealing that between 2007 and 2009 only 5% of sports injury prevention research involved implementation (i.e., TRIPP Stage 5) or effectiveness (i.e., TRIPP Stage 6) studies.⁹⁸

Application of TRIPP Stage 6 in pre-elite netball

To date, there have not been any studies evaluating the effectiveness of injury prevention strategies (i.e., TRIPP Stage 6) applied to the netball pre-elite population. We have identified one netball effectiveness study investigating the effect of an integrated functional stability program on injury rates in an international netball squad.⁹⁹ It is a good example of applying the ecological systems theory, ensuring athletes, support staff, coaches and management have a role to play in injury prevention. It is of concern that netball has been played for over 100 years¹⁰⁰ and yet there is only one study assessing the effectiveness of an injury prevention strategy in a 'real-world' setting.

Knowledge gaps

There is a significant lack of sports injury prevention research that has real-world relevance.⁹⁸ When all sports injury prevention publications between 1938 and 2010 were retrieved (n=11,859), only 4% assessed the effectiveness of implementation of their research findings.⁹⁸ Most research addresses the early stages of the TRIPP framework with little regard for implementation of findings, which limits injury prevention potential. For injury prevention intervention implementation to be successful, we need to acknowledge that effective interventions are multifaceted and complex and must target multiple socioecological levels.^{76,101} In both community and high-performance sport there are often many levels of hierarchy, with many stakeholders at all levels who can influence the success or failure of an injury prevention intervention. Therefore, the development of interventions and their implementation strategies must be holistic with a whole-of-sport and whole-of-industry approach. Furthermore, when targeting the pre-elite population, interventions and their implementation strategies need to be specific to that population to succeed.⁶⁷

Thesis aim

The purpose of this thesis is to develop a holistic strategy to reduce injuries in pre-elite athletes, using netball as an exemplar sport. This strategy was trialled within one of our national sporting organisations (i.e., Netball Australia), working closely with the sport to determine their injury profile and then continuing to develop, plan and implement effective injury prevention interventions. This research, using netball as an example will provide much-needed guidance for injury prevention in pre-elite athletes to reduce injuries sustained throughout their development pathway. Stemming from this will be the long-term aim of supporting robust elite athletes achieving greater success with less injury.

Thesis structure

This thesis follows the TRIPP framework to develop a process for real-world injury prevention gains in the pre-elite population, using netball as an exemplar sport. The TRIPP framework is made up of a series of six sequential steps focused on developing and implementing one preventative strategy. Prior to the commencement of this research, Netball Australia undertook injury surveillance at the 2017 17/U & 19/U Australian National Netball Championships (ANNC). It had also developed focused activities in this area, including, a neuromuscular warm-up designed to prevent injury, and prophylactic ankle taping was encouraged. While all stages of the TRIPP framework are necessary, it was not possible to apply the steps sequentially for a single preventative strategy because there were multiple injury types to address, and some preventative strategies were already in place. Instead, we used each step of the TRIPP framework with a pragmatic sequence approach (see Figure 6 below for a summary of the studies in relation to the TRIPP framework).



Figure 7: Schematic representation of the thesis as relates to the TRIPP approach.

This thesis has seven chapters. Chapters 2, 3, 4 and 5 incorporate published works and the research work reported in Chapter 6 has been submitted for publication. Chapter 7 is an overall discussion of the findings from each study and the implications of the study findings for future research and practice.

Chapter 2 presents TRIPP Stage 1 (Injury surveillance) and TRIPP Stage 2 (Establish aetiology and mechanisms of injury) in the form of injury surveillance findings of the 2018 17/U & 19/U Australian National Netball Championships.

Published as: **Smyth E**, Piromalli L, Antcliff A, et al. A prospective study of health problems at the 2018 17/U and 19/U Australian National Netball Championships with comparison of surveillance methodology. *J Sci Med Sport*. 2020;23:215-21.

Chapter 3 is a systematic review of injury prevention intervention in pre-elite athletes. This is a vital part of TRIPP Stage 3 (Develop preventive measures).

Published as: **Smyth E**, Newman P, Waddington G, et al. Injury prevention strategies specific to preelite athletes competing in Olympic and professional sports – a systematic review. *J Sci Med Sport*. 2019;22:887-901.

Chapter 4 reports the findings of a randomised controlled trial investigating the effect of ankle tape on proprioception. Ankle taping is routinely used in netball to prevent ankle sprains but the mechanism for its effectiveness is not well understood. This study fulfils TRIPP Stage 4 ('Ideal conditions'/scientific evaluation).

Published as: **Smyth E**, Waddington G, Witchalls J, et al. Does ankle tape improve proprioception acuity immediately after application and following a netball session? A randomised controlled trial. *Phys Ther Sport.* 2021; 48:20-25.

Chapter 5 presents TRIPP Stage 5 (Describe intervention context to inform implementation strategies). Netball Australia has developed a neuromuscular warm-up called the KNEE program to prevent lowerlimb injuries. This chapter introduces concept mapping to improve the implementation of this program.

Published as: **Smyth E**, Appaneal R, Drew M, et al. Identifying the challenges to implementing a neuromuscular warm-up in pre-elite netball. *Int J Sports Sci Coach*. 2021;16:913-24.

Chapter 6 reports injury surveillance results for the 2019 17U & 19U Australian National Netball Championships following the implementation of an injury prevention strategy. This process is TRIPP Stage 6 (Evaluate effectiveness of preventive measures in implementation context).

Submitted for publication in the peer-reviewed journal Translational Sports Medicine.

Research Questions

Chapter 2 Study 1: What is the injury incidence and prevalence for 17/U & 19/U netball athletes competing at the 2018 Australian National Netball Championships?

Chapter 3 Study 2: What injury prevention interventions have been determined to be effective in pre-elite athletes competing in Olympic and professional sports?

Chapter 4 Study 3: Does ankle tape improve proprioception acuity in pre-elite netball athletes immediately after application and following netball training? Is there any difference in proprioception if ankle tape is self-applied compared to being applied by a Sport and Exercise Physiotherapist?

Chapter 5 Study 4: What are the barriers to implementing the 'KNEE' injury prevention strategy in a pre-elite netball population?

Chapter 6 Study 5: What is the injury incidence and prevalence for 17/U & 19/U netball athletes competing at the 2019 Australian National Netball Championships? Can foot blister incidence be reduced at the 2019 17/U & 19/U ANNC compared to the 2018 17/U and 19/U ANNC by implementing a foot blister prevention intervention?

Chapter 2 Injury surveillance and establishing aetiology and mechanism of injury (TRIPP 1 & 2)

Published work

This chapter is based on a manuscript that has been published in the Journal of Science and Medicine in Sport.

Smyth E, Piromalli L, Antcliff A, et al. A prospective study of health problems at the 2018 17/U and 19/U Australian National Netball Championships with comparison of surveillance methodology. *J Sci Med Sport*. 2020; 23:215-21.

At the time of publishing, there were no studies reporting injury rates specifically in pre-elite netball players in the previous decade. The TRIPP framework provides a structure for developing and implementing injury prevention strategies. The critical first stage of the TRIPP process is injury surveillance to determine which injuries have the highest burden to direct future injury prevention strategies. The second TRIPP stage is to determine aetiology and mechanisms of injury to also direct how these costly injuries may be prevented.

None of the 192 pre-elite athletes have constant physiotherapy access throughout the year like they do at the ANNC, which limits surveillance of accurate 'whole of year' injury data. Therefore, conducting injury surveillance at a tournament where all athletes have access to a team physiotherapist was considered a practical method for obtaining injury information for this population within real-world constraints. An alternative method of injury surveillance that is not reliant on access to a health professional, the Health Problems Questionnaire (HPQ),²⁰ was also trialled. This injury surveillance tool is a self-report method that has been found to be sensitive and valid in documenting acute injuries, overuse injuries and illnesses.^{102,103} The HPQ self-report injury surveillance method was used in addition to medical attention injury surveillance to allow for injuries to be recorded for four weeks following the tournament and to compare percentage of agreement between the two surveillance tools. The medical attention injury records were also designed to capture mechanism of injury, onset of injury

(i.e., new versus recurrent) and nature of injury (i.e., trauma versus overuse). This additional information will help inform future injury prevention strategies.

The aim of this study was to describe the injuries that occurred over the six-day 2018 17/U & 19/U ANNC, compare the injury profile between age groups, describe contributing factors, compare two injury surveillance methods and determine whether exposure to the tournament alters the health of participants in the following four weeks.

Co-authors' declaration

Declaration of Co-Authored Publications

For use in theses which include co-authored publications. This declaration must be completed for each coauthored publication and to be placed at the start of the thesis chapter in which the publication appears, or as a preface to the thesis.

UNIVERSITY OF

Declaration for Thesis Chapter Two

Publication: "A prospective study of health problems at the 2018 17/U and 19/U Australian National Netball Championships with comparison of surveillance methodology"

DECLARATION BY CANDIDATE

In the case of Chapter Four, the nature and extent of my contribution to the work was the following:

 Nature of Contribution
 Extent of Contributions (%)

Conception and design of the research proposal	80%
Data collection	50%
Analysis and interpretation of the findings	70%
Writing the paper and critical appraisal of content	70%
Corresponding author for journal communication and publication peer-review process.	100%

The following co-authors contributed to the work:

Name	Nature of Contribution	Contributor is also a UC student (Yes/No)
Gordon Waddington	Conception and design of the research proposal Data collection Analysis and interpretation of the findings Writing the paper and critical appraisal of content	No
Alanna Antcliff	Conception and design of the research proposal Analysis and interpretation of the findings	No
Phil Newman	Writing the paper and critical appraisal of content	No
Juanita Weissensteiner	Writing the paper and critical appraisal of content	No
Laura Piromalli	Conception and design of the research proposal Analysis and interpretation of the findings	No
Michael Drew	Conception and design of the research proposal Analysis and interpretation of the findings Writing the paper and critical appraisal of content	Νο

2

_17/09/2021____ Date

Candidate's Signature

DECLARATION BY CO-AUTHORS

The undersigned hereby certify that:



Declaration of Co-Authored Publications

(1) the above declaration correctly reflects the nature and extent of the

- candidate's contribution to this work, and the nature of the contribution of each of the co-authors.
- (2) they meet the criteria for authorship in that they have participated in the conception, execution, or interpretation, of at least that part of the publication in their field of expertise;
- (3) they take public responsibility for their part of the publication, except for the responsible author who accepts overall responsibility for the publication;
- (4) there are no other authors of the publication according to these criteria;
- (5) potential conflicts of interest have been disclosed to (a) granting bodies, (b) the editor or publisher of journals or other publications, and (c) the head of the responsible academic unit; and
- (6) the original data are stored at the following location(s) and will be held for at least five years from the date indicated below:

[Please note that the location(s) must be institutional in nature, and should be indicated here as a department, centre or institute, with specific campus identification where relevant.]

Location(s):	Australian Institute of Sport		
Laura Piroma	lli I O : MA		
Signed:	Leromalli		

31/8/2021 Date: Alanna Antcliff Signed: Date:

Phil Newman Signed: 11/09/2021

Date: Gordon Waddington 122

Signed:	- 0	_
Date:	17/09/2021	

Juanita	Weissensteiner	
Signed:	Juanita weissensteiner	
Date:	30/08/2021	

Michael Drew	Michael Themi
Signed:	i dioreoro.
Date:	17/09/2021

A prospective study of health problems at the 2018 17/U and 19/U Australian Netball National Championships with comparison of surveillance methodology



Original research

A prospective study of health problems at the 2018 17/U and 19/U Australian National Netball Championships with comparison of surveillance methodology



Erin A. Smyth^{a,b,*}, Laura Piromalli^c, Alanna Antcliff^c, Phillip Newman^{b,e}, Gordon Waddington^{a,b}, Juanita R. Weissensteiner^f, Michael K. Drew^{a,d}

^a Athlete Availability Program, Australian Institute of Sport, Australia

^b Research Institute for Sport and Exercise, University of Canberra, Australia

^c Netball Australia, Australia

^d Australian Collaboration for Research into Injury in Sport and its Prevention (ACRISP), Australia

^e Faculty of Health, University of Canberra, Australia
^f Sport Development Group, New South Wales Office of Sport, Australia

sport Development Group, new south wates Office of sport, Austric

ARTICLE INFO

Article history: Received 12 April 2019 Received in revised form 4 September 2019 Accepted 7 October 2019 Available online 15 October 2019

Keywords: Sports medicine Surveillance Sprains and strains Athletic injuries

ABSTRACT

Objective: To investigate the incidence, site, nature and cause of injuries sustained during and four weeks following the 2018 Australian National Netball Championships (ANNC's) using medical attention and self-report surveillance tools.

Design: Prospective cohort study.

Method: Injuries were recorded prospectively using medical attention and self-report data collection methods. One hundred and ninety-two athletes competed at the 2018 ANNC's with 96 athletes in each age group (17/U & 19/U).

Results: There were 103 medical attention injuries sustained by 80 athletes resulting in an incidence rate of 89.4 per 1000 player hours. The most frequently recorded medical attention injury diagnoses across both age groups were lateral ankle ligament sprain (n=14, 13.6%), foot blisters (n=11, 10.7%), and lumbar pain (n=10, 9.7%). Ankle sprains (n=4), anterior cruciate ligament (ACL) ruptures (n=3) and concussion (n=3) recorded as the highest sports incapacity injuries. The self-report data collection revealed that 46 (27.2%) athletes arrived at the tournament with an existing self-reported injury/illness and 57 (39.3%) athletes had a self-reported injury/illness at the conclusion of the ANNC (RR 1.44 95%CI 1.05–1.99, p=0.030).

Conclusion: There are no recent studies reporting injury rates specifically in pre-elite netball players. This study found an incidence rate of 89.4 per 1000 player hours. Ankle sprains are the highest medical attention and sports-incapacity injury in pre-elite netball athletes. Foot blisters and low back pain also feature in the highest medical attention injuries and ACL rupture and concussion were high sports incapacity injury/injuries at the ANNC's. Finally, combining both the medical attention and self-report injury/illness data collection methods identified more injuries/illnesses than the use of one method alone.

© 2019 Sports Medicine Australia. Published by Elsevier Ltd. All rights reserved.

Practical implications

- Prevention of ankle sprains and ACL rupture should be a priority. The use of a neuromuscular warm-up such as the Netball Australia 'KNEE' program should be considered.
- Foot blisters are common but can be prevented with load management, appropriate footwear, good hygiene and prophylactic taping.
- Implementation planning of injury prevention interventions must be prioritised.

1. Introduction

* Corresponding author. E-mail address: erin.smyth@ausport.gov.au (E.A. Smyth). Netball is a sport which involves a combination of highintensity short sprints and agility movements, jumping, landing,

https://doi.org/10.1016/j.jsams.2019.10.004

^{1440-2440/© 2019} Sports Medicine Australia. Published by Elsevier Ltd. All rights reserved,

E.A. Smyth, L. Piromalli, A. Antcliff et al. / Journal of Science and Medicine in Sport 23 (2020) 215-221



Fig. 1. The integrated FTEM (Foundations, Talent, Elite, Mastery) framework for the optimisation of sport and athlete development. Image courtesy of the Australian Sports Commission, http://www.ausport.gov.au.

passing and catching.^{1,2} Given the nature of netball, high injury rates are common^{2,4,5} but unlike other sports,^{6–8} current pre-elite injury rates are unknown. Pre-elite athletes, as defined by the FTEM (Foundation, Talent, Elite and Mastery) athlete development framework¹⁰ are 'Talent' level athletes (i.e. Talent 3 and 4) commonly emerging from the junior ranks, who have been formally recognised as having future high performance potential and receive limited financial and service support (Fig. 1). Injury has the potential to prevent pre-elite athletes progressing to be elite.¹¹ The International Olympic Committee (IOC) has released a consensus statement on youth athletic development, outlining challenges to this cohort such as a rapid increase in training load, insufficient rest and recovery, growth and a lack of support services compared to elite athletes.12 Pre-elite athletes often have competing demands between sports, school sport, club and representative commitments with minimal support to manage individual training loads.10

Despite the benefits of regular physical activity, athletes are exposed to injury risk.⁵ There is an economic, physical and psychological cost to injury particularly in pre-elite athletes.¹¹ This may compromise performance resulting in an inability to fulfil full athletic potential with drop out as a possible consequence. This has been demonstrated in Australian track and field athletes where 17.3% of injured athletes retire before the age of 18 due to injury.⁶ It is imperative that governing bodies are aware of these risks so injury prevention interventions can be implemented to minimise these risks.¹³ The 17/U and 19/U ANNC's is a benchmark tournament for developing netball athletes. An increased injury risk is associated with congested matches (multiple matches in a short period of time) and greater match intensity compared to their local competition in other team sports such as football.^{14–16} Consequently, Netball Australia have undertaken thorough injury surveillance at their 17/U and 19/U ANNC to fulfil the first stage of the Translating Research into Prevention Practice (TRIPP) model of prevention.¹⁷

There are no recent studies reporting injury rates specifically in pre-elite netball players. A combined medical attention/sports incapacity injury incidence rate of 11.3 per 1000 player hours over two seasons of community netball in Western Australia has been reported.³ Ankle injuries were most common (53.8%), followed by knee injuries (27.7%) and finger/thumb injuries (26.2%). Hopper et al.¹⁸ reported a medical attention injury incidence of 5.4% across five years of community netball in Western Australia, with ankle injuries being most common (84%). A medical attention injury incidence of 23.8 per 1000 player hours was reported during the three day 1995 NSW State netball championships with knee (14%) and ankle (14%) injuries most common.⁴ In 2009, across three pre-elite tournaments in South Africa a medical attention injury rate of 500.7 injuries per 1000 player hours was reported.⁵ Ankle injuries (34%) were most highly recorded followed by knee injuries (15%).⁵

As demonstrated there are large variations in reported injury rates for the sport of netball. A number of variables could contribute to this such as whether injuries are recorded from tournament play or weekly matches in-season, the data collection methods used, injury definition and study participants. The very high South African injury rate is most likely the result of using a different definition for athlete exposure. Only the minutes played prior to injury were used for this calculation as opposed to minutes played by all players.

A number of injury definitions have been adopted in previous netball epidemiology papers. The injury definitions concept framework (IDCF) describes them as medical attention, athlete self-report or sports incapacity. No previous netball epidemiology study has used athlete self-report despite its benefits for capturing overuse injuries more accurately¹⁹ and obtaining longitudinal data without reliance on medical personnel to capture the data. The Oslo Sports Trauma Research Centre (OSTRC) Health Problems Questionnaire (HPQ)²⁰ has been utilised in this study for this purpose. The authors wanted to calculate whether there was an increase in injury rates across the four weeks following the tournament which has been previously described due to a spike in load.²¹

The aims of this paper are to (1) describe the type, location and severity of injuries and illnesses that occur across the six day 2018 17/U & 19/U ANNC and four weeks following the tournament; (2) determine if the injury profile is different between 17/U & 19/U netball athletes; (3) describe contributing factors to injury; (4) compare percentage of agreement when clinical assessment and self-report injury data collection methods are employed and (5) determine whether exposure to the tournament alters the health of participants in the following four weeks.

2. Methods

The incidence of injuries occurring in the 2018 17/U & 19/U ANNC was prospectively studied in 192 netball athletes across 16 teams (17/U, athletes n = 96; 19/U, athletes n = 96). The tournament took place over six consecutive days in April (17th-22nd). Each team played nine matches with two matches played on days one, two and four, and one match played on days three, five and six. A total of one hundred and forty-four matches were played and matches consisted of four, ten minute quarters. The reporting of this study follows the Strengthening the Reporting of Observational studies in Epidemiology (STROBE) statement.²² Ethics approval was obtained from the Australian Institute of Sport Ethics Committee (Approval number: 20180404). All participants or their parents/guardians provided consent via their Athlete Management System (AMS; Smartabase, Fusion Sport, Brisbane, Australia) account, accessible from personal computers, tablets and smartphones.

Concurrent Injury Definitions Concept Framework (IDCF)²³ definitions were utilised. Clinical examination injuries were defined as any health complaint that required the team physiotherapist to assess, irrespective of time-loss (sports incapacity). Sport incapacity injuries were defined as an injury that required an athlete to miss any duration of match play during the tournament or where the athlete reported a reduction in capacity during the tournament as assessed by the Oslo Sports Trauma Research Centre (OSTRC) Health Problems Questionnaire (HPQ).²⁰

The HPQ asks the athlete four key questions around athlete health. These are:

- 1 Have you had any difficulties participating in normal training and competition due to injury, illness or other health problems during the past week?
- 2 To what extent have you reduced your training volume due to injury, illness or other health problems during the past week?

- 3 To what extent has injury, illness or other health problems affected your performance during the past week?
- 4 To what extent have you experienced symptoms/health complaints during the past week?

All data (clinical examination and athlete self-reported) were collected via a centralised database (Smartabase, Fusion Sport Pty Ltd, Brisbane Australia).²⁴ Team physiotherapists met with the primary author to discuss injury definitions such as new versus recurrent, trauma versus overuse to ensure consistency. Each medical attention injury/illness recorded in the AMS database by team physiotherapists was assigned a four-character Orchard Sports Injury Classification System 10.125 injury diagnosis code, which details the body part and nature of the injury sustained. Additional information was recorded to classify the body side of the injury (i.e. left, right, bilateral, central), date of injury occurrence, training or match environment and the number of days the player was unable to participate both fully and partially in netball training/competition. In addition, the physiotherapist recorded mechanism of injury, court location, position, time of match, number match of the day, tournament day and whether the athlete could continue playing immediately after injury.

Injuries can occur four weeks following a spike in load²¹ therefore athletes were asked to complete the HPQ at the start of the tournament, end of the tournament, one week post tournament and four weeks post tournament. Prevalence measures were calculated for each of the four HPQ's. Prevalence of self-report substantial problems and severity of health problems were calculated according to published methods.^{20,26,27} We chose to report period prevalence rather than point prevalence for this study as it was over a smaller time period and with the reduced response rate for the third and fourth HPQ's, this approach provides a more accurate interpretation of the data. Averages and confidence intervals were calculated by identifying the athletes that reported any health problem across the four questions for any of the four HPQ's (ie, problems where athletes selected option 2, 3, 4 or 5 in any of the questions at any time). This number of athletes were then divided by the total number of participants. The prevalence of substantial problems was calculated by identifying those who had moderate or severe reductions in training volume or moderate or severe reductions in sports performance or complete inability to participate in sport (ie, problems where athletes selected option 3, 4 or 5 in either Ouestion 2 or 3).

Athlete exposure was captured by Netball Australia nominated national selectors using the AMS (Smartabase, Fusion Sport Pty Ltd, Brisbane Australia). The position, including those not playing was recorded for each quarter of every match. Where there was significant missing data, average team exposure was calculated by multiplying the number of athletes (n=12) by the number of teams (n=16) by the number of matches (n=9) by the number of minutes/match (n=40).

Descriptive analysis was performed separately and combined for 17/U and 19/U athletes at the 2018 ANNC's. The difference between specified injury variables for 17/U and 19/U were measured by calculating risk ratio. Due to low sample size, Fisher's exact test was used to test for significance. Injury incidence was calculated using the formula: number of injuries/exposure per 1000 h. Incidence Rate Ratios (IRR) were calculated by dividing the 17/U incidence rate by the 19/U incidence rate to compare the 17/U (referent) and 19/U age groups in terms of total injuries and sports incapacity injuries as well as comparing total new/recurrent injuries and traumatic/overuse injuries. A statistical significance level was set a priori, where 95% confidence intervals did not include 1.0 for analyses including rate ratios or incidence rate ratios. A chi-square test was used to compare injury region and nature. Mechanism of injury, court location, position, time of match, number match of the day and tournament day were not analysed due to a large amount of missing data. Statistical significance was set a priori at p <0.05. Similar injuries were grouped together for analysis eg. Lumbar spine facet joint pain/stiffness (OSICS code: LGXX) and Lumbar muscle trigger points (OSICS code: LMYX) have a similar prognosis and management therefore they were grouped together for better representation of the data. Previous research have shown that clustering of injuries is accurate²⁸ and appropriate from an epidemiological approach.^{29,30} Details of all combined OSICS code are described in Supplement 1.

A generalised estimating equation (GEE) was attempted for analysis of the HPQ data but given the high levels of missing data for the 3rd and 4th HPQ's, prevalence's across periods were used rather than introducing bias via unsubstantiated imputation methods for missing data. The 2nd HPQ data was manually matched to medical attention records from the tournament. Records that appeared in the HPQ and not in AMS were recorded and percentage of agreement was calculated as the percentage of records reported in both. All statistical analyses were performed in Stata (Stata/IC 15.1, StataCorp, USA).

3. Results

The data collected by team physiotherapists revealed a total of 103 medical attention injuries sustained by 80 athletes over a 6 day period at an incidence rate of 89.4 per 1000 player hours when 17/U and 19/U injuries were combined. There was significant missing data for athlete exposure, therefore overall exposure was calculated by multiplying the number of athletes (n=12) by the number of teams (n=16) by the number of matches (n=9)by the number of minutes/match (n = 40). Sports incapacity incidence rate was 19.1 per 1000 player hours. The higher injury rate sustained by the 19/U age group was statistically significantly different (17/U, n = 39; 19/U, n = 64; IRR 1.64 95%CI 1.08-2.51). Lateral ankle ligament sprain (AJLX/AJLA, n = 14, 13.6%), foot blisters (FKBX, n = 11, 10.7%), and lumbar pain (LGXX/LMYX, n = 10, 9.7%) recorded the highest incidence across medical attention injuries in both age groups. There were 22 (21% of total injuries) sports incapacity injuries with ankle sprain (AJLX/AJLA, n=4), anterior cruciate ligament (ACL) rupture (KJAX, n = 3) and concussion (HNCO, n = 3) recorded as the highest sports incapacity injuries. No statistical significant difference was observed between age group with respect to sport-incapacity injury incidence (17/U, n = 9, 8.7%; 19/U, n = 13 12.6%; IRR 0.69 95%CI 0.26-1.75).

New injuries occurred nine times more frequently than recurrent injuries (IRR 9.3 95%Cl 4.84–20.03, P < 0.001). There were more traumatic injuries than overuse injuries (trauma n = 51, overuse n=42; IRR 1.24 95%Cl 0.62–2.49, P=0.522). The body region most commonly injured was the lower limb (χ_3^2 = 91.45, p < 0.001) and joint injuries accounted for nearly one third of all injuries (χ_5^2 = 30.69, p < 0.001) (Supplement 2). Daily prevalence rates could not be calculated due to a significant amount of missing exposure data (31%). Details of mechanism of injury, new v recurrent and trauma v overuse according to each body area are outlined in Table 1.

The response rate declined across the four HPQ's (1st 90%; 2nd 76%; 3rd 41%; 4th 34%). The HPQ revealed that 46 (27.2%) athletes arrived at the tournament with an existing self-reported injury/illness and 57 (39.3%) athletes had a self-reported injury/illness at the conclusion of the ANNC (RR 1.44 95%CI 1.05–1.99, p = 0.030). Twenty-two (34%) athletes were training at a reduced volume 4/52 post nationals due to self-reported injury/illness. Fifty-five (38%) athletes thought their performance was affected by injury/illness during the tournament (Fig. 2). Less 17/U athletes (n = 64, 66.7%) sought medical professionals for their

health problems compared to 19/U athletes (n = 68, 74.7%). There was an increase of moderate severity self-reported injuries in the 19/U group, 4/52 post tournament. The period prevalence of self-reported health problems for the 17/U athletes and 19/U athletes was 55% (n = 53, 95%Cl 45–65%). According to the HPQ data, the 19/U athletes had slightly less substantial health problems (n = 29, 30%; 95%Cl 21–40%) compared to the 17/U athletes (n = 30, 31%; 95%Cl 22–42%).

Seventy tournament self-reported injuries/illnesses were identified by the post-tournament HPQ. Twenty-six percent (n = 18) of these injuries were not recorded in AMS. Twelve of the missing records were overuse injuries and ten were illnesses.

4. Discussion

This study recorded 103 medical attention injuries sustained by 80 athletes during the ANNC's. We observed higher medical attention injury rates in 19/U athletes compared to 17/U athletes however the distribution of injury types was similar across both age groups and the HPQ revealed equal self-reported injury/illness period prevalence during the tournament in the 17/U and 19/U age groups (n = 53, 55%, 95%Cl 45–65%). The main sports incapacity injuries recorded were ankle sprains, ACL rupture and concussion.

In the past 30 years only three studies^{2,4,5} have specifically reported injury rates at tournaments for development netball athletes. The most recent study reporting injury rates at a pre-elite netball tournament conducted a decade ago in South Africa, reporting a medical attention injury incidence rate of 500.7 per 1000 player hours.⁵ However, their definition of exposure was only the game time a player participated before the injury occurred rather than total exposure, resulting in a higher injury incidence rate. Hume and Steel (2000)⁴ reported a medical attention injury rate of 23.8 per 1000 player hours at a state netball tournament was conducted over 20 years ago and the athletes were at a T2 and T3 FTEM level rather than the pre-elite (T3-4) level. The difference in injury rate could possibly be due to the difference in FTEM level. Our study revealed a medical attention injury incidence rate of 89.4 per 1000 player hours when 17/U and 19/U injuries were combined. This is the first publication in a decade to report tournament injury rates for pre-elite netball athletes and fulfils the first stage of the TRIPP¹⁷ framework for sports injury prevention.

We observed the 19/U athletes had a higher medical attention incidence (IRR 1.64). This indicates that 19/U athletes sustained 64% more medical attention injuries compared to the 17/U athletes. However, whilst there was a difference in medical attention incidence between age groups there was no difference between the 17/U and 19/U athletes in terms of sports incapacity, body region injured, nature of injury, new versus recurrent, trauma versus overuse and mechanism of injury as demonstrated by incidence rate ratios and risk ratios. Interestingly, the HPQ identified equal number of injuries/illnesses in the 17/U & 19/U age group. This may just suggest that 19/U athletes were more likely to seek medical attention whereas the younger athletes did not. Future research should explore why the two year age difference results in apparently different medical attention injury rates.

The most frequently recorded medical attention injury diagnoses across both age groups were lateral ankle ligament sprain, foot blisters, and lumbar pain. There were 22 (21% of total injuries) sports incapacity injuries, 41% of these were ankle and foot injuries and 18% were knee injuries. Knee injuries in netball have attracted media attention and thought to be the most common injury which there is some evidence for^{31,32} however these claims are based on hospital admission data. Most netball injuries are managed without hospital admission therefore this research is not a true reflection of injury rates in netball. The vast majority of netball

Table 1

Mechanism of injury, new v recurrent & trauma v overuse for each body area
--

	Body area										
	Ankle	Foot	Trunk ^a	Lower leg	Head & Neck	Клее	Wrist & Hand	Pelvic area ^b	Shoulder	Thigh	Total
Total (n=)	26	19	17	10	6	6	6	5	5	3	103
Mechanism of injury											
Accident/other	0	2	0	0	0	0	0	0	1	1	4
Catching	0	0	0	0	0	0	4	0	1	0	5
Change of direction	4	0	0	0	0	1	0	0	0	1	6
Collision	2	0	1	4	4	0	1	2	2	1	17
Contested landing	7	0	1	0	1	3	0	0	1	0	13
]umping	2	1	0	0	0	0	0	0	0	0	3
Land on another	2	1	0	0	0	0	0	0	0	0	3
Non-contested landing	1	0	1	0	0	0	0	0	0	0	2
Running	1	8	1	5	0	0	0	1	0	0	16
Throw/Passing	0	0	0	0	0	1	1	0	0	0	2
Unknown/missing	7	7	13	1	1	1	0	2	0	0	32
New v recurrent											
New injury	25	16	16	8	4	6	6	4	5	3	93
Recurrent injury	1	3	2	0	2	1	0	0	1	0	10
Trauma v overuse											
Trauma	18	2	2	4	5	5	6	2	4	3	51
Overuse	7	15	11	4	0	1	0	3	1	0	42
Other	1	2	4	2	1	0	0	0	0	0	10

^a Thoracic spine, Chest and Lumbar spine.

^b Buttock, pelvis, hip and groin.



Fig. 2. HPQ results across time for participation (top left), volume (top right), performance (bottom left) and symptoms (bottom right).

epidemiology data has reported ankle sprains to be the most common.^{2,3,5} It is difficult to determine the most common mechanism of injury for the most frequent injuries because as outlined in Table 1 'unknown/missing' data was twice as high as the next most common mechanism 'collision'. Foot blisters can be easily prevented with correct footwear and foot hygiene. The cause of ankle sprains, low back pain, ACL rupture and concussion require further investigation to inform future preventative strategies. Additional strategies to reduce the problem of 'unknown/missing' data could be (1) making AMS data entry fields compulsory to increase data collection for 'contributing factors' or (2) utilising match video footage to record such information.

ACL rupture at the ANNC's is not common. We found three ACL ruptures in a population of 192 athletes resulting in a period prevalence rate of 1.6% chance of rupturing an ACL at the ANNC. This is similar to the 1.8% overall incidence rate reported between

219

1985 and 1989.¹⁸ It has been demonstrated that minor injuries are often precursors to more significant injuries,³³⁻³⁵ perhaps the best strategy for preventing these more significant, but less common injuries is by preventing the precursor injury such as ankle sprain. Injuries to one bodily area have been shown to precede injuries to another area within the same person.33 With recent evidence in Kayak,³⁶ Rugby Sevens,³⁰ Australian Football³⁵ demonstrating this non-causal association. Non-time-loss injuries have been shown to precede time-loss injuries in Australian Football with survival rates (of not incurring a time loss injury) being approximately 50% within the 12 weeks following a lower limb muscle injury.³⁵ Of the three ACL injuries, one athlete developed foot blisters and hip pain at the ANNC prior to their ACL injury. We're not inferring direct causation but we are suggesting that future ACL epidemiology studies need to record minor injuries or aetiological factors that may result in early detection preventative strategies.

Due to discrepancies in reporting, multiple tools are required for quality injury surveillance. The HPQ identified twenty-six injuries/illnesses that were not recorded in AMS, 12 of these were overuse injuries and ten were illnesses. The AMS recorded 33 injuries not identified in HPQ. The HPQ was developed as a way of capturing overuse injury data that is not captured when a 'medical attention' injury definition is used throughout the tournament. The HPQ is also useful for capturing illnesses when physiotherapists are the sole health professional and recording medical attention data as their scope of practice excludes them from diagnosing illnesses in Australia. Using a dual injury/illness capture system may be preferable in this setting to ensure all injury/illness records are captured, resulting in improved injury prevention interventions.

Many netball athletes are starting the ANNC's with injury or illness. Twenty-seven percent of athletes arrived at the tournament with an existing injury/illness. This replicated the finding for U/21 and U/16 netball athletes competing at the 1988 Australian netball championships.² This prevalence has not changed in 30 years and considering that many injury prevention interventions have been trialled over this period suggests that developing effective implementation strategies are still important.¹⁷ A recent systematic review investigating injury prevention strategies in pre-elite athletes has shown that implementation planning does not routinely occur in sport despite it being a key stage of the TRIPP framework.³⁷

A limitation of this research is the varied response rate occurring across the four times the HPQ was administered. While response rates decreased over the tournament, the first and second questionnaire response rates, 88% and 76% respectively, enable robust conclusions to be made.³⁸ We attempted to analyse the data using all responses across the tournament however the missing data limits analysis. To ensure that we didn't introduce potential bias we did not utilise imputation methods as reported in similar research.²⁰ Future research can improve through targeting compliance of the netballers in completing all questionnaires issued over the championship period.

5. Conclusion

There are no recent studies reporting injury rates specifically in pre-elite netball players. This study found an IRR of 89.4/1000 player hours when 17/U and 19/U injuries were combined. Ankle sprains are the highest medical attention and sports-incapacity injury in pre-elite netball athletes. Foot blisters and low back pain also feature in the highest medical attention injuries at the ANNC's. Finally, combining both the medical attention (AMS) and self-report (HPQ) injury data collection methods optimises injury surveillance. Injury prevention efforts for pre-elite netballers should focus on ankle sprains for best effect.

Funding

This study received no funding. This work was supported by a joint AIS and University of Canberra scholarship awarded to the author ES for support during her PhD.

Acknowledgements

The authors thank the team physiotherapists and selectors attending the 2018 Australian Netball National Championships for collection of the data. We would also like to thank the athletes who also attended these championships for entering their own injury/illness data and their coaches for their support for this research.

We would also like to thank the Athlete Management System team, Liam Toohey and Marijke Welvaert at the Australian Institute of Sport (AIS) for their ongoing support.

Appendix A. Supplementary data

Supplementary material related to this article can be found, in the online version, at doi:https://doi.org/10.1016/j.jsams.2019.10. 004.

References

- Cormack SJ, Smith RL, Mooney MM et al. Accelerometer load as a measure of activity profile in different standards of netball match play. Int J Sports Physiol Perform 2014; 9(2):283–291.
- Hopper D, Elliott B. Lower limb and back injury patterns of elite netball players. Sports Med 1993; 16(2):148–162.
- Finch C, Costa AD, Stevenson M et al. Sports injury experiences from the Western Australian sports injury cohort study. Aust N Z J Public Health 2002; 26(5):462–467.
- Hume PA, Steele JR. A preliminary investigation of injury prevention strategies in Netball: are players heeding the advice? J Sci Med Sport 2000; 3(4):406–413.
 Langeveld E, Coetzee FF, Holtzhausen LJ. Epidemiology of injuries in elite
- South African netball players. South Afr J Res Sport Phys Educ Recreation 2012; 34(2):83–93.
- 6. Huxley DJ, O'Connor D, Healey PA. An examination of the training profiles and injuries in elite youth track and field athletes. *Eur J Sport Sci* 2014; 14(2):185–192.
- Loose O, Fellner B, Lehmann J et al. Injury incidence in semi-professional football claims for increased need of injury prevention in elite junior football. *Knee* Surg Sports Traumatol Arthrosc 2018; 27(3):978–984 https://doi-org.ezproxy. canberra.edu.au/10.1007/s00167-018-5119-8.
- Moseid CH, Myklebust G, Fagerland MW et al. The prevalence and severity of health problems in youth elite sports: a 6-month prospective cohort study of 320 athletes. Scand J Med Sci Sports 2017; 28(4):1412–1423.
- 10. Weissensteiner JR. Method in the madness: working towards a viable 'paradigm' for better understanding and supporting the athlete pathway, in *Handbook of Talent Identification and Development in Sport*, Baker J, Cobley S, Schorer J, Wattie N, editors, Abingdon, T&F Routledge, 2017.
- DiFiori JP, Benjamin HJ, Brenner J et al. Overuse injuries and burnout in youth sports: a position statement from the American medical society for sports medicine. *Clin J Sport Med* 2014; 24(1):3–20.
- Bergeron MF, Mountjoy M, Armstrong N et al. International Olympic Committee consensus statement on youth athletic development. Br J Sports Med 2015; 49(13):843–851.
- Fuller C, Drawer S. The application of risk management in sport. Sports Med 2004; 34(6):349–356.
- Bengtsson H, Ekstrand J, Hägglund M. Muscle injury rates in professional football increase with fixture congestion: an 11-year follow-up of the UEFA Champions League injury study. Br J Sports Med 2013; 47(12):743–747.
- Carling C, McCall A, Le Gall F et al. The impact of short periods of match congestion on injury risk and patterns in an elite football club. Br J Sports Med 2016; 50(12):764.
- Wollin M, Pizzari T, Spagnolo K et al. The effects of football match congestion in an international tournament on hip adductor squeeze strength and pain in elite youth players. J Sports Sci 2018; 36(10):1167–1172.
- Finch Č. A new framework for research leading to sports injury prevention. J Sci Med Sport 2006; 9(1):3–9.
- Hopper D, Elliott B, Lalor J. A descriptive epidemiology of netball injuries during competition: a five year study. Br J Sports Med 1995; 29(4):223.
 Clarsen B, Myklebust G, Bahr R. Development and validation of a new method
- Clarsen B, Myklebust G, Bahr R. Development and validation of a new method for the registration of overuse injuries in sports injury epidemiology: the Oslo Sports Trauma Research Centre (OSTRC) overuse injury questionnaire. Br J Sports Med 2013; 47(8):495–502.

- 20. Clarsen B, Rønsen O, Myklebust G et al. The Oslo Sports Trauma Research Center questionnaire on health problems: a new approach to prospective monitoring of illness and injury in elite athletes. Br J Sports Med 2014; 48(9):754–760.
- 21. Drew MK, Finch CF. The relationship between training luad and injury, ness and soreness: a systematic and literature review. Sports Med 2016; 46(6):861-883.
- 22. von Elm E, Altman DG, Egger M et al. The Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) statement: guidelines for reporting observational studies. *Lancet* 2007; 370(9596):1453–1457.
- 23. Timpka T, Jacobsson J, Bickenbach J et al. What is a sports injury? Sports Med 2014: 44(4):423-428
- 24. Toohey LA, Drew MK, Fortington LV et al. An updated subsequent injury categorisation model (SIC-2.0): data-driven categorisation of subsequent injuries in sport. Sports Med 2018; 48(9):2199-2210.
- 25. Rae K, Orchard J. The Orchard Sports Injury Classification System (OSICS) version 10, Clin | Sport Med 2007; 17(3):201-204, 26. Harøy J, Clarsen B, Thorborg K et al. Groin Problems in Male Soccer Players Are
- More Common Than Previously Reported. Am J Sports Med 2017; 45(6)
- Bromley S, Drew M, Talpey S et al. Collecting health and exposure data in Aus-tralian Olympic Combat Sports: feasibility study utilizing an electronic system. JMIR Human Factors 2018; 5(4), e27-e27. 28, Finch CF, Orchard JW, Twomey DM et al. Coding OSICS sports injury diagnoses
- in epidemiological studies: does the background of the coder matter? Br J Sports Med 2014; 48(7):552. Ekegren CL, Gabbe BJ, Donaldson A et al. Injuries in community-level Australian
- football: results from a club-based injury surveillance system. J Sci Med Sport 2015; 18(6):651-655.

- 30. Toohey LA, Drew MK, Finch CF, Cook JL, Fortington LV. A 2-year prospective study of injury epidemiology in elite australian rugby sevens : exploration of incidence rates, severity, injury type, and subsequent injury in men and women. Am J Sports Med 2019; 47(6):1302-1311.
- Kreisfeld R, Harrison J, Tovell A. Hospitol Care for Australian Sports Injury, 2012–13, Canberra, AIHW, 2017.
- 32. Flood L, Harrison JE. Epidemiology of basketball and netball injuries that resulted in hospital admission in Australia, 2000–2004. *Med J Aust* 2009; 190(2):87. Toohey LA, Drew MK, Cook JL et al. Is subsequent lower limb injury associated
- 33. with previous injury? A systematic review and meta-analysis. Br J Sports Med 2017: 51(23):1670.
- Lynall RC, Mauntel TC, Padua DA et al. Acute lower extremity injury rates increase after concussion in college athletes. Med Sci Sports Exerc 2015; 47(12):2487-2492.
- 35. Stares JJ, Dawson B, Peeling Pet al, Subsequent injury risk is elevated above baseline after return to play: a 5-year prospective study in elite Australian football. Am J Sports Med 2019; 47(9):2225-2231.
- 36. Toohey LA, Drew MK, Bullock N et al. Epidemiology of elite sprint kayak injuries: a 3-year prospective study. J Sci Med Sport 2019; 22(10):1108-1113. 37. Smyth EA, Newman P, Waddington G et al. Injury prevention strategies spe-
- cific to pre-elite athletes competing in Olympic and professional sports a systematic review. J Sci Med Sport 2019; 22:887–901.
- 38. Fincham JE. Response rates and responsiveness for surveys, standards, and the Journal. Am J Pharm Educ 2008; 72(2), 43-43.

Unpublished data from the 2018 17/U & 19/U ANNC

Of the 103 medical attention injuries that occurred at the 2018 17/U & 19/U ANNC, the team

physiotherapist did not ask about menstrual cycle 37 times (36%) or the data were missing for 18 (17%)

injury records (Figure 8). Therefore, over half (53%) of the menstrual cycle data were missing and

inference about injury/illness aetiology cannot be made.



Figure 8: Menstrual cycle data collected by team physiotherapists

There were 22 sports incapacity injuries recorded by team physiotherapists across the six-day 2018 17/U & 19/U ANNC. Following ankle sprain, ACL rupture and concussion, foot blisters and foot sprain were the next most frequent sports incapacity injuries (Table 1).

Diagnosis	Sports incapacity
	(n=)
Ankle sprain	5
ACL rupture	3
Concussion	3
Foot blister	2
Foot sprain	2
Thigh strain	1
Thoracic pain	1
Lumbar pain	1
Calf tightness	1
Shoulder subluxation	1
Pec strain	1
Shin pain	1
Total	22

Table 1: Sports incapacity injuries at the 2018 17/U 7 19/U ANNC

Chapter 3 Developing preventative measure (TRIPP 3)

Published work

This chapter is based on a manuscript that has been published in the Journal of Science and Medicine in Sport.

Smyth E, Newman P, Waddington G, et al. Injury prevention strategies specific to pre-elite athletes competing in Olympic and professional sports – a systematic review. *J Sci Med Sport*. 2019; 22:887-901.

Chapter 2 reported the injury rates for pre-elite netball athletes. The most frequently recorded medical attention injury diagnoses across both the age groups were lateral ankle ligament sprain (n=14, 13.6%), foot blisters (n=11, 10.7%), and lumbar pain (n=10, 9.7%). Ankle sprains (n=4), ACL ruptures (n=3) and concussion (n=3) were recorded as being the highest sports incapacity injuries. The self-report data collection revealed that 46 (27.2%) athletes arrived at the tournament with an existing self-reported injury/illness and 57 (39.3%) athletes had a self-reported injury/illness at the conclusion of the ANNC (RR 1.44 95%, CI 1.05–1.99, p=0.030). New injuries occurred nine times more frequently than recurrent injuries (IRR 9.3 95%, CI 4.84–20.03, p<0.001). There was no significant difference between the overall number of traumatic (n=51) and overuse (n=42) injuries sustained in the tournament (IRR 1.24 95%, CI 0.62–2.49, p=0.522). It is difficult to determine the most common mechanism of injury for the most frequent injuries because, as outlined in Table 1, 'unknown/missing' data was twice as high as the next most common mechanism, 'collision'. These results are similar to Hume & Steele⁵⁸ and highlights the need to prioritise mechanism of injury data collection. Perhaps reviewing match play video would result in more accurate data collection.

Following injury surveillance and establishment of aetiology and mechanism of injury, the next step is to develop a preventative measure for the priority injuries. The Intervention Development Process (IDP) has been created to guide intervention development in the sporting environment.⁶⁷ The first IDP step is to 'use the research evidence and clinical experience'; therefore, the following systematic review helps

inform TRIPP Stage 3 (develop preventive measure). A review of injury prevention strategies has not previously been conducted specifically for pre-elite athletes.

Co-authors' declaration

Declaration of Co-Authored Publications

For use in theses which include co-authored publications. This declaration must be completed for each coauthored publication and to be placed at the start of the thesis chapter in which the publication appears, or as a preface to the thesis.

UNIVERSITY OF

Declaration for Thesis Chapter Three

Publication: "Injury prevention strategies specific to pre-elite athletes competing in Olympic and professional sports - A systematic review"

DECLARATION BY CANDIDATE

In the case of Chapter Four, the nature and extent of my contribution to the work was the following: Nature of Contribution Extent of Contributions (%)

Conception and design of the review	70%
Development and application of search strategy and selection criteria	50%
Completion of risk of bias assessment	50%
Extraction, analysis and interpretation of data	50%
Writing the paper and critical appraisal of content	70%
Corresponding author for journal communication and publication peer-review process.	100%

The following co-authors contributed to the work:

Name	Nature of Contribution	Contributor is also a UC student (Yes/No)
Gordon Waddington	Conception and design of the review Writing the paper and critical appraisal of content	No
Phil Newman	Extraction, analysis and interpretation of data Writing the paper and critical appraisal of content	Νο
Juanita Weissensteiner	Writing the paper and critical appraisal of content	No
Michael Drew	Conception and design of the research proposal Extraction, analysis and interpretation of data Completion of risk of bias assessment Writing the paper and critical appraisal of content	No

Candidate's Signature

___17/09/2021____ Date

DECLARATION BY CO-AUTHORS

The undersigned hereby certify that:

- the above declaration correctly reflects the nature and extent of the candidate's contribution to this work, and the nature of the contribution of each of the co-authors.
- (2) they meet the criteria for authorship in that they have participated in the conception, execution, or interpretation, of at least that part of the publication in their field of expertise;

Declaration of Co-Authored Publications



(3) they take public responsibility for their part of the publication, except for the responsible author who accepts overall responsibility for the publication;

- (4) there are no other authors of the publication according to these criteria;
- (5) potential conflicts of interest have been disclosed to (a) granting bodies, (b) the editor or publisher of journals or other publications, and (c) the head of the responsible academic unit; and
- (6) the original data are stored at the following location(s) and will be held for at least five years from the date indicated below:

[Please note that the location(s) must be institutional in nature, and should be indicated here as a department, centre or institute, with specific campus identification where relevant.]

Location(s): Australian Institute of Sport

Phil Newman	Reema
Signed:	
	Coll and a fair fair fair

Date: 11/09/2021

Gordon Waddington

Signed:	202	45-
Date:	17/09/2021	

Juanita Weissensteiner Signed: Juanita Weissensteiner

Date: 30/08/2021

Michael Drew	
Signed:	Michael Mass
Date:	17/09/2021

Injury prevention strategies specific to pre-elite athletes competing in Olympic and professional sports – a systematic review

Journal of Science and Medicine in Sport 22 (2019) 887-901



Review

Injury prevention strategies specific to pre-elite athletes competing in Olympic and professional sports – A systematic review



Erin A. Smyth^{a,b,*}, Phillip Newman^{b,d}, Gordon Waddington^{a,b}, Juanita R. Weissensteiner^e, Michael K. Drew^{a,c}

^a Athlete Availability Program, Australian Institute of Sport, Australia

^b Research Institute for Sport and Exercise, University of Canberra, Australia

^c Australian Collaboration for Research into Injury in Sport and its Prevention (ACRISP), Australia

^d Faculty of Health, University of Canberra, Australia

* Sport Development Group, New South Wales Office of Sport, Australia

ARTICLE INFO

Article history: Received 12 July 2018 Received in revised form 17 December 2018 Accepted 6 March 2019 Available online 12 March 2019

Keywords: Athletic injuries Adolescent Prevention & control Athletes Sports

ABSTRACT

Objectives: To describe and evaluate injury prevention interventions for pre-elite athletes who compete in an Olympic or professional sport. Design: Systematic review.

Methods: This review was prospectively registered (PROSPERO CRD42017065083) and a systematic electronic search was conducted in May 2017. The following inclusion criteria were applied: (1) studies including and analysing data specific to pre-elite athletes (determined by the T3/T4 levels of the FTEM model); (2) featured injury prevention interventions; (3) provided sufficient data related to injury such that the effect can be analysed e.g. injury rates, incidence, prevalence, injury rate ratios; (4) featured randomised and non-randomised controlled trials or prospective cohorts.

Results: A total of 13,480 articles were retrieved with 121 titles identified and 11 studies satisfying the inclusion criteria. No studies demonstrated a low risk of bias. Four different interventions were identified: exercise (n = 7, 64%), psychological (n = 2, 18%), equipment (n = 1, 9%), nutrition (n = 1, 9%). Of the seven exercise interventions, four showed a protective effect and three found no significant effect, providing conflicting evidence. Caution is advised due to high risk of bias, low intervention reporting and minimal evidence for implementation planning in all seven studies.

Conclusions: There is limited evidence from level 2 and 3 studies suggesting exercise and psychology interventions may prevent injury in pre-elite athletes. There is an absence of evidence to support the use of equipment and nutrition interventions in pre-elite athletes. There is a need for quality research designs confirming the clinical impact of existing injury prevention interventions for pre-elite athletes. © 2019 Sports Medicine Australia. Published by Elsevier Ltd. All rights reserved.

1. Introduction

Injuries reduce the chance of successful performance and progression by sportspeople.¹ There is a plethora of injury prevention research for adult, youth and elite athletes.^{2–4} To date, there has not been a systematic review of intervention studies for injury prevention in pre-elite athletes who compete in an Olympic or professional sport. Pre-elite athletes, as defined by the FTEM Athlete Development framework⁵ (Foundation, Talent, Elite and Mastery) are 'Talent' level athletes (i.e., Talent 1–4) that are commonly emerging

Corresponding author.
 E-mail address: erin.smyth@ausport.gov.au (E.A. Smyth).

from the junior ranks and who have been formally recognised as having future high performance potential by their respective state or national level sporting organisation, state academy or institute, university or professional club and receive limited financial and service support but sit below a senior elite level. Recognised by the International Olympic Committee's recent consensus statement on youth athlete development,⁶ the FIEM framework is utilised exten sively within Australia, Japan and Switzerland to review and refine strategy, practice and support specific to the sports pathway considerate of its foundational, pre-elite (talent) and elite components.

At a T1 level, an athlete demonstrates their initial potential through formal and informal talent identification processes. At a T2 level an athlete's holistic talent potential is confirmed by immersing them into realistic training and competitive contexts. Following

https://doi.org/10.1016/j.jsams.2019.03.002

1440-2440/© 2019 Sports Medicine Australia. Published by Elsevier Ltd, All rights reserved.

this level, an athlete is then embedded into a dedicated developmental period where they are provided with quality coaching, daily training environments, education, service support and competition exposure. Transition to the final talent level T4, signals that the athlete has achieved a breakthrough performance and that they are soon ready to progress to a senior elite level in their chosen sport (i.e., E1 level). A podium performance in the highest levels of elite competition equates to an E2 (Elite 2) level and multiple podium achievement at this level an M (Mastery) level. Specific to the preelite athlete population (i.e., FTEM T levels) it is well recognised that the onset of puberty and rate of biological maturation can vary greatly between athletes within and across sports.⁷ This variation in maturation and often an increase in workload, insufficient recovery and lack of support services creates a challenging environment for coaches to develop their athletes while keeping them injury and illness free.⁶ Previous injury predisposes an athlete to further injuries of the same nature8 or a different site and nature,9 To prevent injuries in elite and professional athletes, it is plausible that preventive activities should commence at the pre-elite level. Preventing the index injury (the first injury) will potentially increase the long term health of athletes as well as maximising the pool of athletes eligible to compete at an elite and professional setting. This premise is supported by the work of Huxley et al.¹⁰ demonstrating that 17.3% of injured pre-elite track and field athletes retire due to injury prior to turning eighteen years of age.

This systematic review will inform practitioners working with pre-elite athlete populations in three ways:

- Describe and assess injury prevention interventions trialled in pre-elite populations.
- Synthesise the available evidence on injury prevention strategies used in pre-elite athletes.
- Provide future directions and opportunities for future research in pre-elite injury prevention

2. Methods

The Preferred Reporting Items for Systematic Reviews and Metaanalyses (PRISMA) guidelines¹¹ were followed and the protocol was registered on the PROSPERO International prospective register for systematic reviews website (http://www.crd.york.ac.uk/ PROSPERO) on 9 May 2017, with the following registration number: CRD42017065083.

The inclusion criteria were: (1) studies that separately analysed pre-elite athletes who are formally recognised and supported by their respective National Sporting Organisation or professional club as having future high performance potential who compete in Olympic and professional sports; (2) injury prevention intervention specific to pre-elite athletes (as determiend by the T3 and T4 levels of the FTEM model) competing in Olympic and professional sports; (3) studies provide sufficient data related to injury such that the effect can be analysed, e.g. injury rates, incidence, prevalence, injury rates ratios; (4) study design of randomised controlled trial, prospective cohort studies and non-randomised controlled trials. Exclusion criteria were: (1) elite athletes e.g. athletes who compete in an Olympic sport and represent their country at key international events including the Olympics, World Championships etc. (2) Professional athletes e.g. elite athletes who compete at the highest professional level in their sport nationally (e.g. Australian Football League for Australian Football) and/or internationally (e.g. ATP World Tour in tennis). (3) Recreational athletes that are not formally recognised and supported as emerging pre-elite athletes by their respective national sporting organisation; (4) study designs of case reports, case series and any cross-sectional designs.

A comprehensive electronic search of the literature in MEDLINE, CINAHL, PubMed, SportDISCUS and Web of Science was conducted on 23 May 2017 with a date restriction of 2000 onwards and only English language as a limitation. Details of the PubMed search strategy is included as a supplement. All potential references were imported into Endnote X7 (Thompson Reuters, Carlsbad, California, USA) and then into Covidence (Covidence systematic review software, Veritas Health Innovation, Melbourne, Australia. Available at www.covidence.org) and duplicates removed. Titles and abstracts were independently screened for eligibility by two authors (ES and PN). The full text of each included study was then independently examined according to the inclusion and exclusion criteria, and reference lists of included studies were manually searched to find any further articles for inclusion. In the event of disagreement between authors, additional authors (GW and JW) were consulted to reach consensus.

Data from included studies were extracted and collated by one reviewer (ES) and confirmed by another (MD). The following data were extracted and collated: (1) demographics, FTEM level¹², outcome measures, working injury definition, injury definitions concept framework (IDCF)¹³ (2) study design, data collection period, statistical analysis, effectiveness, level of strength, level of evidence (OCEBM)¹⁴ (3) intervention summary, co-intervention, reproducibility and intervention reporting standard.

Where person-time exposures were not explicitly reported we documented the number of athletes in each group and assumed the intervention time in each group was identical. The key outcomes were presented in a forest plot using incidence rate ratio (JRR) as the comparator. Incidence rates reported in each study were used to calculate the JRR, where incidence rates were not presented they were determined by using raw data.

Intervention Mapping facilitates effective health promotion program planning, implementation and evaluation. IM (Step 5) can be used independently of other IM steps and focuses on planning program adoption, implementation and maintenance.¹⁵ This practical implementation planning protocol has been previously used in a sporting environment with good effect.¹⁶ Each included article was mapped according to IM (Step 5) to determine the risk of bias in implementation. To be judged as low risk of bias the authors needed to explicitly state who the adopters, implementers and planners were as well as outline the implementation performance & change objectives, determinants, strategies and design. A judgement of unsure was given when some information was provided with insufficient detail. A high risk of bias was recorded when no detail was provided for that task.

Two reviewers (ES and MD) independently conducted a risk of bias assessment on each selected article using the Cochrane Collaboration's tool for assessing the risk of bias in randomised trials.¹⁷ If at least one of the criteria was rated as high, the trial was considered to have a high risk of bias. To be considered as a low risk of bias, all criteria had to be rated low risk. Any trials not meeting these criteria were rated unclear.

The level of evidence of interventions for the prevention of injury in the pre-elite athlete was evaluated using previously published guidelines.¹⁸ The level of evidence was defined as strong: provided by two or more studies with a low risk of bias and by generally consistent findings in all studies (>75% of the studies reported consistent findings); moderate: provided by one study with a low risk of bias and/or two or more studies with a high risk of bias, and by generally consistent findings in all studies(>75% of the studies reported consistent findings); limited: provided by only one study with a high risk of bias; conflicting: inconsistent findings among multiple trials (>75% of the studies reported consistent findings).¹⁸

The Oxford Centre of Evidence-based Medicine (OCEBM) — Levels of Evidence was used to determine the hierarchical levels of evidence according to the type of research question with the highest level of evidence. The highest level (Level 1) is a systematic

888

E.A. Smyth et al. / Journal of Science and Medicine in Sport 22 (2019) 887-901



Fig. 1. PRISMA flow chart, PRISMA: Preferred Reporting Items for Systematic Reviews and Meta-Analyses.

review with specific criteria, and the lowest level (Level 5) is 'mechanism-based reasoning'.¹⁴

The IDCF classification was used to assist with injury definition. It is a system for categorising injuries and illnesses according to clinical examination (injury or disease), athlete selfreport (trauma or illness) and/or sports performance (incapacity or sickness).¹³ This is a classification system derived from World Health Organisation's (WHO) framework: International Classification of Functioning, Disability and Health (ICF) but adapted for sport impairment.¹⁹

The reporting standard of exercise intervention was assessed using the Consensus on Exercise Reporting Template (CERT): explanation and elaboration statement.²⁰ This template has been designed to evaluate the reporting of exercise programmes across all study designs used in exercise research. It has previously been utilised in similar studies where the authors used the following criteria: studies satisfying >75% of criteria were considered to have a high level of reporting standard; 60–74% moderate; and those satisfying less than 60% of criteria were considered to have poor exercise reporting standards.²

The 'Workgroup for Intervention Development and Evaluation Research' (WIDER) recommendations provide a framework to identify and provide detailed reporting of the essential components of behaviour change interventions in order to facilitate replication, further development, and progression of the interventions.²¹ This 20-item checklist was used to evaluate the standard of reporting for any psychological interventions. The following criteria were applied: studies satisfying >75% of criteria were considered to have a high level of reporting standard; 60–74% moderate; and those satisfying less than 60% of criteria were considered to have met minimal reporting standard.²

Results of randomised controlled trials were considered eligible for meta-analysis if all the following criteria were met: (1) athlete populations were similar or comparable; (2) the same outcome measures were reported (3) studies demonstrated a low risk of bias. Where three or more studies examined the same intervention using an equivalent summary statistic, a meta-analysis was considered.²²

In the circumstances where it was not appropriate to pool the results due to clinical heterogeneity (varying athlete population, outcome measure, high risk of bias), the results were ordered systematically via a forest plot without summary estimates as recommended in the PRISMA statement.²² The results were grouped according the type of injury prevention intervention or strategy utilised in the study.

3. Results

The electronic search identified 13,480 records; duplicates were then removed resulting in 9831 articles. The titles and abstracts were screened which reduced the number of articles to 121. The full texts of these articles were obtained and assessed for eligibility against the inclusion criteria. No articles were added following citation checking. One hundred and ten studies were excluded due to not meeting the required inclusion criteria. The two most common reasons for exclusion were (1) the patient population was not deemed to be of a pre-elite level (n=70) and 2) the study did not feature an intervention (n = 27). The remaining eleven articles were included for full review and data synthesis.^{23–33} A third author (JW) was consulted to reach consensus regarding study population. The flow chart of this process is presented in Fig. 1.

Table 1 presents the characteristics of the eleven studies including intervention type, participant characteristics, FTEM level, injury incidence rate & ratios, injury definition and injury definitions concept framework (IDCF). Study design, data collection period, statistical analysis, effectiveness, level of strength and level of evidence are provided in Table 2. A summary of the intervention is outlined in Table 3 as well as duration, frequency, compliance, specifics of intervention, co-intervention, reproducibility and reporting standards.

One article introduced new equipment as their injury prevention intervention,²³ seven studies looked at exercise as an injury

889

Table 1

Intervention, description of the study participants, level of competition, incidence of injury of included studies, injury definition and injury definitions concept framework (IDCF).

			Injury incidence				
Intervention; article	Participants (male/female)	FTEM level, age $(Age \pm SD)$	Intervention (exposure)	Control (exposure)	IRR (95%CI)	Working injury definition	IDCF
Equipment Barbic et al. ²³ Comparison of mouth guard designs and concussion prevention in contact soorts.	Football (M) Rngby (M&F)	FTEM: T 3/4, Con: 20.9 ± 1.9	0.95/1000 athlete days (75 days × 308 = 23 256 athlete days)	0.90/1000 athlete days (75 days × 306 = 23 408 athlete days)	1.05 (.58-1.92)	As per American Academy of Neurology Concussion Guidelines	Sumb in 1900'
	Canadian inter-university sport	Int: 20.8 ± 2.1	22/308 p'pants (22/23 256)	21/306 p'pants (21/23 408)			
Exercise							
Gilchrist et al. ²⁴	Soccer (F) NCAA Div 1	FTEM: 19.88	Total knee injury: 1.1/1000 AE	Total knee injury: 1,1/1000 AE	1.04 (0.95-1.13)	Injury to the area about the knee occurring in a game, practice or conditioning activity that required medical care by ATC or Physician, and caused one or more missed	entre de la constante de la co
A randomised controlled trial to prevent non-contact Anterior Cruciate Ligament injury in female collegiate soccer players			Practice ACL: 0.0/1000 AE	Practice ACL: 0.2/1000 AE	0.0	uays of training	
Socce players			History of ACL, non-contact ACL:	History of ACL, non-contact ACL:	0.0		
			0.0/1000 AE Late in season: 0.0/1000 AE	0.1/1000 AE Late in season: 0.0/1000 AE	NC		dama and report.
Junge et al. ²⁷	Soccer (M)	FTEM: T 3/4	High skill total injuries: 6.35/1000 h	High skill total injuries: 6.78/1000 h	0.94 (0.84–1.04)	Any physical complaint caused by soccer that lasted for more than 2 weeks or resulted in absence from a subsequent match or training consider	dan bertak
Prevention of soccer injuries: a prospective intervention study in	High skill (HS)& low skill (LS) teams	Con HS: 15.6 ± 0.86				cianing session	
youur amateur piayers	Switzerland	Int HS: 16.2 ± 1.18					

							Anne stape
Owoeye et al. ²⁰	Soccer (M)	FTEM: T 3/4	All injuries: 0.8/1000 h	All injuries: 1.5/1000 h	0.53 (0.23-1.26)	Injuries that resulted in players being unable to fully participate in subsequent football	Carrier Carrier
Efficacy of the FIFA 11+ warm-up programme in	Premier league, Lagos junior league	Con: 17.80±0.94	Overuse injuries: 0.0/1000 h	Overuse injuries: 0.2/1000 h	0.0	training sessions or matches	
male youth lootball,		Int: 17.49 ± 1.10	Acute injuries:	Acute injuries:	0.69 (0.26-1.75)		
			0.9/1000 h Mild injuries: 0.2/1000 h	1.3/1000 h Mild injuries: 0.4/1000 h	0.5 (0.06-1.81)		
Scase et al. ³⁰	Aus Football (M)	FTEM: T3	All injuries: 26.33/1000 h	All injuries: 38,51/1000 h	0.69 (0.66–0.73)	Any incident occurring during a football game or training session that resulted in a player missing one or more elite competition games	Energister
Teaching landing skills in elite junior Australian football: evaluation of an injury prevention strategy	Elite U18 National Football comp	Con: 17.0±2.6				<u>P</u>	
		Int: 17 ± 2.5					
Silvers-Granelli et al. ³²	Soccer(M)	FTEM: T 3/4	Div 1 ACL's: 0.114/1000 AF	Div 1 ACL's: 0.317/1000 AE	0.36 (0.29-0.45)	No injury definition	No injury definition
Does the FIFA 11+ Injury prevention program reduce the incidence of ACL injury in male soccer players?	NCAA Div I and II	Con: 21 ± 1					
		Int: 20 ± 2					

45

E.A. Smyth et al. / Journal of Science and Medicine in Sport 22 (2019) 887-901

Table 1 (Continued)

			Injury incidence				
Intervention; article	Participants (male/female)	FTEM level, age $(Age \pm SD)$	Intervention (exposure)	Control (exposure)	IRR (95%CI)	Working injury definition	IDCF
							Arraye caliman
Silvers-Granelli et al. ³¹	Soccer (M)	FTEM: T 3/4	Div 1 Total injuries: 9.3/1000 AE (198/21248 AE)	Div 1 Total injuries: 16.05/1000 AE (355/22112 AE)	0.58 (0.49-0.69)	Any physical complaint sustained by a player that resulted from a	Tary Correction Correc
Efficacy of the FIFA 11+ injury Prevention Program in the Collegiate male soccer player	NCAA Div I and II	Con: 21 ± 1	Div 1 Game injuries: 18.83/1000 AE	Div 1 Game injuries: 29.36/1000 AE	0.64 (0.6-0.68)	football match or football training, irrespective of the need for medical attention or	
maie societ prayer		Int: 20 ± 2	Div 1 Practice: 5.146/1000 AE	Div 1 Practice: 10.13/1000 AE	0.51 (0.49-0.53)	time loss from football.	
Strength training reduces injury rate in elite young soccer players during one season	Tunisian player development program	13-14ya					
Psychological							
ivarsson et al. ²⁵	Soccer (M&F)	FTEM: T3	Total injuries: 7/21	Total injuries: 12/20	0.56 (0.46-0.67)	A condition that	(Campo)
It pays to pay attention: a mindfulness based program for injury prevention with soccer players	Swedish junior elite	16.97±0.79	181 days(7/3801)	181 days (12/3620)		participation in a soccer practice or game and resulted in 4 days or more of restricted or no practice including	Character and Character
projets			1.84/1000 athlete days	3.31/1000 athlete days		the day of injury	
johnson at al. ²⁶	Soccer (M&F)	FTEM:	Total injuries: 3/13 p'pants	Total injuries: 21/16 p'pants	0.18 (0.15-0.21)	Caused him or her to miss practice or	American
Injury prevention in Sweden: helping soccer players at risk	Swedish regional level teams	Male: 22.9	150 days (3/1950)	150 days (21/2400)		competition, or to substantially modify participation for at	
più) ers ac risa		Female: 20.1	1.54/1000 athlete days	8.75/1000 athlete days		least one day	karaala
Nutrition							
Lewis et al. ²⁸	Swimmers & Divers (M&F)	FTEM:	Total injuries: 9/19 p'pants	Total injuries: 4/13 p'pants	1.54 (1.27-1.87)	Bone, connective tissue and muscle injury not	Thirty
The effects of season-long Vitamin D supplementation on collegiate swimmers and	NCAA Div I	Con: 19±1.1	182 days (9/3458)	182 days (4/2366)		caused by acute trauma	este Carvester Carvester
divers		Int: 19+1.6	2 60/1000 athlete days	1.69/1000 athlete days			

892

FTEM: foundation, talent, elite, mastery framework, IRR: incidence rate ratio, IDCF: injury definitions concept framework, FIFA: Fédération Internationale de Football Association, NCAA: The National Collegiate Athletic Association, AE: athlete exposure, ACL: anterior cruciate ligament.

E.A. Smyth et al. / Journal of Science and Medicine in Sport 22 (2019) 887-901

Table 2 Characteristics of the studies including data collection period, statistical analysis, efficacy, level of evidence.

Intervention; article	Study design	Data collection period	Statistical analysis	Effectiveness/level of strength for intervention type	Level of evidence {OCEBM level}
Equipment					
Barbic et al. ²³	Randomised controlled trial	September 3–November 17, 2003	Intention-to-treat analysis	Х	Level 2
Comparison of mouth guard designs and concussion prevention in contact sports			Injury rates between groups were compared using a P value and odds ratio		
prevention in contact sports.			Tuto -	Level of strength: Limited	
Exercise					
Silchrist et al. ²⁴	Randomised controlled trial	Fall 2002 NCAA season	Z statistic for rate ratios.	\checkmark	Level 2
A randomised controlled trial to prevent non contact Anterior Cruciate Ligament injury in female collegiate soccer players			As-treated analysis		
Junge et al. ²⁷	Prospective controlled intervention study	Two seasons of Swiss football: 1999 & 2000	Incidences of injury compared by calculating z-values.	High-skill:	Level 3
Prevention of soccer injuries: a prospective intervention study in youth amateur players			Differences between groups examined using t-tests.	Х	
Gwoeye et al. ²⁰	Cluster randomised controlled trial	Six months: 2012/2013 season	Intention to treat principles.	Х	Level 2
Efficacy of the FIFA 11+ warm-up programme in male youth football.		·	lnjury rate ratios calculated using Poisson regression analysis		
Scase et al. ³⁰	Prospective non-randomised controlled trial	U18 Australian football.	Injury incidence.	√	Level 3
Teaching landing skills in elite junior Australian football: evaluation of an injury prevention strategy		2 seasons: 2002/2003	Independent tests/Mann-Whitney U test used for comparison.		
			Cox proportional hazards regression. Relative risk and 95% Cl		
Silvers-Granelli et al, ¹²	Prospective cluster randomised controlled trial	August-December 2012	Frequency counts, t-tests, chi-square tests, factorial analysis of variance, logistic regression tests. Injury rates.	4	Level 2
Does the FIFA 11+ Injury prevention program reduce the incidence of ACL injury in male soccer players?					
Silvers-Granelli et al. ³¹	Randomised controlled trial	Angust-December 2012	T tests, X ² tests and generalised linear regression models	\checkmark	Level 2
Efficacy of the FIFA 11+ injury Prevention Program in the Collegiate male soccer player			-		
Zouita et al. ³³	Randomised controlled trial	One soccer season. October to March	Mean±SD. Kolmogorov-Smimov tests. 2-way repeated-measures analysis of variance. Bonferroni post-hoc analysis. Chi segura test	✓	Level 2
Strength training reduces injury rate in elite young soccer players during one season			יישטעריוטע מומויצאא, כדוו-אקטמוע נפגר,		
Psychological Ivarsson et al. ²⁵	Randomised	August 2013–June	Mann-Whitney U test.	x	Level 2
It pays to pay attention: a mindfulness based program for injury prevention with soccer players	controlled trial	2014	Cohen's d effect size, 80% confidence interval	Λ	LEVEL Z

894

E.A. Smyth et al. / Journal of Science and Medicine in Sport 22 (2019) 887-901

Table 2 (Continued)

Intervention; article	Study design	Data collection period	Statistical analysis	Effectiveness/level of strength for intervention type	Level of evidence (OCEBM level)
Johnson et al. ²⁶	Randomised controlled trial	January-June	Mann Whitney U test	~	Level 2
Injury prevention in Sweden:					
helping soccer players at risk				Level of strength: moderate	
Nutrition					
Lewis at al. ³⁸	Randomised, placebo-controlled trial	August-March	Pearson correlations.	Х	Level 2
The effects of season-long Vitamin D supplementation on collegiate swimmers and			Independent t tests. Linear regression model		
divers				Level of strength: limited	

OCEBM: Oxford centre for evidence based medicine, FIFA: Fédération Internationale de Football Association, NCAA: The National Collegiate Athletic Association.

prevention intervention,^{24,27,29–33} two examined the use of psychological interventions^{25,26} and one was a nutrition intervention.²⁸ The equipment study found the 'WIPSS Brain-pad mouth guard' not to prevent concussions.²³ Five of the exercise interventions looked at whether a specific warm-up could prevent injuries in soccer/football players.^{24,27,29,31,32} The other two exercise intervention programs provided training in developing landing, falling & recovery skills and strength training.^{30,33} Mindfulness, acceptance & commitment approach was used in one of the psychological interventions,²⁵ and one-on-one psychology training was used in the other.²⁶ The nutrition study provided vitamin D supplementation as their injury prevention intervention.²⁸

Eight of the eleven articles studied football players.^{24–27,29,31–33} Five of these were conducted to help develop or assess the FIFA 11+ program developed by Fédération Internationale de Football Association (FIFA).^{24,27,29,31,32} The remaining three investigations studied Canadian football and rugby players,²³ Australian football players³⁰ and swimmers & divers.²⁸

The definitions and IDCF classification for each study are outlined in Table 1. A variety of injury definitions were used across the included studies with most studies using a sports incapacity (time loss) definition. According to the IDCF classification, seven of the studies definitions were based on sports performance,^{25–27,29,30,33} two used clinical examination as their definition,^{23,28} one study used a definition that included both sports performance and clinical examination,²⁴ one study used athlete self-report³¹ and one study did not define injury.³²

The risk of bias assessments are outlined in Fig. 2. Risk of bias assessment concluded that none of the included trials were low risk, ten (91%) were high risk, and one (9%) was unclear risk.¹⁷ A major contributor to assessments of high risk of bias was lack of blinding and allocation concealment.

The level of strength according to guidelines previously published ¹⁸ and effectiveness based on IRR are outlined in Table 2. The strength of evidence was deemed to be moderate for exercise and psychological intervention and limited evidence for the use of equipment and nutrition interventions for injury prevention in pre-elite athletes. The hierarchical level of evidence was Level 2 for most of the intervention studies except for two which were Level 3 because they were non-randomised controlled trials.^{27,30}

Outcomes with incident rate ratios are presented in Table 1. The use of a WIPSS Brain-pad mouth guard was shown to be ineffective for preventing concussion in 19–23 yo male and female Rugby and Football players.²³ The 'Prevent Injury and Enhance Performance' (PEP) program was trialled with 19 yo female soccer players. It

was found to be ineffective for "all knee injuries" but was effective at reducing the number of ACL injuries sustained in practice and for reducing the number of non-contact ACL injuries in athletes who had previous ACL injury.²⁴ The Swiss study, using 15-17 yo male soccer players which was conducted as part of the FIFA 11+ development did not prove to be effective for the pre-elite portion of the population.²⁷ Of the three studies that implemented the FIFA 11+,^{29,31,32} there were favourable outcomes for reduction of overuse injuries,²⁹ reduction of ACL injuries in pre-elite athletes,³² reduction of total injuries, game injuries and practice injuries in pre-elite athletes.31 These studies used 16-18 yo male soccer players²⁹ and 18-22 yo male soccer players.^{31,32} Strength training in 13-14 yo male soccer players,33 landing skills training in 15-19 yo male Australian Rules football players³⁰ and psychological training in 16–17 yo male and female soccer players^{25,26} and 23 yo male and 20 yo female soccer players²⁶ were also shown to be effective. Vitamin D supplementation for 18-20yo male and female swimmers and divers was ineffective for injury prevention.²⁸ One of the FIFA 11+ studies reported a significant (IRR 0.59 95%Cl 0.40-0.86) reduction in injuries for the intervention group but our calculations did not yield the same result (IRR 0.53 95%Cl 0.23-1.26).²⁹ This is most likely due to a difference in statistical approach, Owoeye²⁹ et al. reported using Poisson regression analysis but we adopted a random-effects inverse-variance model.

Reporting standards were also assessed. It was not possible to identify a reporting standard for equipment and nutrition intervention studies. Consensus on Exercise Reporting Template (CERT) and Workgroup for Intervention Development and Evaluation Research (WIDER) scores were determined for exercise and psychological interventions respectively. Intervention details and reporting standard scores are displayed in Table 3. The CERT scores ranged from 26 to 74% for the exercise interventions. Five were deemed to be of poor standard ^{24,27,29,30,33} and two were to be of moderate standard.^{31,32}

The WIDER scores were 30 and 65% for the psychology interventions. One was of poor standard,²⁶ and one was of moderate standard.²⁵ The article of poor standard gave no detailed information about the intervention content and no information regarding the control group.

There was no evidence of the use of IM (Step 5) to guide the implementation process for any of the included articles. Results are shown in Fig. 3. Adopters and implementers were identified in 45% (5/11) of the intervention studies.^{24,26–28,33} Details of activities and resources used to improve reach, adoption and implementation of their injury prevention intervention programs were reported in

Table 3

Intervention summary, duration and frequency, specific details of intervention (ie reps, sets, intensity, progression), co-intervention, reproducible, intervention reporting standard.

Intervention; article	Intervention summary	Duration & frequency	Compliance levels	Specifics of intervention (reps, sets, intensity, progression)	Co-intervention	Reproducible	Intervention repor standard (e.g. CER	ting T}
Equipment Barbic et al. ²³ Comparison of mouth guard designs and concussion prevention in contact sports.	Use of WIPSS Brain-pad mouth guard	Not reported	Control: 73.8% (range by team, 68.1%-97.8%)	NA	Nil	Yes. WIPSS Brain pad available commercially	NA	NA
			l'vention: 69.6% (range by team, 68.4%-78.9%)					
Exercise							CERT score (%)	Interpretation
Gilchrist et al. ²⁴ A randomised controlled trial to prevent non contact Anterior Cruciate Ligament injury in female collegiate soccer players	Prevent injury and enhance performance (PEP) program. Stretching, strengthening, plyometrics, agilities and avoidance of high-risk positions.	12 weeks, 1–3/week	Average: 25.8 PEP sessions/team (2.15/week) Range: 12-37/team (1-3.1/week)	Warm-up movement (50 yards each) Strengthening: walking lunges (20yds × 2)	Nil	No. Web based program expired 19/10/17	7(37)	Poor
				Russian H-S's (3 × 10) Single toe raises (30reps/side) Stretching (30 s × 2) Plyometrics (20 reps each) Agilities (40 yards)				
Junge et al. ²⁷	F-MARC Bricks: Ankle & Knee stability	Not reported	Not reported	Not stated	Coach/player education/supervision; ankle taping, adequate rehabilitation, promotion of fair play	No. 'F-Marc Bricks' not clearly defined.	5(26)	Роаг
Prevention of soccer injuries: a prospective intervention study in youth amateur players	Trunk, hip, leg strength & flexibility				procession on the proj	Several interventions.		
-	Co-ordination, reaction time & endurance							
Owoeye et al. ²⁰	FIFA 11+	One season (25 weeks), 2/week	60% trainings	Not stated	Nil	Yes. FIFA 11+ is well documented	7(37)	Poor
Efficacy of the FIFA 11+ warm-up programme in male youth football.		_,	(Range, 5-22 sessions/team)					
			1.6 tímes/week					

E.A. Smyth et al. / Journal of Science and Medicine in Sport 22 (2019) 887-901

Intervention; article	Intervention summary	Duration & frequency	Compliance levels	Specifics of intervention (reps, sets, intensity, progression)	Co-intervention	Reproducible	Intervention repo standard (e.g. CEF	rting (T)
Scase, et al 2006.	Landing, falling & recovery skills:	8 weeks, 1/week, 30 min sessions	Not reported	Not stated. Reference for intervention no longer available: http:// www.kidskills.com.au	Aerobic endurance, weight training, skill acquisition	No. Specific program for training landing, falling and recovery skills not provided	6(32)	Poor
Teaching landing skills in elite junior Australian football: evaluation of an injury prevention strategy	Prone fall, backward fall, sideways rolling, two foot landing, back shoulder roll							
Silvers-Granelli et al. ³²	FIFA 11+:	4 months, 2–3/week, 15–20 min	2.19 sessions/week	Details on website	Nil	Yes. Details of FIFA 11+ program can be found on provided website	12(63)	Moderate
Does the FIFA 11+ Injury prevention program reduce the incidence of ACL injury in male soccer players?	Dynamic warm-up:		Average: 32.78±12.13 sessions/season					
	Strength, agility, proprioceptive and plyometric exercises							
Silvers-Granelli et al. ³¹	FIFA 11+:	4 months, 3/week, 20 min	Average: 32.78±12.13 sessions/season (2.19 sessions/week)	Details on website	Nil	Yes. Details of FIFA 11+ program can be found on provided website	14(74)	Moderate
Efficacy of the FIFA 11+ injury Prevention Program in the Collegiate male soccer player	Dynamic warm-up:					·		
	Running exercises (8 mins) – cutting, change of direction, decelerating, landing techniques; Strength, plyometric and balance exercises (10 mins); Running exercises (2 mins)							
Zouita et al. ³³	Progressive resistance training	12 weeks, 2–3/week, 90 min	Not reported	Individualised	4–5 soccer sessions/week	No. specific details of strength training,	6(32)	Роог
Strength training reduces injury rate in elite young soccer players during one					1 match/week	program for provided		

season

E.A. Smyth et al. / Journal of Science and Medicine in Sport 22 (2019) 887-901

50

Psychological lvarsson et al. ²⁵ It pays to pay attention: A mindfulness based program for injury prevention with soccer players	'Mindfullness, acceptance and commitment approach' smalt group sessions	7 weeks, 1/week, 45 min	Not reported	Topics covered: Theoretical & practical aspects of intervention	10–18h soccer/week	No. specific details of MAC approach not provided	WIDER score (%) 13(65)	Interpretation Moderate
				Intro to concept of acceptance Intro how to enhance commitment Intro how to combine mindfulness, acceptance & commitment in practice How to maintain & enhance mindfulness, acceptance & commitment				
Johnson, et al. ²⁶	One-on-one psychological training	20 weeks, 6 sessions and 2 phone calls	Not reported	Somatic & cognitive relaxation, stress management skills, goal setting skills, attribution & self-confidence training, identification & discussion about critical incidents	Not stated	No. details of psychological training not provided	6(30)	Poor
Injury prevention in Sweden: helping soccer players at risk		45-90 min						
Nucritión Lewis et al. ²⁸	4000 IU Vitamin D supplement	6 months, daily	Control: 76% (range. 64–87%)	NA	Nil	Yes. Specific details of Vitamin D supplement provided	NA	NA
The effects of season-long Vitamin D supplementation on collegiate swimmers and divers			l'vention: 70% (range, 29–88%)					

CERT: consensus on exercise reporting template, WIDER: workgroup for intervention development and evaluation research, FIFA: Fédération Internationale de Football Associati.



Fig. 2. Risk of bias summary (Please use colour).

36% (4/11) of the included articles.^{23,24,27,29} There was no evidence of tasks 3–6 of IM (Step 5) being utilised to assist with implementation planning.

A meta-analysis was not undertaken due to high risk of bias across all included studies. For the purpose of qualitative synthesis, the results were ordered systematically in a forest plot grouped according to intervention type Fig. 4.

4. Discussion

This systematic review highlights a lack of evidence-based strategies for injury prevention specific to pre-elite athletes. Exercise and psychological interventions have been shown to have efficacy in preventing injury in pre-elite athletes. However, a high proportion of these studies were shown to have a high risk of bias and poor reporting standards of the intervention.

Synthesised results indicate moderate strength of evidence for the efficacy of exercise interventions for the prevention of injury in pre-elite athletes. Three of the studies^{29,31,32} assessed FIFA 11+,



Fig. 3. Intervention Mapping (IM) Step 5 (Please use colour).

two were carried out in the USA College system,^{31,32} and one was in Nigeria,²⁹ All three examined the effect of the FIFA 11+ warmup on all injuries that resulted in players' absence from a match or training session. The American studies showed a 44% reduction in all Division 1 injuries and a 64% reduction in ACL injuries,^{31,32} The Nigerian study reported a significant reduction in injuries but when we re-calculated IRR, the intervention was deemed to be ineffective,²⁹ While significant reductions were produced in some studies, there is still scope for greater injury reduction in the pre-elite environment. This could be achieved by improved implementation planning through the use of IM (Step 5). This will ensure the intervention is appropriate for the population and environment and strategies are employed to enhance reach, adoption and implementation.

There is a lack of evidence for the use of equipment to prevent injury in pre-elite athletes. However, we cannot dismiss the use of equipment for injury prevention because a previously published literature review identified over 600 studies in other athlete populations focussing on the use of equipment such as ankle braces and wrist guards which have been shown to prevent injury.^{34–36} There is uncertainty whether these equipment interventions would also menters have input from the very beginning and they have a strong contribution to the intervention development. Given the lack of reporting of consultation with end-users, it appears that all studies developed interventions and implementation strategies with little consultation with adopters and implementers and limited planning. Future studies should report any engagement with adopters and implementers to allow readers to understand the development process in full.

Adherence of participants to an intervention is vital as this will greatly influence the effectiveness of the intervention. Implementation and intervention planning is essential to maximise compliance. It is difficult to draw conclusions on effectiveness without knowing compliance levels. Five of the eleven studies did not report their compliance levels^{25-27,30,33}, three of these were exercise interventions^{27,30,33} and two were psychology interventions.^{25,26} The nutrition intervention²⁸ had a large range of compliance with the lowest level being 29%, perhaps this research should be repeated after undertaking the implementation mapping process in order to improve compliance and gain a true assessment of efficacy. Of the two exercise interventions that were shown to be ineffective, one did not report compliance²⁷ and the other reported sub-optimal compliance,²⁹

The pre-elite athletic population's demographics can be wide ranging across sports. The age range of athletes in our included studies is 13-23 yo. This population can include pre-pubescent gymnasts and post-pubescent footballers. Nevertheless they can experience similar challenges such as a sudden increase in training load,⁴¹ limited access to support services and a tendency for reduced recovery time due to other commitments such as school, university or work.^{5,6} Perhaps addressing these extrinsic risk factors could be the most effective method for reducing injury in this athletic population.

There is evidence supporting the efficacy of exercise interventions in this population, 24, 27, 29-33 while there is a high risk of bias we need to understand it is very difficult to blind participants so this may be a case of misclassification. For those working in football there is evidence supporting the use of FIFA11+ in elite and community level,^{42,43} however, our findings are purely focussed on pre-elite athletes. Additionally it must be noted that one of the psychology studies produced the best reported reduction in injury rates.²⁶ The authors of this study identified at risk athletes before administering the targeted intervention. Perhaps intervention studies need to be aimed at subgroups of athletes who may be at greater risk due to factors such as previous injury⁹ or maturation level in this population.⁶ There is limited evidence that supports the use of screening for injury prediction⁴⁴ but at a practical level it might be still useful to apply targeted interventions to pre-elite athletes with at-risk profiles.

There is a paucity of research with low risk of bias specific to injury prevention in the pre-elite population. We have identified a lack of exercise intervention studies for female pre-elite athletes. It is recommended that injury prevention interventions that have been shown to work in other athletic populations should be trialled in a variety of sports for pre-elite male and female athletes, and where possible trials should be conducted to ensure a low risk of bias and with intervention reporting standards such as CERT and WIDER.^{20,21} None of the interventions included in this review utilised more than a single discipline approach; multimodal interventions may improve the observed effect given the complexity of injury aetiologies.⁴⁵ Greater attention needs to be directed toward implementation planning to bring about sustained intervention compliance in order to achieve injury reduction in the 'real-world'.⁴⁶ Finally, adoption of a model such as the FTEM framework allows greater consistency in the classification of the sporting population with clear descriptions of the requirements of each level.

This Systematic Review provides a synthesis and summary of evidence, as well as providing clear recommendations for future research investigating injury prevention interventions used in preelite athletes. A limitation of this review was the inability to perform a meta-analysis of results and the generally low number of published studies in this population. Additionally, although we employed a sensitive search strategy, we may have misclassified excluded articles due to low reporting standards of the level of athlete in many interventions.

5. Conclusion

There is limited evidence as to the efficacy or otherwise of exercise and psychological interventions for injury prevention in pre-elite athletes who compete in an Olympic or professional sport. No evidence is available for the use of equipment (mouthguards) and Vitamin D supplementation interventions for the pre-elite athletic population. None of the included studies demonstrated low risk of bias and intervention and implementation planning are not well reported in injury prevention research of pre-elite athletes limiting the ability to implement the results in the real-world. There needs to be a call to action across disciplines to address this apparent shortfall in the literature in order to inform effective practical outcomes.

What is already known?

- Injuries in pre-elite development are common.
- Injuries can compromise a developing athlete's sporting potential, reduce their chance of successful performance and can lead to athletes dropping out of the sport.
- Previous injury predisposes athletes to further injury.

What are the new findings?

- Exercise and psychological interventions may prevent injury in pre-elite athletes.
- There is insufficient evidence to evaluate the use of equipment and nutrition interventions in pre-elite athletes.
- There is a lack of level 1 evidence for injury prevention in the pre-elite population.
- More injury prevention intervention studies need to focus on female athletes.

Funding

This study received no funding. This work was supported by a joint AIS and UC scholarship awarded to the author ES for support during her PhD.

Acknowledgment

Nil.

Appendix A. Supplementary data

Supplementary data associated with this article can be found, in the online version, at https://doi.org/10.1016/j.jsams.2019.03.002.

References

- 1. Drew MK, Raysmith BP, Charlton PC. Injuries impair the chance of successful
- performance by sportspeople: a systematic review. *Br J Sports Med* 2017; 5. 2. Charlton PC, Drew MK, Mentiplay BF et al. Exercise interventions for the prevention and treatment of groin pain and injury in athletes: a critical and systematic review. Sports Med 2017; 47(10):2011-2026.

- 3. Leppänen M, Aaltonen S, Parkkari J et al. Interventions to prevent sports related injuries: a systematic review and meta-analysis of randomised controlled trials. Sports Med 2014; 44(4):473-486.
- 4. Vriend I, Gouttebarge V, Finch CF et al. Intervention strategies used in sport injury prevention studies: a systematic review identifying studies applying the haddon matrix. Sports Med 2017; 47(10):2027-2043.
- 5. Weissensteiner JR. Method in the madness: working towards a viable 'paradigm' for better understanding and supporting the athlete pathway, in Handbook of Talent Identification and Development in Sport, Baker J, Cobley S, Schorer J, Wattie N, editors, Abingdon, T&F Routledge, 2017.
- Gergeron MF, Mountjoy M, Armstrong N et al. International Olympic Commit-tee consensus statement on youth athletic development. Br J Sports Med 2015; 49(13):843-851 7. Brown KA, Patel DR, Darmawan D, Participation in sports in relation to adoles-
- cent growth and development. Transl Pediatr 2017; 6(3):150-159.
- 8. Hägglund M, Waldén M, Ekstrand J. Previous injury as a risk factor for injury in elite football: a prospective study over two consecutive seasons, Br I Sports Med 2006; 40(9):767-772.
- 9. Toohey LA, Drew MK, Cook JL et al. Is subsequent lower limb injury associated with previous injury? A systematic review and meta-analysis. Br J Sports Med 2017: 51(23):1670
- 10. Huxley DJ, O'Connor D, Healey PA. An examination of the training profiles and injuries in elite youth track and field athletes. Eur J of Sport Sci 2014; 14(2):185-192.
- Moher D, Liberati A, Tetzlaff J et al. Preferred reporting items for system-atic reviews and meta-analyses; The PRISMA statement. Int J Surg 2010; 8(5):336-341.
- 12. Gulbin JP, Croser MJ, Morley EJ et al. An integrated framework for the optimisation of sport and athlete development: a practitioner approach. J Sports Sci 2013; 31(12):1319.
- Timpka T, Jacobsson J, Ekberg J et al. Meta-narrative analysis of sports injury reporting practices based on the Injury Definitions Concept Framework (IDCF); a review of consensus statements and epidemiological studies in athletics (track and field). J Sci Med Sport 2015; 18(6):643–650.
- 14. Oxford Centre for Evidence-Based Medicine Working Group. The Oxford 2011 Levels of Evidence. Available at: http://www.cebm.net/index.aspx?o=5653. 15. Kok G. A practical guide to effective behavior change; how to apply theory-and
- evidence-based behavior change methods in an intervention. Eur Health Psychol 2014; 16(5):156-170. 16. Donaldson A, Lloyd DG, Gabbe BJ et al. We have the programme, what next?
- Planning the implementation of an injury prevention programme. Inj Prev 2017; 23(4)-273
- 17. Higgins JP, Altman DG, Gøtzsche PC et al. The Cochrane Collaboration's tool for
- assessing risk of bias in randomised trials. *BMJ* 2011;343, d5928.
 Reurink G, Goudswaard GJ, Moen MH et al. Myotoxicity of injections for acute muscle injuries: a systematic review. *Sports Med* 2014; 44(7):943–956.
- 19. Timpka T, Jacobsson J, Bickenbach J et al. What is a sports injury? Sports Med 2014: 44(4): 423-428.
- 20. Slade SC, Dionne CE, Underwood M et al. Consensus on Exercise Reporting Template (CERT); modified delphi study. Phys Ther 2016; 96(10):1514–1524. 21. Albrecht L, Archibald M, Arseneau D et al. Development of a checklist to
- assess the quality of reporting of knowledge translation interventions using the Workgroup for Intervention Development and Evaluation Research (WIDER) recommendations. Implement Sci 2013; 8(1):52.
- 22. Liberati A, Altman DG, Tetzlaff J et al. The PRISMA statement for reporting systematic reviews and meta-analyses of studies that evaluate health care interentions: explanation and elaboration. PLoS Med 2009; 6(7):e1000100.
- 23. Barbic D, Pater J, Brison RJ. Comparison of mouth guard designs and concussion prevention in contact sports: a multicenter randomized controlled trial. Clin J Sport Med 2005; 15(5):294-298.
- 24. Gilchrist J, Mandelbaum BR, Melancon H et al. A randomized controlled trial to prevent non contact anterior cruciate ligament injury in female collegiate soccer players. Am J Sports Med 2008; 36(8):1476–1483.

- 25. Ivarsson A, Johnson U, Andersen MB et al. It pays to pay attention: a mindful nessbased program for injury prevention with soccer players. | Appl Sport Psychol 2015; 27(3):319-334.
- Johnson U, Ekengren J, Andersen MB. Injury prevention in Sweden: helping soccer players at risk. J Sport Exerc Psychol 2005; 27(1):32. 26.
- Junge A, Rosch D, Peterson L et al. Prevention of soccer injuries: a prospective intervention study in youth amateur players. Am J Sports Med 2002; 30(5):652-659.
- Lewis RM, Redzic M, Thomas DT, The effects of season-long vitamin D supplementation on collegiate swimmers and divers. Int J Sport Nutr Exerc Metab 2013; 23(5):431-440
- Owoeye OB, Akinbo SR, Tella BA et al. Efficacy of the FIFA 11+ warm-up pro-29. gramme in male youth football: a cluster randomised controlled trial. J Sj Sci Med 2014: 13(2):321-328.
- Scase E, Cook J, Makdissi M et al. Teaching landing skills in elite junior Australian football: evaluation of an injury prevention strategy. Br J Sports Med 2006; 40(10):834-838.
- Silvers-Granelli H, Mandelbaum B, Adeniji O et al. Efficacy of the FIFA 11+ injury prevention program in the collegiate male soccer player. Am J Sports Med 2015; 43(11):2628-2637
- Silvers-Granelli HJ, Bizzini M, Arundale A et al. Does the FIFA 11+ injury pre-vention program reduce the incidence of ACL injury in male soccer players? *Clin* Orthop Relat Res 2017; 475(10):2447-2455.
- Zouita S, Zouita AB, Kebsi W et al. Strength training reduces injury rate in elite young soccer players during one season. J Strength Cond Res 2016; 30(5):1295-1307.
- Klugi M, Shrier I, McBain K et al. The prevention of sport injury: an analysis of 12,000 published manuscripts. *Clin J Sport Med* 2010; 20(6):407–412. Verhagen E, Bay K. Optimising ankle sprain prevention: a critical review and 34.
- practical appraisal of the literature, Br J Sports Med 2010; 44:1082-1088. Russell K, Hagel B, Francescutti LH. The effect of wrist guards on wrist and
- 36. arm injuries among snowboarders; a systematic review, Clin J Sport Med 2007; 17(2):145-150.
- Maddison R, Prapavessis H. A psychological approach to the prediction and prevention of athletic injury. J Sport Exerc Psychol 2005; 27(3):289–310. 37.
- 38 Perna FM, Antoni MH, Baum A et al. Cognitive behavioral stress management effects on injury and illness among competitive athletes: a randomized clinical trial. Ann Behav Med 2003; 25(1):66-73.
- Mountjoy M, Sundgot-Borgen JK, Burke LM et al. IOC consensus statement on relative energy deficiency in sport (RED-S): 2018 update. Br J Sports Med 2018; 39 52(11):687.
- Drew MK, Vlahovich N, Hughes D et al. A multifactorial evaluation of illness risk factors in athletes preparing for the Summer Olympic Games. J Sci Med Sport 2017; 20(8):745-750.
- Bahr R. Demise of the fittest: are we destroying our biggest talents? Br J Sports 41. Med 2014; 48(17):1265.
- 42. Daneshjoo A, Mokhtar AH, Rahnama N et al. The effects of injury prevention warm-up programmes on knee strength in male soccer players. Biol Sport 2013; 30(4):281-288
- 43. Steffen K, Emery CA, Romiti M et al. High adherence to a neuromuscular injury prevention programme (FIFA 11+) improves functional balance and reduces injury risk in Canadian youth female football players: a cluster randomised trial. Br J Sports Med 2013; 47(12):794-802.
- Bahr R. Why screening tests to predict injury do not work and probably never will...; a critical review. Br J Sports Med 2016; 50(13):776–780.
- 45. Bittencourt NFN, Meeuwisse WH, Mendonca LD et al. Complex systems approach for sports injuries; moving from risk factor identification to injury pattern recognition-narrative review and new concept. Br J Sports Med 2016; 50(21):1309-1314.
- 46. Thorborg K, Krommes KK, Esteve E et al. Effect of specific exercise-based football injury prevention programmes on the overall injury rate in football: a systematic review and meta-analysis of the FIFA 11 and 11+ programmes. Br J Sports Med 2017; 51(7):562-571.


Figure 9 (Figure 4 in publication). Forest plot showing IRR and 95%CI of number of injuries in intervention group compared to control group. IRR: incidence rate ratio, CI: confidence interval.

Systematic review update

Methods

Subsequent to the published systematic review, an electronic search of the literature in MEDLINE, CINAHL, PubMed, SportDISCUS and Web of Science was repeated on 22 February 2021 with a date restriction of May 2017 onwards and English language only as a limitation. The same search strategy used in our published Systematic Review¹⁰⁴ was employed. All potential references were imported into Endnote X7 (Thompson Reuters, Carlsbad, California, USA) and then into Covidence (Covidence systematic review software, Veritas Health Innovation, Melbourne, Australia. Available at www.covidence.org) and duplicates removed. Titles and abstracts were independently screened for eligibility by two authors (ES and PN). The full text of each included study was then independently examined according to the inclusion and exclusion criteria used in our published Systematic Review, and reference lists of included studies were manually searched to find any further articles for inclusion. In the event of disagreement between authors, an additional author (GW) was consulted to reach

consensus. Data extraction from the published Systematic Review¹⁰⁴ was repeated and presented in the Tables below. Two authors (ES and MD) repeated an independent risk of bias assessment on each included article using the Cochrane Collaboration's tool for assessing the risk of bias in randomised trials (Figure 9).¹⁰⁵ As this new search was conducted for the purposes of updating our published Systematic Review,¹⁰⁴ the same risk of bias tool was applied despite two of the articles being cohort studies rather than randomised studies. Risk of bias for implementation was repeated by assessing each article against IM (Step 5) criteria (Figure 8).^{77,79}

Results

The electronic search identified 1971 articles; duplicates were then removed, resulting in 1893 articles. The titles and abstracts were screened, which reduced the number of articles to 17. The full texts of these articles were obtained and assessed for eligibility against the inclusion criteria. No articles were added following citation checking. Fourteen studies were excluded due to not meeting the inclusion criteria. Two of the most common reasons for exclusion were 1) wrong patient population (n=8) and 2) insufficient detail (n=5). The three remaining articles were included for full review and data synthesis.¹⁰⁶⁻¹⁰⁸ The flow chart of this process is presented in Figure 7.



Figure 10: PRISMA flow chart, PRISMA: Preferred Reporting Items for Systematic Reviews and Meta-Analyses

One article examined whether load management using the Acute:Chronic Workload Ratio (ACWR) could prevent health problems in male and female elite youth football (soccer) players.¹⁰⁶ The other two articles assessed the effectiveness of exercise programs to reduce injury.^{107,108} More specifically, one of the exercise studies involved the delivery of a sensorimotor training intervention for male and female collegiate football (soccer) players. The other exercise study investigated the effect of a hip-focused injury prevention program for reducing the incidence of ACL injury in female collegiate basketball players.

The injury definitions and Injury Definitions Concept Framework (IDCF) classification for each study are outlined in Table 1. According to the IDCF classification, one study used athlete self-report¹⁰⁶ and the other two studies relied on clinical examination.^{107,108} The level of strength according to guidelines previously published¹⁰⁹ and effectiveness based on IRR are outlined in Table 1. The strength of evidence was deemed to be limited for load management and moderate for exercise. The hierarchical level of

evidence was Level 2 for load management¹⁰⁶ and Level 3 for exercise because they were non-randomised cohort studies (Table 2).^{107,108}

The risk of bias assessment is outlined in Figure 9. Risk of bias assessment concluded that these three studies were deemed at high risk of bias.¹⁰⁶⁻¹⁰⁸ A major contributor to assessments of high risk of bias was lack of randomisation, blinding and allocation concealment.

The Consensus on Exercise Reporting Template (CERT) was employed to rate the reporting standard of the exercise studies. The following criteria has been previously applied¹¹⁰: studies satisfying >75% of criteria were considered to have a high level of reporting standard; 60–74% moderate; and those satisfying less than 60% of criteria were considered to have poor exercise reporting standards. The CERT scores were $53\%^{107}$ and $37\%^{108}$ for the exercise studies, which were therefore deemed to have poor reporting standards (Table 3).

There was no evidence of IM (Step 5) being employed to guide the implementation process for any of the included articles. Results are shown in Figure 8. Adopters and implementers were identified in all studies; however, there was no evidence of developing an implementation planning group or developing objectives, determinants, change objectives or strategies for reach, adoption and implementation. There was, however, some evidence of designing reach, adoption and implementation interventions.

Discussion

The two exercise studies were found to be effective for preventing injury in pre-elite athletes, which supports the findings of the published Systematic Review.¹⁰⁴ However, these studies were shown to have a high risk of bias and poor reporting standards. Load management as an intervention was not included in the published Systematic Review.¹⁰⁴ This subsequently identified study demonstrated that load management using the ACWR was not successful in preventing health problems in pre-elite footballers.¹⁰⁶ The authors identified that a major limitation of the study was the difficulty with assessing coaches' adherence to the intervention and the extent to which their training planning was influenced by the ACWR. Another limitation was a low (69%) average athlete response rate for the HPQ.

Table 2: Intervention – description of study participants, level of competition, incidence of injury of included studies, injury definition and injury definitions concept framework (IDCF)

			Injury Incidence	ce or Prevalence			
Intervention type; Article	Participants (male/female)	FTEM level*, age (Age±SD)	Intervention (exposure)	Control/ pre- intervention (exposure)	IRR or RR (95%CI)	Injury definition	IDCF
Load management							
Dalen-Lorentsen, 2020. Does load management using the acute:chronic workload ratio prevent health problems? A cluster randomised trial of 482 elite youth footballers of both sexes.	Soccer (M&F)	FTEM: T 3/4 Con: 17.4 <u>+</u> 1.1 Int: 17.2 <u>+</u> 1.2	All health problems: 65.7% (61.1%– 70.2%) Substantial health problem: 31.1% (26.7% – 35.5%)	All health problems: 63.8% (60.0% – 67.7%) Substantial health problem: 35.3% (31.6% – 39.1%)	All health problems: RR 1.01 (0.91– 1.12); p=0.84 Substantial health problem: RR 0.88 (0.72– 1.06); p=0.17	All complaints, irrespective of their consequences on football participation or their need to seek medical attention, including illness & injury.	Athlete sel-report Injury Disease Clinical Examination Athlete sel-report Incapacity or Sickness Perfort
Exercise							
Reneker, 2019. Sensorimotor training for injury prevention in collegiate soccer players: An experimental study	Soccer (M&F) NCAA Div 2	FTEM: T 3 20.2 <u>+</u> 1.46	8.94 injuries/1000 athlete exposures	11.8 injuries/1000 athlete exposures	IRR 0.76 (0.69– 0.83); p=0.00	Occurred during participation in a NCAA-sanctioned practice or competition and required attention from an athletic trainer and or physician. Contusions and overuse injuries were not included.	Traums or liness liness clince Essentington
Omi, 2018. Effect of hip-focused injury prevention training for anterior cruciate ligament injury reduction in female basketball players. A 12-year prospective intervention study	College basketball (F) Equivalent to NCAA Div 2	FTEM: T3 19.6 <u>+</u> 1.1	0.10 ACL injuries/1000 athlete exposures	0.25 ACL injuries/1000 athlete exposures	RR 0.38 (0.17– 0.87); p=0.017	ACL injury diagnosed by a physician and verified by arthroscopy or magnetic resonance imaging.	Inceptory Clences

FTEM: Foundation, Talent, Elite, mastery Framework; IRR: Incidence Rate Ratio; IDCF: Injury definitions Concept Framework; NCAA: The National Collegiate Athletic Association; AE: Athlete Exposure; ACL: Anterior Cruciate Ligament

Intervention; Article	Data collection period	Statistical analysis	Effectiveness/ level of strength for intervention type	Level of evidence (OCEBM Level)
Load management				
Dalen-Lorentsen, 2020. Does load management using the acute:chronic workload ratio prevent health problems? A cluster randomised trial of 482 elite youth footballers of both	Feb–Nov 2018	Outcome measures: Between-group difference in prevalence & relative risk ratio. Generalised estimating equations with intention-to-treat principle	Х	Level 2
seres.		principle	Level of strength: Limited	
<u>Exercise</u>				
Reneker, 2019. Sensorimotor training for injury prevention in collegiate soccer players: An experimental study	2017–2018	Descriptive statistics: means with standard deviation, counts with percentages. Right skewed outcomes: log- gamma mixed effect random intercept models. Normal outcomes: Gaussian	✓	Level 3
Omi, 2018. Effect of hip-focused injury prevention training for anterior cruciate ligament injury reduction in female basketball players. A 12-year prospective intervention study	2003–2014	random intercept models Injury rates: Poisson models Outcome measure: ACL injuries. Incidence rates Relative risk: chi-square analysis	✓ Level of strength:	Level 3
			Moderate	

Table 3: Characteristics of the studies including data collection period, statistical analysis, efficacy, level of evidence.

OCEBM: Oxford Centre for Evidence Based Medicine, ACL: Anterior Cruciate Ligament

Intervention; Article	Intervention summary	Duration & frequency	Specifics of intervention (reps, sets, intensity, progression)	Co- intervention	Reproducible	Intervention reporting	standard (e.g., CERT)
Load management							
Dalen-Lorentsen, 2020. Does load management using the acute:chronic workload ratio prevent health problems? A cluster randomised trial of 482 elite youth footballers of both serves	Individualised load management	NA	NA	Nil	No. A lot of variability in application.	NA	NA
Exercise						CERT score (%)	Interpretation
Reneker, 2019. Sensorimotor training for injury prevention in collegiate soccer players: An experimental study.	Vestibular, visual & oculomotor activities, cervical neuromotor control & strength training, and postural/balance exercises.	4 weeks 2 face-to-face sessions/week & home exercise program	Circuit of exercises face-to- face: (45mins total) 2–3 rounds, 5–6 exercises, 2.5– 3.5 mins/station HEP: Horizontal gaze stability Near-point convergence Horizontal eye-head coordination	Nil	Yes.	10(53)	Poor
Omi, 2018. Effect of hip- focused injury prevention training for anterior cruciate ligament injury reduction in female basketball players. A 12- year prospective intervention study	Hip-focused injury prevention (HIP) training: Jump-landing +/- band Strengthening Balance	8 years. Education: 3/season HIP training: 20 mins, 3/week	Detailed program outlined in Table 2 of publication	Education for players, coaches & student athletic trainers	Yes.	7(37)	Poor

Table 4:Intervention summary, duration and frequency, specific details of intervention (i.e., reps, sets, intensity, progression), co-intervention, reproducible, intervention reporting standard

CERT: Consensus on Exercise Reporting Template; HEP: Home Exercise Program; HIP: Hip-focused Injury Prevention



Conclusion

lacking. more included female athlete participants, which the studies from the published Systematic Review¹⁰⁴ were Two of the additional studies that met the inclusion criteria support the finding of the published to assess the effectiveness of load management in pre-elite athletes based on one study and, as such, Systematic Review¹⁰⁴ that exercise interventions may prevent injury in pre-elite athletes. It is difficult research in this area is indicated. On a final note, all the included studies from this new search

Chapter 4 'Ideal conditions'/Scientific evaluation of the preventative measure (TRIPP 4)

Published work

This chapter is based on a manuscript that has been published in the journal, Physical Therapy in Sport.

Smyth E, Waddington G, Witchalls J, et al. Does ankle tape improve proprioception acuity immediately after application and following a netball session? A randomised controlled trial. *Phys Ther Sport.* 2021; 48:20-25.

Chapter 3's systematic review identified level 2 and 3 evidence for exercise and psychology interventions preventing injury in pre-elite athletes. Additionally, there was an absence of evidence to support the use of equipment, nutrition and load management interventions within the pre-elite athlete cohort. This third study provides evidence for the use of tape (i.e., equipment) in pre-elite netball athletes.

The most frequent injury reported at the 2018 ANNC was ankle sprain. A recent literature review¹¹¹ supported the use of a brace or tape to reduce the risk of recurrent and initial ankle sprain. Additionally, a significant association between poor single leg balance and ankle sprain has been identified.¹¹² To control balance, the central nervous system (CNS) combines visual, vestibular, and proprioceptive information to coordinate the activation patterns of muscles.¹¹³ At the time of this thesis being developed, Netball Australia had a universal prevention policy requiring all pre-elite athletes to tape their ankle for training and matches. Therefore, to assess the efficacy or effectiveness of injury prevention in this population was deemed unfeasible as it would require removing preventative actions from the group. Discussion with Netball Australia and relevant ethics committees determined that removing a preventative action was not ethical, noting it would be the best scientific approach. Therefore, we decided to test the hypothesis that ankle tape improves proprioception to determine if there is evidence to support the rationale for the continued use of the ankle taping policy. It also provides evidence for whether there is a difference between a Sport and Exercise Physiotherapist (SEP) taping

an athlete's ankle and an athlete performing self-applied taping, which has practical and clinical importance.

Co-authors' declaration

Declaration of Co-Authored Publications

CANBERRA

For use in theses which include co-authored publications. This declaration must be completed for each coauthored publication and to be placed at the start of the thesis chapter in which the publication appears, or as a preface to the thesis.

Declaration for Thesis Chapter Four

Publication: "Does ankle tape improve proprioception acuity immediately after application and following a netball session?: a randomised controlled trial"

DECLARATION BY CANDIDATE

In the case of Chapter Four, the nature and extent of my contribution to the work was the following:

 Nature of Contribution
 Extent of Contributions (%)

Conception and design of the research proposal	80%
Data collection	50%
Analysis and interpretation of the findings	70%
Writing the paper and critical appraisal of content	70%
Corresponding author for journal communication and publication peer-review process.	100%

The following co-authors contributed to the work:

Name	Nature of Contribution	Contributor is also a UC student (Yes/No)
Gordon Waddington	Conception and design of the research proposal Data collection Analysis and interpretation of the findings Writing the paper and critical appraisal of content	No
Jeremy Witchalls	Data collection Analysis and interpretation of the findings	No
Phillip Newman	Analysis and interpretation of the findings Writing the paper and critical appraisal of content	No
Juanita Weissensteiner	Writing the paper and critical appraisal of content	No
Steven Hughes	Data collection	No
Theo Niyosenga	Analysis and interpretation of the findings	No
Michael Drew	Conception and design of the research proposal Analysis and interpretation of the findings Writing the paper and critical appraisal of content	No



___17/09/2021____

Candidate's Signature

Date

Declaration of Co-Authored Publications DECLARATION BY CO-AUTHORS



The undersigned hereby certify that:

- (1) the above declaration correctly reflects the nature and extent of the candidate's contribution to this work, and the nature of the contribution of each of the co-authors.
- (2) they meet the criteria for authorship in that they have participated in the conception, execution, or interpretation, of at least that part of the publication in their field of expertise;
- (3) they take public responsibility for their part of the publication, except for the responsible author who accepts overall responsibility for the publication;
- (4) there are no other authors of the publication according to these criteria;
- (5) potential conflicts of interest have been disclosed to (a) granting bodies, (b) the editor or publisher of journals or other publications, and (c) the head of the responsible academic unit; and
- (6) the original data are stored at the following location(s) and will be held for at least five years from the date indicated below:

[Please note that the location(s) must be institutional in nature, and should be indicated here as a department, centre or institute, with specific campus identification where relevant.]

Location(s):	Australian Institute of Sport		

Gordon Waddington

Signed:	ga use
Date:	17/09/2021

Jeremy	Witchalls	1.00
Signed:	Jeremy	Witchalls
Date:	03/09/20	121

Phil Newman Signed:

Date: 11/09/2021

Juanita Weissensteiner Signed: Juanita Weissensteiner Date: 30/08/2021

Steven Hughes Signed: Staron Uffer Date: 3/9/2021

Theo Niyosenga

Signed: <u>Theo Niyonsonga</u> Date: 07/09/2021

Michael Drew Signed: Michael Drew Date: 17/09/2021

Does ankle tape improve proprioception acuity immediately after application and following a netball session? A randomised controlled trial

Physical Therapy in Sport 48 (2021) 20-25



Original Research

Does ankle tape improve proprioception acuity immediately after application and following a netball session? A randomised controlled trial



^a University of Canberra Research Institute for Sport and Exercise, Building 29, University of Canberra, Bruce, ACT, 2617, Australia ^b Australian Institute of Sport, Leverrier St, Bruce, ACT, 2617, Australia

^c Sport Development Group, New South Wales Office of Sport, Level 3, 6B Figtree Drive, Sydney Olympic Park, Sydney, NSW, 2127, Australia

^d New South Wales Institute of Sport, Bdg B, Level 1, 6 Figtree Drive, Sydney Olympic Park, NSW, 2127, Australia
 ^e Health Research Institute, Locked Bag 1, University of Camberra, Bruce, ACT, 2601, Australia

ARTICLE INFO

Article history: Received 2 Sept Received in revi 9 December 202 Accepted 11 Dee

Keywords: Ankle Tape Proprioception

ABSTRACT

ember 2020 sed form 20 rember 2020	Objectives: To assess whether ankle tape applied by a Sport and Exercise Physiotherapist (SEP) or self- applied by the athlete results in a change in proprioception and whether it is maintained during a netball session. Design: Randomised controlled trial. Setting: Australian Institute of Sport.
	— Participants: 53 pre-elite netball athletes.
	Main outcome measures: Athlete proprioception was assessed using the Active Movement Extent Discrimination Apparatus (AMEDA) on four occasions for each taping condition: 1) pre-tape, 2) post-
	Results: Mixed effect linear models were used for analysis. A significant increase in proprioception was observed when self-tape: 0.022 (95% CI: [-0.000 – 0.044], $p = 0.05$), and SEP tape: 0.034 (95% CI: [0.012 – 0.055], $p < 0.01$), were initially applied. These improvements were maintained during a netball session for both, self-taping: 0.01 (95% CI: [-0.01 – 0.02], $p = 0.45$) and SEP-taping: <0.01 (95% CI: [-0.02 – 0.01], $p = 0.56$). Results also indicate there was no significant difference between taping conditions ($\beta = -0.001$, 95% CI: [-0.02 – 0.02], $p = 0.90$).
	Conclusions: Proprioception improves and is maintained during a netball session with either SEP or self- applied taping.

Crown Copyright © 2020 Published by Elsevier Ltd. All rights reserved.

Introduction

Ankle sprains have been reported to have the highest injury incidence in the sport of netball (Finch et al., 2002; Hume & Steele, 2000; Langeveld et al., 2012), particularly in 17/U & 19/U netball athletes (Hume & Steele, 2000). Taping is often recommended as a strategy for preventing ankle sprain (Verhagen & Bay, 2010), but the mechanism by which taping works is not completely understood. Taping and bracing have been shown to reduce the risk of

* Corresponding author. University of Canberra Research Institute for Sport and Exercise, Building 29, University of Canberra, Bruce, ACT, 2617, Australia. E-mail address: erin.smyth@ausport.gov.au (E. Smyth).

https://doi.org/10.1016/j.ptsp.2020.12.010

1466-853X/Crown Copyright @ 2020 Published by Elsevier Ltd. All rights reserved.

initial and recurrent ankle sprains (Vuurberg et al., 2018). Studies have shown that the mechanical protection offered by tape declined nearly completely after 45 minutes (Best et al., 2014) with a reduction after 10 minutes of sporting activity (Verhagen et al., 2001) thereby challenging mechanical protection as the sole mechanism for effectiveness.

Improved proprioception, defined as the perception of body position and movements in three-dimensional space (Han et al., 2016), has been widely accepted as a plausible reason for injury prevention. It is hypothesised that tape stimulates the cutaneous sensory receptors of the leg and foot, resulting in better awareness of foot position and muscle activation prior to, and during, foot contact with the floor (Robbins et al., 1995; Semple et al., 2012). A

recent systematic review refutes these claims, concluding that using an ankle brace or ankle tape has no effect on proprioception acuity in participants with recurrent ankle sprain or functional ankle instability (Raymond et al., 2012). The majority of ankle tape studies investigating proprioception have not been performed in the last decade (Halseth et al., 2004; Refshauge et al., 2009; Simoneau et al., 1997), and we now have more sophisticated methods for assessing proprioception (Han et al., 2016).

The ankle Active Movement Extent Discrimination Apparatus (AMEDA) is a device used to measure proprioception of the ankle joint in a functional context. It is a purpose-built device constructed by the project team for research purposes. Using a dynamic platform, the device records the athlete's accuracy in discriminating changes in floor angle while moving a foot plate actively, standing up, with body weight on the feet (Steinberg, Waddington, Adams, & et al, 2016). This testing apparatus has evolved over the last 20 years, resulting in a user friendly, reliable (Witchalls et al., 2014) and valid (Han et al., 2016; Lewkowicz, 2001) measure of proprioception. Availability of this sophisticated apparatus warrants reassessment of the effect tape has on proprioception.

This study will contribute to the evidence by examining whether ankle taping changes proprioceptive acuity and whether it is maintained during a netball session. We will also assess whether athlete self-taping is less effective than if the tape was applied by trained personnel. This research gives us the opportunity to test whether the different taping techniques result in different proprioceptive acuity.

This paper will aim to answer the following questions:

- Does tape improve the ankle proprioception of pre-elite netball athletes?
- 2) If pre-elite netball athlete proprioception improves with the application of tape, is it maintained during a netball session?
- 3) Is there a difference in proprioception change when pre-elite athletes apply their own tape compared with a Sport & Exercise Physiotherapist (SEP) applying the ankle tape?

Methods

Participants

We recruited a representative sample of pre-elite netball players to further investigate taping in this population. A situation camp hosted by the national netball organisation represented the most feasible and most comparable training programs, environment, equipment, staff and stage of the season. All fifty-three pre-elite netball athletes (athletes with a State Academy/Institute scholarship) attending the camp were invited to take part in this randomised controlled trial. They were attending a netball camp at the Australian Institute of Sport from the 11–14 December 2018. Sample size was based on previous published data demonstrating statistical significance using similar testing protocols (Long et al., 2017; Steinberg et al., 2015).

Athletes were excluded if they had a current injury preventing them from full netball participation. All athletes were female and the mean age was 17.7 (range 16–23). Given that the athletes regularly tape their ankles, a superiority randomised controlled trial was employed as it was deemed unacceptable to withhold taping from athletes by the sporting organisation.

Protocol

Proprioceptive acuity was tested using the AMEDA (Waddington & Adams, 1999). The athlete stood with one foot on a fixed platform

Physical Therapy in Sport 48 (2021) 20-25

and the other on a moveable plate with the pivot axis through the centre of the footprint, allowing inversion movement of the ankle. The computer-controlled stepper motor device provides five exact inversion angles at one-degree intervals between 10 and 14 degrees of inversion, and the test determines the sensitivity of the individual with respect to detecting differences between the positions. Each athlete was familiarised with the AMEDA apparatus before data collection commenced. This included an explanation and then three trials of the five-movement displacement distances, in order, from the smallest to the largest: fifteen movements in total. Athletes then performed 50 trials of testing, in which all five positions were presented ten times, in a random order. Each trial consisted of one active inversion movement out to a stop at a steady pace, followed by return to the start position. After each trial, the athlete was asked to make a judgement as to the position number (1-5) of each test movement. They were not provided with any feedback regarding the correctness for each trial. They relied on their memory of the five movement extents from the familiarisation session to enable them to identify each trial and make a numerical judgement (1-5) (Han et al., 2016). AMEDA results were generated using the receiver operating characteristic (ROC) method (Maher & Adams, 1996). The area under the curve (AUC) measures were averaged over the five movement stimulus set to give a single discrimination score for each test.

Following the first proprioception assessment, athletes were randomly allocated to either a self-applied or SEP applied taped group using computer generated randomisation. The lead author (ES) generated the random allocation sequence, enrolled athletes and assigned athletes to the taping conditions. Following ankle taping, the athlete's proprioception was promptly assessed again using the same protocol. AMEDA testers were blinded to the taping condition. Immediately after a netball session, proprioception was assessed with tape insitu and again after the tape was removed. This was to assess whether tape was influencing proprioception following a netball session. These final two assessments occurred within 20 minutes of completing the netball session. There was a minimum of 6 hours between the 1st and 2nd netball sessions. To assess whether there was equal loads across athletes, individual loads were monitored using the ClearSky T6 local positioning system^a (Catapult Sports, Melbourne, VIC, Australia). The ClearSky system utilizes ultra-wideband anchor nodes to triangulate the location of receiver tags, secured inside a vest and worn between the scapulae of each athlete (Luteberget et al., 2018; Serpiello et al., 2018). This provides indoor motion analysis (i.e. total distance, average velocity etc.) for athletes. Testing was performed court side at the Australian Institute of Sport indoor netball courts and a crossover design was employed so that each athlete experienced both tape conditions.

The SEP applied tape consisted of applying 'skin-prep', then 5 cm microfoam^b (3 M Australia) to the anterior and posterior ankle to prevent tape cut. The athlete sat on a physiotherapy plinth in a long sit position with the ankle dorsiflexed. Rigid 38 mm sports tape^c (Elastoplast Leuko Premium) was used to apply an anchor at the distal 1/3 shin, 2 x medial to lateral stirrups, 2×6 's starting and ending laterally, 1 x half heel lock starting and ending laterally, 1 x half heel lock starting and ending laterally, 1 x half heel lock starting and ending medially, 2-3 closing anchors, ensuring full skin coverage of the bottom 1/3 leg (Fig. 1). (Zuluaga, 1995) The self-tape group were encouraged to apply tape in the manner that they normally do, using the same products used by the Sports & Exercise Physiotherapist.

Ethics approval was obtained from the University of Canberra Human Research Ethics Committee and written informed consent was gained from athletes or their parents/legal guardians before commencement of data collection. The CONSORT standards of reporting of RCTs were followed (Schulz et al., 2010).



Fig. 1. Sport & Exercise Physiotherapist taping method.

Statistical analysis

Data were analysed using STATA® statistical software^d (STATA®/ IC 15; StataCorp, College Station, TX, USA). A mixed effects linear model was used to test the effect of the SEP intervention (SEP versus self-applied applied taping) on athletes' ankle AMEDA scores as well changes in AMEDA score over time. The fixed effects were the testing period (pre-tape, post-tape, post-netball, or postnetball no-tape), the tape condition (self-applied or SEP applied) and the order of condition (self-applied or SEP applied tape first). A random effect for participant was included in models to account for within player covariance as players were repeatedly measured over time and conditions. A three-way interaction model (order, tape, testing period) was employed to assess learning effect. A two-way interaction model (tape, testing period) was used to analyse tape effect as there was no significant interaction between tape and testing period. We used the Wald test to determine model significance. The likelihood ratio test was utilised to compare models with and without random effects. It was hypothesised that workload in the training session may affect the AMEDA results, therefore total minutes and distance (metres) covered in each session were compared for each tape condition (self-applied or SEP applied) using mixed effect linear models with random effect for participant. These models included a two-way interaction between minutes and distance with the player as a random effect to account for within player covariance. The a priori level of significance was set at $\alpha = 0.05$ with 95% confidence intervals (95% CI) reported to indicate the precision of estimates.

Results

All 53 netball athletes completed testing with the two different tape conditions (Fig. 2). All netball sessions were 45-60 minutes in

Physical Therapy in Sport 48 (2021) 20-25



duration and consisted of either netball specific drills with a ball or match-play. All mean AMEDA scores are presented in Table 1 and Fig. 3. The participant random effect was significant (χ (Hume & Steele, 2000) = 122.1, p < 0.01) and the overall significance of the mixed-effects restricted maximum likelihood (REML) regression was significant (Wald $\chi_7^2 = 21.72$, p < 0.01). The athletes who had SEP taping for the first session and self-taping for the second session improved their pre-test by 0.053 (95% CI: [0.032–0.073], p < 0.01) and the athletes that had self-applied tape for the first session and SEP applied taping for the second session improved their pre-test by 0.058 (95% CI: [0.036–0.080], p < 0.01).

A significant increase in AMEDA score, indicating improved proprioception was observed when SEP applied tape, 0.034 (95% CI: [0.012-0.055], p < 0.01) and self-applied tape, 0.022 (95% CI: [-0.000 - 0.044], p = 0.05), were initially applied. These improvements were maintained during a netball session for both, self-applied tape: 0.01 (95% CI: [-0.01 - 0.02], p = 0.45) and SEP applied taping: 0.00 (95% CI: [-0.02 - 0.01], p = 0.56). Results also indicate that there was no significant difference in AMEDA scores between self-applied or SEP applied taping conditions ($\beta = -0.001$, 95% CI: [-0.02 - 0.02], p = 0.90). There was no statistically significant difference (Wald $\chi^2_7 = 0.80$, p = 0.85) in minutes ($\beta = -0.001$, 95% CI: [-0.02 - 0.001], p = 0.52) for either tape condition.

Discussion

A significant improvement in proprioception was observed when either SEP or self-tape were applied. This improvement was maintained for both tape conditions during a netball session. Additionally, there was a learning effect between the 1st and 2nd session of testing (i.e. between the 1st and 5th AMEDA assessment) for both tape conditions. The implications of this study are that either taping condition is acceptable and will equally improve the ability of a pre-elite netball athlete to discriminate small

Table 1

Physical Therapy in Sport 48 (2021) 20-25

AMEDA scores for session one and two.				
	Pre-tape (mean \pm SD)	Post-tape (mean \pm SD)	Post-netball (mean \pm SD)	Post-netball, no tape (mean \pm SD)
Session 1				
Combined	0.64 ± 0.06	0.67 ± 0.06	0.68 ± 0.06	0.69 ± 0.05
Self-applied tape	0.64 ± 0.06	0.66 ± 0.06	0.68 ± 0.07	0.69 ± 0.06
SEP applied tape	0.65 ± 0.06	0.69 ± 0.05	0.68 ± 0.05	0.69 ± 0.04
Session 2				
Combined	0.70 ± 0.05	0.70 ± 0.05	0.69 ± 0.06	0.70 ± 0.05
Self-applied tape	0.70 ± 0.05	0.71 ± 0.05	0.71 ± 0.06	0.71 ± 0.04
SEP applied tape	0.69 ± 0.05	0.69 ± 0.05	0.68 ± 0.06	0.70 ± 0.06

AMEDA (Active Movement Extent Discrimination Apparatus) scores are presented for both testing sessions. Scores for both groups 1) Self-applied tape and 2) Sport and Exercise Physiotherapist (SEP) tape are presented.



Fig. 3. Mean AMEDA (Active Movement Extent Discrimination Apparatus) scores are presented for each taping condition (Self-tape/SEP*-tape) and session number (session 1/ session 2) across the four testing periods (Pre-tape/Post-tape/Post-netball/Post-netball no-tape). *SEP: Sport and Exercise Physiotherapist.

movements at the ankle. It may inform scientists as to the mechanisms for the observed protective effect of taping on ankle sprains.

A significant improvement in proprioception was observed when either SEP or self-tape were applied. This contradicts the systematic review (Raymond et al., 2012) reporting no improvement in proprioception but the four papers reporting no improvement used alternative assessment tools i.e. non-weight bearing threshold to detection of passive motion (Hubbard & Kaminski, 2002; Refshauge et al., 2000, 2009) or joint position reproduction with contralateral side (Schenker, 1989). However, it has been previously noted that different methods of proprioception testing do not correlate strongly to each other when testing the ankle (de Jong et al., 2005), so the selection of the most relevant tool is important to each context. A review comparing three methods for testing proprioception: threshold to detection of passive motion, joint position reproduction and active movement extent discrimination (Han et al., 2016) suggests that the AMEDA is the most effective method for assessing proprioception during active, functional body movements that occur in most sports. Therefore, the findings from this study are more applicable when compared to previous studies investigating the effect of tape on

proprioception.

The current study demonstrates that proprioception is maintained during a netball session, and this improvement is apparently independent of either SEP applied or self-applied tape conditions (p = 0.90). A study undertaken in New Zealand (Waterman et al., 2004) using athletes participating in the two highest grades of club netball showed that proprioception deteriorated during a netball match. It is not clear whether any of the participants in that study wore ankle tape during the proprioception testing or netball match. Our findings are supported by research that showed following a 30 min exercise session consisting of running and basketball, joint position sense was 2.5% worse when their male university student participants were taped compared to 35.5% worse when they did not wear tape (Robbins et al., 1995). There is evidence that reduced proprioception increases the rate of ankle sprain (Attenborough et al., 2017; Witchalls et al., 2012) and the findings of the current study support continuing the common practice of ankle taping for netball sessions to help minimise any potential reduction in proprioception during play or training.

An AMEDA learning effect was found when comparing proprioception scores at the start of the second session compared to the

start of the first session of testing. This is comparable to previous studies using the AMEDA to assess proprioception (Steinberg et al., 2016; Witchalls et al., 2014). This suggests that either the employed familiarisation protocol was inadequate or four exposures to the assessment on the AMEDA results in a positive training effect. However, if the observed improvement in proprioception was solely due to learning effect rather than tape effect, we would expect to see consistent improvement in proprioception at each trial. The AMEDA continues to be a valid and reliable tool for proprioception but it may also be used for its training effect.

Study limitations

A first limitation of this study is the apparent learning effect of the AMEDA. If this could be eliminated it would provide greater certainty in determining the contributing effects of the tape and the netball session. In our study, we pre-planned to minimise this in the design stage by randomising order and using each participant as their own control. Nonetheless, this effect was still observed and future studies may require an additional training session to ensure the initial learning effect on the AMEDA is reduced. A second limitation of our study is our inability to have a passive control where no taping occurred. The netball organisations we worked with had a universal prevention program in place, which encourages athletes to tape at each training or competition event. Therefore, we rather opted to assess whether self-taping or SEP taping is superior, using a superiority randomised controlled trial (RCT) methodology. We are unable to report results comparing self-taping or SEP taping to a no-taping condition.

Conclusions

This study has demonstrated that tape does improve proprioception and it is maintained during a netball session. There was no difference in response whether the tape was self-applied or applied by trained personnel. It was beyond the scope of this paper to determine whether poor proprioceptive performers had a better response to tape or whether athletes with an injury history had reduced proprioceptive scores compared to uninjured athletes as previously described (Long et al., 2017; Steinberg et al., 2019). This information could change the common practice of all netball athletes taping prophylactically to only netball athletes with a history of ankle injury or reduced proprioceptive performance only taping routinely and should be investigated in the future. Nevertheless, our current findings continue to support the use of routine ankle taping for netball pre-elite athletes.

Ethical statement

Ethics approval was obtained from the University of Canberra Human Research Ethics Committee(Ref: 20.181.527) and written informed consent was gained from athletes or their parents/legal guardians before commencement of data collection.

Funding

This work was supported by a joint Australian Institute of Sport and University of Canberra scholarship awarded to author ES for support during her PhD.

Suppliers

^a ClearSky T6 local positioning system: Catapult Sports, Melbourne, VIC, Australia.

^b 5 cm microfoam: 3 M Australia, NSW, Australia.

^c Rigid 38 mm Leuko Premium sports tape: Elastoplast, Beiersdorf Australia Ltd, NSW, Australia.

^d STATA® statistical software: STATA®/IC 15; StataCorp, College Station, TX, USA.

Declaration of competing interest

None declared.

References

- Attenborough, A. S., Sinclair, P. J., Sharp, T., et al. (2017). The identification of risk factors for ankle sprains sustained during netball participation. Physical Therapy in Sport, 23, 31–36.
- Best, R., Mauch, F., Böhle, C., et al. (2014). Residual mechanical effectiveness of external ankle tape before and after competitive professional soccer perfor-mance, *Clinical Journal of Sport Medicine*, 24(1), 51–57. ch, C., Costa, A. D., Stevenson, M., et al. (2002). Sports injury experiences from
- the Western Australian sports injury cohort study. Australian & New Zealand Journal of Public Health, 26(5), 462-467. Halseth, T., McChesney, J. W., Debeliso, M., et al. (2004). The effects of kinesio™
- taping on proprioception at the ankle. Journal of Sports Science and Medicine,
- Han, J., Waddington, G., Adams, R., et al. (2016). Assessing proprioception: A critical
- Han, J., Waddington, C., Adams, K., et al. (2016). Assessing proprioception: A critical review of methods. *Journal of Sport and Health Science*, 5(1), 80–90.
 Hubbard, T. J., & Kaminski, T. W. (2002). Kinesthesia is not affected by functional ankle instability status. *Journal of Athletic Training*, 37(4), 481–486.
 Hume, P. A., & Steele, J. R. (2000). A preliminary investigation of injury prevention strategies in netball: Are players heeding the advice? *Journal of Science and Medicine in Sport*, 3(4), 406–413.
 L. Beformarce, K. M. et al. (2005). Reformance in different in the state of the strategies of the state of the strategies of th
- Meditine in sport, 5(4), 400–413.
 de Jong, A., Kilbreath, S. L., Refshauge, K. M., et al. (2005). Performance in different proprioceptive tests does not correlate in ankles with recurrent sprain. Archives of Physical Medicine and Rehabilitation, 86(11), 2101–2105.
- Langeveld, E., Coetzee, F. F., & Holtzhausen, L. J. (2012). Epidemiology of injuries in elite South African netball players. South African Journal for Research in Sport, Physical Education and Recreation, 34(2), 83–93. Lewkowicz, D. J. (2001). The concept of ecological validity: What are its limitations
- Lewkowicz, D. J. (2001). Interformer concept of colorgical values, what are its initiations and is it bad to Be invalid? Infancy. 2(4), 437–450.
 Long, Z., Wang, R., Han, J., et al. (2017). Optimizing ankle performance when taped: Effects of kinesiology and athletic taping on proprioception in full weight-bearing stance. Journal of Science and Medicine in Sport, 20(3), 236–240.
 Luteberget, L. S., Spencer, M., & Gilgien, M. (2018). Validity of the Catapult ClearSky
- T6 local positioning system for team sports specific drills, in indoor conditions. Frontiers in Physiology, 9(115).
- Products in Physiology, 9(13), Maher, C., & Adams, R. (1996). A comparison of pisiform and thumb grips in stiff-ness assessment. *Physical Therapy*, 76(1), 41–48.
 Raymond, J., Nicholson, L. L., Hiller, C. E., et al. (2012). The effect of ankle taping or bracing on proprioception in functional ankle instability: A systematic review and meta-analysis. *Journal of Science and Medicine in Sport*, 15(5), 386–392.
 Refshauge, K. M., Kilbreath, S. L., & Raymond, J. (2000). The effect of recurrent ankle inversion sprain and taping on proprioception at the ankle. *Medicine & Science & Science*.
- inversion sprain and taping on proprioception at the ankle. Medicine & Science in Sports & Exercise, 32(1), 10–15.
 Refshauge, K. M., Raymond, J., Kilbreath, S. L., et al. (2009). The effect of ankle taping
- on detection of inversion-eversion movements in participants with recurrent ankle sprain. The American Journal of Sports Medicine, 37(2), 371-375.
- Robbins, S., Waked, E., & Rappel, R. (1995). Ankle taping improves proprioception before and after exercise in young men. British Journal of Sports Medicine, 29(4), 242-247.
- Schenker, M. (1989). Tape versus mikros. Scientific Journal of Orienteering, 5(1), 42-47.
- Schulz, K. F., Altman, D. G., Moher, D., et al. (2010). CONSORT 2010 statement: Updated guidelines for reporting parallel group randomised trials. BMC Medicine, 8(1), 18. Semple, S., Esterhuysen, C., & Grace, J. (2012). The effects of kinesio ankle taping on
- postural stability in semiprofessional rugby union players. Journal of Physical
- Therapy Science, 24(12), 1239-1242. Serpiello, F. R., Hopkins, W. G., Barnes, S., et al. (2018). Validity of an ultra-wideband
- Jocal positioning system to measure locomotion in indoor sports. *Journal of Sports Science*, 36(15), 1727–1733.
 Simoneau, G. G., Degner, R. M., Kramper, C. A., et al. (1997). Changes in ankle joint proprioception resulting from strips of athletic tape applied over the skin. *Journal of Athletic Training*, 32(2), 141–147.
- Steinberg, N., Adams, R., Ayalon, M., et al. (2019). Recent ankle injury, sport participation level, and tests of proprioception. *Journal of Sport Rehabilitation*, Mar(3), 1-7.
- Steinberg, N., Tirosh, O., Adams, R., et al. (2015). Does wearing textured insoles during non-class time improve proprioception in professional dancers? Int J Sports Med, 36(13), 1093-1099.
- Steinberg, N., Waddington, G., Adams, R., et al. (2016). Can textured insoles improve ankle proprioception and performance in dancers? Journal of Sports Science, 34(15), 1430-1437.

- Verhagen, E., & Bay, K. (2010). Optimising ankle sprain prevention: A critical review and practical appraisal of the literature. *British Journal of Sports Medicine*, 44, 1082–1088.
- 1082–1088.
 Verhagen, E., van der Beek, A. J., & van Mechelen, W. (2001). The effect of tape, braces and shoes on ankle range of motion. Sports Medicine, 31(9), 667–677.
 Vuurberg, G., Hoorntje, A., Wink, L. M., et al. (2018). Diagnosis, treatment and prevention of ankle sprains: Update of an evidence-based clinical guideline. British Journal of Sports Medicine, 52(15), 956.
 Waddington, G., & Adams, R. (1999). Discrimination of active plantarflexion and inversion movements after ankle injury. Australian Journal of Physiotherapy, 45(1), 7–13.

Physical Therapy in Sport 48 (2021) 20 25

- Waterman, N., Sole, G., & Hale, L. (2004). The effect of a netball game on parameters of balance. *Physical Therapy in Sport, 5,* 200–207.
 Witchalls, J., Blanch, P., Waddington, G., et al. (2012). Intrinsic functional deficits associated with increased risk of ankle injuries: A systematic review with meta-analysis. *British Journal of Sports Medicine, 46*(7), 515–523.
 Witchalls, J. B., Waddington, G., Adams, R., et al. (2014). Chronic ankle instability affects learning rate during repeated proprioception testing. *Physical Therapy in Sport, 15*(2), 106–111.
 Zuluaga, M. (1995). *Sports physiotherapy: Applied science and practice*. Churchill Livingstone.

Chapter 5 Describing intervention context to inform implementation strategies (TRIPP 5)

Published work

This chapter is based on a paper that has been published in the journal, International Journal of Sport Science & Coaching.

Smyth E, Appaneal R, Drew M, et al. Identifying the challenges to implementing a neuromuscular warm-up in pre-elite netball. *Int J Sports Sci Coach*. 2021;16(4):913-924.

Ankle sprains were the most frequently reported injury and ACL ruptures were the most severe at the 2018 17/U & 19/U ANNCs.¹¹⁴ Over the past two decades ankle sprains and knee injuries have had the highest incidence of reported injuries in the sport of netball.^{57,58,115} Kreisfeld et al.,¹¹⁶ reporting on sports injury hospitalisation within Australia in 2011–12, identified that in netball, knee and lower leg injuries were the most frequently injured body region (60% of all netball injuries) with the youngest age group (15–17yrs) having the highest incidence rate.

Netball Australia, Australia's governing netball body, introduced the 'Knee injury prevention for Netballers and Enhance performance and Extend play' (KNEE) program in 2015. The aim of this program is to reduce not only knee injuries but all lower body injuries and to enhance performance by improving movement efficiency, strength and power.

Netball Australia developed a website (https://knee.netball.com.au) with resources to assist with implementing the KNEE program in junior, recreational and elite level netball programs. There is evidence demonstrating that high athlete compliance with an injury prevention program leads to significantly lower injury risk compared to an intermediate or lower level of compliance.¹¹⁷ Each head coach at the 2019 17/U & 19U ANNC was contacted by email in April 2019 and followed up in person if an email response was not received, to determine athlete compliance to the KNEE program. It was revealed that 38% of teams competing at the 2019 17/U & 19U ANNC were not completing the KNEE program as frequently as recommended (i.e., 2–3 times/week). A recent observational study of

community junior netballers also demonstrated a lack of compliance to the KNEE program,⁹⁴ suggesting a review of the KNEE program's implementation is required.

Concept Mapping (CM) was used to harness and organise the ideas of netball coaches, strength & conditioning coaches and physiotherapists about the challenges they faced when implementing the KNEE program. The primary aim of this approach was to recognise gaps within the system implementation of this program and, in turn, develop viable and practical solutions including engagement and education to support compliance and in turn, the impact of the program. The process adopted within this study, partly fulfills the requirements of TRIPP Stage 5.

Co-authors' declaration

Declaration of Co-Authored Publications

For use in theses which include co-authored publications. This declaration must be completed for each coauthored publication and to be placed at the start of the thesis chapter in which the publication appears, or as a preface to the thesis.

UNIVERSITY OF

Declaration for Thesis Chapter Five

Publication: "Identifying the challenges to implementing a neuromuscular warm-up in pre-elite netball using concept mapping"

DECLARATION BY CANDIDATE

 In the case of Chapter Four, the nature and extent of my contribution to the work was the following:

 Nature of Contribution

 Extent of Contributions (%)

Conception and design of the research proposal	60%
Data collection	80%
Analysis and interpretation of the findings	60%
Writing the paper and critical appraisal of content	70%
Corresponding author for journal communication and publication peer-review process.	100%

The following co-authors contributed to the work:

Name	Nature of Contribution	Contributor is also a UC student (Yes/No)
Gordon Waddington	Writing the paper and critical appraisal of content	No
Alanna Antcliff	Conception and design of the research proposal Analysis and interpretation of the findings	No
Phil Newman	Writing the paper and critical appraisal of content	No
Juanita Weissensteiner	Writing the paper and critical appraisal of content	No
Renee Appaneal	Conception and design of the research proposal Analysis and interpretation of the findings Writing the paper and critical appraisal of content	No
Alex Donaldson	Conception and design of the research proposal Analysis and interpretation of the findings Writing the paper and critical appraisal of content	No
Michael Drew	Conception and design of the research proposal Analysis and interpretation of the findings Writing the paper and critical appraisal of content	No

Candidate's Signature

____17/09/2021_ Date

DECLARATION BY CO-AUTHORS

The undersigned hereby certify that:

Declaration of Co-Authored Publications

(1) the above declaration correctly reflects the nature and extent of the

- candidate's contribution to this work, and the nature of the contribution of each of the co-authors. (2) they meet the criteria for authorship in that they have participated in the conception, execution, or interpretation, of
- at least that part of the publication in their field of expertise;
 (2) they meet the conception, execution, or interpretation, or
- (3) they take public responsibility for their part of the publication, except for the responsible author who accepts overall responsibility for the publication;
- (4) there are no other authors of the publication according to these criteria;
- (5) potential conflicts of interest have been disclosed to (a) granting bodies, (b) the editor or publisher of journals or other publications, and (c) the head of the responsible academic unit; and
- (6) the original data are stored at the following location(s) and will be held for at least five years from the date indicated below:

[Please note that the location(s) must be institutional in nature, and should be indicated here as a department, centre or institute, with specific campus identification where relevant.]

 Location(s):
 Australian Institute of Sport

Buen	
Date:	17/09/2021
Alann	a Antcliff
Signer	1: AT
Date:	2/9/21
Phil Ne	wman Pourson
Signed:	pumar
Date:	11/09/2021
Juanita	Weissensteiner
Signed:	Juanita Weissensteiner
Date:	30/08/2021
Renee	Appaneal
Rence /	Appaneal
Rence A Signed: Date:	Appaneal Reversed
Renee / Signed: Date:	Appancal Reversed
Rence A Signed: Date: Alex D	Appancal River 06/09/2021
Rence A Signed: Date: Alex D Signed:	Appancal 06/09/2021 onaldson Alex Donaldson



Identifying the challenges to implementing a neuromuscular warm-up in pre-elite netball using concept mapping



2

of netball,2-5 particularly in 17/U & 19/U pre-ente female netball athletes,3 Pre-elite athletes have been formally recognised as having future high performance potential, often rewarded with financial support but not yet selected for senior international competition or a senior professional team.⁶ A study of Australian sports injury hospitalisations in 2016-17 identified that hip and lower limb injuries were the most frequently injured body region in netball (57% of all netball injuries), and 15-24 year old athletes had the highest incidence rate,7 In pre-elite netball, approximately 2% of participants will have an ACL injury in the National tournament.5 It has been shown in other sports ",9 that injuries sustained by pre-elite athletes can lead to premature sports career termination. These studies highlight the need to successfully implement effective lower limb injury preventative strategies.

Netball Australia, Australia's governing netball body, introduced the Knee injury prevention for Nethallers and Enhance performance and Extend play' (KNEE) program in 2015.10 The KNEE program aims to reduce knee injuries and all lower body injuries and to enhance athletic performance by improving movement efficiency, strength and power. The KNEE program applies principles from evidence-based injury prevention programs in other sports including football,11-13 basketball14 and handball.15 KNEE consists of footwork, strength, balance, landing and agility activities. The KNEE program takes approximately 15-20 mins to complete and should be performed before each court work or match play session at least twice a week. Netball Australia developed a website (https://knee.netball.com.au) with resources to assist with implementing the KNEE program in junior, recreational and elite netball programs. Pre-elite netball athletes are encouraged to use the 'elite' KNEE program

The effectiveness of preventive strategies, like the KNEE program, is determined in part by athlete compliance rates. High compliance leads to significantly lower injury risk compared to intermediate compliance.16,17 Athlete and coach compliance with completing and delivering lower limb injury prevention programs across several sports and settings is low.19-2 Therefore, understanding the program implementation barriers will assist researchers and policy makers to improve the frequency of use of the KNEE program in this pre-clite netball population. Barriers to implementing injury prevention programs have been investi-gated in football, ^{19,20,23-26} basketball¹⁹ and netball.²⁷ The netball study explored the challenges to implementing a program teaching safe-landing techniques to community junior netballers and identified a need to modify the program to make it more suitable for younger players and that more coach education was International Journal of Sports Science & Coaching 0/0)

required.²⁷ To the best of our knowledge, this is the only previous research investigating the implementation of an injury prevention intervention in non-elite netball. Given the high number of ankle and knee injuries in pre-elite netball there is a critical need for the current study to inform the implementation of a program to reduce these lower limb injuries.^{2–5} The aim of this study was to explore the challenges to implementing the KNEE program encountered by netball coaches, strength & conditioning (S&C) coaches and physiotherapists of pre-elite netball teams in Australia.

Methods

Study design: Concept mapping

As we were interested in exploring the applied, contextspecific knowledge of the coaches and support staff of State file regional/provincial representative) 17/U & 19/U female netball teams in Australia, this study used concept mapping (CM).^{28,29} Females were the focus of this study as netball is a female dominated sport and currently there are no pre-elite male representative teams in Australia. CM is a mixed-methods approach incorporating qualitative and quantitative data collection through participants' brainstorming of ideas, sorting of ideas, and rating of ideas. This research method has been previously used for injury prevention research in public health¹⁰ and in sport.^{26,31,32} Participants completed brainstorming, sorting and rating using the Concept Systems Global MAX³³ online platform. The authors also used this platform to analyse the data and to generate concept maps and additional displays of the data.

Brainstorming

Participants were asked to write a brief statement (one idea) to complete the following focus prompt: "A challenge for me to implement the KNEE program with the store [7/U and 19/U reams I work with is" Once contributed, every idea is included in a pooled list of statements generated by those that have participated already and can be viewed by all participants. Participants were encouraged to add multiple statements and review statements made by others to check if an idea had already been contributed.

Statement synthesis

Three authors (ES, AD, RA) synthesised and edited the brainstormed challenges to ensure they were relevant to the project focus, contained a set of unique ideas with each idea represented once, and that each statement was clear. This iterative process involved: 1) identifying and deleting statements not related to the focus

Smyth et al.

prompt; 2) identifying and splitting statements with two or more unique ideas; 3) identifying and editing statements as required to ensure all members of the group agreed on the essential meaning of the statement; and 4) identifying statements that represented the same idea, selecting the statement that best represented the idea and deleting the other similar statements. Synthesising and editing occurred in one meeting until consensus was reached among the three authors. The resulting list of synthesized statements were made available to participants to complete the sorting and rating process.

Statement sorting

Using the Concept Systems Global MAX website, participants sorted the synthesised statements into themed groups. They were instructed to sort the statements into groups according to similarity in meaning and to name each group based on its theme. They were informed that the number of groups people create varies, and 5 to 15 groups was recommended as a guideline to organise the statements.^{33,34} If any statements were not sorted, ES followed up with the participant to ensure this was intentional

Statement rating

Participants were asked to rate each of the synthesised statements by answering the following questions: "On a scale from 0 (least important) to 5 (most important), how important do you think this challenge is to using the KNEE program with the State 17/U and/or 19/U team (s) you work with?"; and "On a scale from 0 (easiest to overcome) to 5 (hardest to overcome) how difficult is this challenge for you to overcome when using the KNEE program with the State 17/U and/or 19/U team(s) you work with?"

Procedures

Netball Australia invited each Australian State netball association (n=8) to participate in this study. Each State association invited all the coaches (n=23), S&C coaches (n=5) and physiotherapists (n=19) working with their State 17/U & 19/U teams to participate. A minimum of four and a maximum of 10 participants were invited from each State. In total, 47 participants were invited to contribute to the atudy.

Netball Australia provided the Stme associations with study information and the link to the online platform. The State associations forwarded this information to participants via email. Participants provided informed consent to participate in the project the first time they registered on the Concept Systems Global MAX web platform. Ethics approval was obtained from the University of Canberra Committee for Ethics in Human Research (No 1551).

Before participating in their first CM activity, participants provided information about their role in netball, the age group they work with, their years of experience working with the State 17/U & 19/U teams, their total years of experience in netball, and their experience with the KNEE program.

After providing the participant information, participants were asked to complete the brainstorming activity. Two reminder emails were sent before this step was closed. All participants, even if they did not participate in the brainstorming, were emailed an invitation to participate in the next two tasks, statement sorting and rating. Two reminder emails were also sent before the sorting and rating tasks were closed. The CM data was collected over three weeks in March 2019, before the 17/U & 19/U 2019 Australian National Netball Championships (ANNC).

Concept mopping analysis

Standard CM data analysis techniques using the Concept Systems Global Max were applied to this study.29 The first step involved constructing a similarity matrix based on the frequency with which each statement was sorted with every other statement by all participants. Multidimensional scaling was applied to the similarity matrix data to place each sorted statement as a separate point on a two-dimensional relational point map'. Hierarchical cluster analysis was then used to partition the statements on the point map into clusters of related statements based on participants' aggregated sorting data ('cluster maps').28 Cluster maps were produced for a 4-cluster solution through to a 9-cluster solution to find the cluster level that retained the most useful detail between clusters while merging the clusters that logically belong together. Once the most appropriate cluster solution was subjectively determined, the research team reviewed every statement in each cluster and if a statement was considered to be a better conceptual fit in an adjacent cluster (i.e. the theme of an adjacent cluster more accurately and appropriately encompassed the idea underpinning the statement), it was reassigned to that cluster.35 All reassignment decisions were guided by cross referencing back to the similarity matrix to check how frequently participants had grouped the statement being considered for reassignment with statements in adjacent clusters, and based on consensus among the research team.

Importance and difficulty rating descriptive statistics were calculated and used to generate a 'gozone'²⁸ in which the mean importance rating for each statement was plotted along the x-axis and the mean difficulty rating for each statement was plotted along

International Journal of Sports Science & Coaching D(0)

the y-axis. Figure I demonstrates how the statements are plotted on the 'go-zone map'. The resulting scatterplot was divided into four quadrants using the allstatement mean of each rating as the axis. The top right quadrant is considered the 'go-zone' as it includes the most important and difficult challenges that should be prioritised to improve implementation of the KNEE program. A correlation coefficient was calculated to demonstrate the relationship between the importance and difficulty of a challenge.

Results

Participants

Thirty-nine of the 47 (83%) invited participants contributed CM data – 38 (97%) participants brainstormed ideas, 30 (77%) participants sorted statements, 30 (77%) participants rated importance of the challenge and 29 (74%) participants rated difficulty to overcome (see Supplementary Figure 1). Twenty-eight participants contributed data to all activities and seven contributed to only the brainstorming task. Characteristics of the 39 participants are summarised in Table 1.

Brainstorming: Challenges identified

Thirty-eight respondents generated 41 statements, which after synthesis and editing, yielded 46 individual statements (Table 2).

Sorting: Concept map

Figure 2 presents the cluster map developed from the multidimensional scaling and hierarchical cluster

analysis. The distance between the points reflects how frequently participants grouped the statement together and considered as a proxy indicator of the degree of similarity between statements. For example, the statements most frequently grouped together were statement 8 ('Optimal technique on knee program components not monitored") and statement 11 ('Difficult to ensure technique is correct across all athletes and exercises'), so they are very close to each other on the cluster map. Conversely, statement 33 ('I modify it to suit our needs') was never grouped together with statement 23 ('Lack of mobile access to videos to enable better correction of technique at point of training'), therefore these statements are very far apart on the cluster map. The stress value, a measure of the degree to which the distances between points on the map are discrepant from the values in the input similarity matrix, was 0.268, which is similar to the average stress value of other CM projects28 and indicates a good fit between the two dimensional point map and the original data as sorted by the participants.

Based on the authors' expert opinion, an eightcluster map was selected as the best representation of the grouped data. The statements within each cluster, with the clusters organised from highest to lowest importance, and the eleven statements that were reassigned by authors to create a better conceptual fit, are presented in Table 2.

Table 2 provides the mean raungs of importance and

difficulty for each of the 46 statements and for the eight

Rating: Perceived importance and difficulty



Figure 1- Go-zone quadrant descriptions

Smyth et al.

Characteristics of participants (n = 39)	Overall on (%)	Brainscormed	Sorted	Rated	Rated difficulty
Gender					_
Male	14 (36%)	11	11	10	11
Female	25 (64%)	25	19	19	18
Current cole					
Physiotherapist	16 (41%)	16	14	15	1.9
Head coach	14 (36%)	14	11	10	10
58C coach	5 (13%)	14	3	3	3
Assistant coach	3 (8%)	3	- SC -	1.1	ĩ
Other Physic coordinator	1 (2%)	1	1.	1	1.
Ase group	. 6.9		~	Contraction of the second	
17/U only	13 (33%)	13	10	ID	10
19/LI only	14 (36%)	14	10	10	9.
17/0 & 19/0	12 (31%)	10.7	10	10	10
17/U or 19/U experience	the formed				
This is my first year	9 (23%)	- 8	6	7	
2-5 years	23 (59%)	.23	19	19	16
5-10 years	7 (18%)	7	5	5	5
Netball experience	0.000				124
-2years	7 (18%)	6	5	é.	5
3-5 years	7(18%)	7	5	X	5
6-10 years	+ (10%)	1.04			
> 10 years	21 (54%)	31	16	15	(9)
KNEE experience					
Always use	12 (3 (%)	12		8	a.
Frequent use	18 (46%)	15	13	14	13
Tried to use	5 (13%)		4	¥	
Never used	+ (10%)				

Table 2. Statements generated during the concept mapping brainstorming process including the cluster in which each statement fits, mean importance and difficulty of overcoming ratings and go-tone graph quadrants for each statement.

		Mean rating		Go-zone
		briportance ⁰	Difficulty	All storements
Cluster	I: Athlete ungagement	3,310	2.48°	
3.	Making sure athletes are committed to executing the KINEE program with correct rechnique.	4.43	2.93	
1'	Making sure athletes are committed to executing the KNEE program warm up.	3.97	2.62	1.0
38	Athletes can sometimes be complacent in carrying the drills out burause they lack the education behind the program.	3.45	2,59	1 A
<u>9</u> 4	The girls sometimes go through it too quickly and don't hold the lands for long enough.	321	1.62	
27	To ensure all aspects of the program are covered as the girls tend pick the aspects they like and not what is necessary.	2.50	2.54	1
26	Some of the achieves are still in the 'learning how po train' stage of development where they struggle with retention of new concepts including a specific warm up.	2.27	2.70	

(continued)

5

б

International Journal	of Sports	Science &	Coaching	0(0)

		Mean rating		Go-zone	
		Importance	Difficulty [®]	quadrant All støtements	
Cluster	2: Supervision/correction of technique	3.030	2.67		
19	Supervising the technique to make sure it is correct.	3.77	2.86	The second se	
45	Getting to all athletes in the time we are doing it -	3.63	3.50		
	detecting and correcting and spending the time to				
	do it correctly - landings and strength in particular.				
в	Optimal technique on longe program components	3.53	2.52	1 A A	
	not monitored.				
0.	Difficult to ensure technique is correct across all	3.50	2.76		
	athletes and exercises.				
18	Given the number of athletes it is hard to review all	2.93	3.10		
	across all movements.	a second			
5 C	Fidelity when I'm not overseeing the training tag in	2.40	2.83	3	
	club environments).				
23	Lack of mobile access to videos to enable better	1.47	1.10		
	corraction of technique at point of training.				
Cluster	1: Time constraints	2.35	2.64		
14	having the time to provide works advestion so that	3.73	2.69	1000	
	the program is implemented as intended, and to				
	improve adherence rates over the long-term.				
40	Lack of time to make sure they are completing the	3.10	2.93		
	eventises correctly each time is challenging		-0.02		
44	Covering all components = a.t. strength agility erc.	3.10	3.00		
19	Time available to spend with each individual athlate	2.67	2.93		
	to choose exercises that will benefit them				
17.	Warking our which parts to use as part of a	5 57	.234		
	method limit with the				
24	Serviced the wall difference in the are	1.00	2.00		
2.9	some coacies into it compare to implement the pro-	1.154	2.00		
Chatter	Athlata technimus	7 709	2 640		
D"	Since warm ups are completed as a strolup may be	2.00	2.91	1000	
14	dende to bound almost of the KARE another	1.00	2.00		
	om control at the among days difference for				
	the individual				
10.1	Most a array all appleas it to provide fundback	5.45	3.95		
	there they say they implement when doing the knew		2.35		
	that they can take implement when doing the knee				
	Optimal technicia or loss and one optimate	2.57	217		
5-1	opened technique en mee program components	+ 47	211		
Charter	E Education	2 249	9-169		
22	Lack of coach/SWT labura advertion contributer to	1.17	2.10	*	
24	Lack of coach/sec/physic education concludes to	2.17	2.34		
	Imited addres resolates about shed position				
£ 10	becondue.	100	510		
	Lack of coach physiolathetd understanding of intent	2.35	210		
	required to get the most quality from the Nivez				
-	program warm up,	2.00	0.72	1000	
	some omounty convincing coaches in the multiple	2.70	K.T.L		
12	actungs the gins train in or the value of it.	1.25	1.40		
14	is the program is not required because bechnique is	1.43	1.96		
	cargeted in the prenadvactivation prior to their				
	warm up so mey have individual reedback on				

(continued)

Smyth et al

Table 7. Cantinued

		Mean rating		Go-zone
		importance ⁰	Difficulty"	All statements
Cluster	6: Support staff resourcing	2.469	2.649	
41	Not having supervised gym sessions (17/U and 19/U) to be able to implement more scrength/power based aspects of the program.	3.00	2.97	
43	Education is presented to coaches and S&C regard ing the K.NEE program but physic is only covered 2 hours/week to cover both 17s and 19s training Never get to see whether it's implemented or how well.	2.80	3.0	x
44"	Implementing it as part of the training session as I only work with the athletes once a week	272	2.83	1 × 1
¥2.	Lack of time with advices at the beginning of cam- paign to implement within warm up drills & gym- program.	2.70	2.38	2
15	I taying the time to properly implement it with tareful supervision of technique while also attending to other roles required of myself as physiotherapist	1.67	2.90	2
30	As physic I only get a maximum of 2-3 hours a week with state teams and I am unable to ensure it is implemented throughout other sessions.	2.63	2.76	1
125	Budget restraints meaning I only can supervise some aspects of the program in the limited time i pet.	2.40	3.07	à
35	I am not present frequently enough to determine If the athletes are consistent with the program.	2.37	2.69	3
5	S&C coaches are not always present at training sat- sions so cannot always be delivered through them.	1.83	1.59	
25	No access to S&C sussions.	1.63	2.14	
Cluster	7: Program flexibility/adaptability	2.02	1.85	
14	Keeping it interesting and fresh for the athletes.	2.30	1.97	9
20	Keeping it interesting with variety for the players.	2.27	2.07	
b."	Lack of individuality of knee program with respect to individual athlete movement screening.	2.07	2.41	1
16	To change it up enough so that it doesn't get borning but ensure it is covering and achieving what it needs to.	1.97	2.10	
33	I modify it to suit our needs,	1.97	1.23	
16	Keeping it functional enough that cosches remain happy	1,60	1.72	
32	I use ideas from the program, but don't use it spe- cifically as I draw on personal experiences and ideas.	1.79	1,45	+
Cluster	3: Coach/support staff prioritisation	1.95	1.810	
10	Athletes emphasis when coming for training sessions is to optimize time spent as a team on court.	2.50	2.17	14
21	Incorporating ball work so it does not take up too	2.13	1.79	
36	As the Physiotherapist it's not my role to go through warm ups etc. with the team.	123	1.48	- +
For all	statements	2.63	2.412	

7



Figure 2. Eight-cluster map of challenges for netball coaches, strength & conditioning coaches and physiotherapists of State 17 U & 19 U athletes implementing the KNEE program.



Figure 3. Go-Zone of challenges for netball coaches, strength & conditioning coaches and physiotherapists of State 17U & 19U athletes implementing the KNEE program.

'Go-zone': Challenges to be prioritised

Figure 3 presents the go-zone for all 46 statements. The 'go-zone' quadrant (top right) contains the statements (n = 19) that received above average ratings on both importance and difficulty. Statements in the go-zone quadrant were rated as the most important and difficult challenges to overcome. As such, it is anticipated that they will require the most support to address. The bottom right quadrant includes statements (n = 4) that were also rated relatively more important but less difficult to overcome and could also be considered as a starting point to achieve early intervention success with minimal resources. The top and bottom left quadrants were rated as less important so the challenges in these two quadrants should be a lower priority. The Pearson Product Moment Correlation coefficient for the gozone graph indicates moderate correlation between 'Importance' and 'Difficulty' (r=0.66).³⁶

Discussion

Ankle sprains and knee injuries have the highest injury incidence in pre-elite netball so improving compliance to the KNEE program in this population is a priority for Netball Australia. This study has identified the challenges encountered by netball coaches, S&C coaches and physiotherapists when implementing the KNEE program in a pre-elite population. The following eight clusters of challenges emerged from this

Smyth et al.

study: "athlete engagement", "supervision and correction of technique", "time constraints", "athlete technique", "education", "support staff resourcing", "program flexibility and adaptability" and 'coach and support staff prioritisation". This highlights that bartiers to implementing the KNEE program are complex and multi-factorial, requiring the application of a wellrecognised implementation sciences framework³⁷ to address these barriers.

According to the Active Implementation Research Network,37 the implementation drivers framework (IDF) can improve implementation of evidencebased interventions. The IDF consists of three sets of implementation drivers, competency drivers (e.g., staff selection, training and performance evaluation), organisational drivers (e.g., systems interventions and facilitative administration) and leadership drivers (e.g., operational and strategic leadership, i.e., planning, procedures, culture, vision, policies and resource aliocation). Application of these drivers can facilitate long-term, high-fidelity intervention implementation Implementing injury prevention strategies in sport is complex¹⁸ and there has been limited use of the IDF in sport, 39 although it has been advocated for 40.41 and is widely recognised in implementation science.42 The authors have used the IDF to inform strategies to overcome the go-zone challenges within each cluster identified within this study.

Focusing on competency drivers can facilitate implementation of the KNEE program by ensuring coaches and support stall have the necessary knowledge, skills and attitude to successfully implement this injury prevention intervention.37 Improving competency can help overcome the go-rone challenges identified in the "athlete engagement', 'supervision and correction of technique', 'athlete technique', 'time constraints' and 'education' clusters (see Table 2) Staff selection is one example of driving competency to identify coaches and support staff that are ready, willing and able to use the KNEE program. If a suitable candidate is identified but they lack KNEE experience, the selection process should assess whether the candidate's philosophy, values and readiness to learn are conducive to learning how to implement the KNEE program in a timely. engaging way with high fidelity. In addition to staff selection, stall training is another competency driver that can be used to educate staff about current injury rates, what the KNEE program is, how it can be adapted and details of its efficacy.¹⁹ Skills can also be developed to assess exercise technique and provide feedback to athletes in ways that lead to improved technique.44,45 Staff training can also educate coaches on how to efficiently integrate the KNEE program into the given time allocated for team training.46

The go-rone challenges within the 'supervision and correction of technique", 'athlete technique", "athlete engagement' and 'time constraints' (see Table 2) can be addressed with organisation drivers. Organisation drivers facilitate changes in the operation of an organisation to support its staff to change their behaviour to deliver an innovation such as the KNEE program." Facilitative administration is an example of an organisation driver to support the staff's use of the KNEE program such as timely access to staff training or completing staff selection well in advance of pre-season training commencing, Additionally, ensuring there is enough 'court time' available to complete the KNEE program to address the challenges in the 'time constraints' cluster is an example of facilitative administration.47 Decision support data system is another example of an organisation driver which involves using data to support decision making. Collecting and communicating injury and intervention effectiveness data to athletes, parents and coaches48 could improve 'athlete engagement' and 'coach and support staff prioritisation".

Leadership is key to successful implementation.49 Leadership drivers involve early identification and engagement of individuals already in executive leadership roles who can make informed decisions to implement the KNEE program and agree to fully participate in the implementation process.³⁷ This can include facilitation of role modelling where a peer-leader led education session is arranged to enhance athlete knowledge of and attitudes towards the KNEE program." Perhaps appointing a highly respected individual to 'train the trainers' can be used to develop high levels of competency and demonstrate how the KNEE program can be implemented within identified constraints such as time.⁵¹ Alternatively, increasing support staff resourcing to deliver the KNEE program may be the best solution.52 These examples of leadership are ways in which the go-zone challenges in 'athlete engagement', 'time constraints' and 'support staff resourcing' clusters (see Table 2) can be mitigated.

Now that the most important and most difficult challenges for implementing the KNEE program have been identified within the go-zone, Intervention Mapping (IM)⁵⁵ can be used to plan strategies to facilitate the adoption, implementation and sustainability of the KNEE program. Intervention Mapping is a six-step process, starting with problem identification and finishing with problem mitigation.⁵⁴ Step 5 of IM focuses specifically on planning the implementation of an evidence-based intervention such as the KNEE program and has been successfully used to improve the implementation of injury prevention programs in community sport.^{55,36}. Applying this process will require Netball Australia to create a planning group of athletes, coaches, support staff and organisation leaders to collaboratively identify desired outcomes and performance objectives for the identified challenges and develop change objectives before designing implementation interventions.

It is important that the strengths and limitations of this study are acknowledged and considered when interpreting its findings. The use of concept mapping is a strength as it is a real-world practical consultative method28 that involves online qualitative structured group data collection from people who are spread across the country and then applies quantitative tools to develop visual maps representing the groups ideas. Thus, we learn from participants individual experiences and gain valuable quantitative data to inform decision making. A limitation of this study is that athletes were not invited to participate due to not wanting to impact their preparations for the 2019 ANNC". Athlete motivation to participate57 in the KNEE program is crucial for program success. If athletes had been included in this study, they may have contributed and prioritised different ideas than coaches and staff, (e.g., 'Program flexibility and adaptability') which would have altered our study outcomes. Athlete insight as end-users is valuable and should be included in future research on KNEE program implementation.

Conclusion

This study identified the following challenges as the most important and difficult to overcome when implementing the Netball Australia KNEE program in 17/1) and 19/U State teams 'athlete engagement', supervision and correction of technique', 'time constraints', 'athlete technique', 'education' and 'support staff resourcing' There is the potential to successfully address these challenges using the implementation driver's framework.35.38 Implementation drivers to consider, include education for athletes, coaches, and organisation leaders to highlight the value of the KNEE program for all involved and provide guidance for program delivery with high fidelity and engagement. Changing position descriptions and performance assessment policy for coaches and support stall and changing the KNEE program structure are also strategies worth trialling. The next step is to gain the athletes perspective on barriers to performing the KNEE program and then applying a systematic approach to implementation planning using a recognised process such as IM Step 5 55,56

Acknowledgements

The anthors would like to thank Netbull Australia and its State Member Organisations for their support and involvement. Finally, we would like to thank the 2019 State 17/U & 19/U netball coaches, strength & conditioning coaches and physiotherapists who participated in this concept mapping process.

Declaration of Conflicting Interests

The author(s) declared no potential conflicts of interest with respect to the research, authorship, and/or publication of this article.

Funding

The author(s) disclosed receipt of the following financial support for the research, authorship, and/or publication of this article. This work was supported by a joint Australiau Institute of Sport and University of Camberra scholarship awarded to author ES for support during her PhD.

ORCID ID

Evin Smyth 🕘 https://ordid.org/0000-0002-8547-324X

Supplemental Material

Supplemental material for this article is available online.

References

- Bergeron MF, Mountjoy M, Armstrong N, et al. International olympic committee consensus statement on youth athletic development. Br J Sport Med 2015, 49: 843–851
- Finch C, Costa AD, Stevenson M, et al. Sports injury experiences from the Western Australian sports injury cohort study. *Aust N Z J Public Health* 2002; 26: 462–467.
- 3 Hume PA and Steele JR. A preliminary investigation of injury prevention strategies in netball: are players heading the advice? J Sci Med Sport 2000; 3: 406–413.
- Langeveld E. Coetzee FF and Holtzhausen LJ. Epidemiology of injuries in elite South African netbull players. S Afr J Res Sport Phys Educ Recreation 2012; 34: 83-93.
- 5 Smyth EA, Piromalli L. Antaliff A, et al. A prospective study of health problems at the 2018 17/U and 19/U Australian national netball championships with comparison of surveillance methodology. J Sci Med Spari 2020; 23: 215-221.
- 6 Gulbin JP, Croser MJ, Morley EJ, et al. An integrated framework for the optimisation of sport and athlete development: a practitioner approach. J Sports Sci 2013; 31: 1319–1331.
- Kreisfeld R and Harrison J. Hospitalised sports interv in Australia 2016-2017, Cauberta: AIHW, 2020.
- Huxley DJ. O'Connor D and Healey PA. An examination of the training profiles and injuries in elite youth track and field athletes. *Eur J of Sport Sci* 2014; 14: 185–192.
- Grygorowicz M, Michaiowska M, Jurga P, et al. Thirty percent of female footballers terminate their careers due to injury: a retrospective study among polish former players. J Sport Robubil 2019: 28: 109-114.

Smyth et al.

- Netball Australia. The KNEE program, https://knee.net ball.com.ou/ (accessed 30 January 2019)
- Owoeye OB, Akinbo SR, Tella BA, et al. Efficacy of the FIFA 11+ warm-up programme in male youth football: a cluster randomised controlled triat. J Sports Sci Med 2014; 13: 321-328.
- Silvers-Granelli H, Mandelbaum B, Adeniji O, et al. Efficacy of the FIFA 11+ injury prevention program in the collegiate male soccer player. Am J Sports Med 2015; 43: 2628–2637.
- Silvers-Granelli HJ, Bizzini M, Arundale A, et al. Does the FIFA 11+ injury prevention program reduce the incidence of ACL injury in male soccer players? Clin Orthop Relat Res 2017; 475: 2447–2455.
- 14. LaBella CR, Huxford MR, Grissom J, et al. Effect of wearonruscular warm-up on injuries in female soccer and backetball athletes in urban public high schools cluster randomized controlled trial. Arch Pediatr Adolesc Med 20(1): 165: 1033-1040.
- Myklebust G, Engebreisen L, Brackken IH, et al. Prevention of anterior cruciate ligament injuries in female team handball players: a prospective intervention study over three seasons. Clin J Sport Med 2005; 13: 71–78.
- Soligard T, Nilstad A, Steffen K, et al. Compliance with a comprehensive warm-up programme to prevent injuries in youth football. Br J Sports Med 2010; 44, 787–793.
- Hagglund M, Atroshi I, Wagner P, et al. Superior compliance with a neuromuscular training programme is associated with fewer ACL injuries and fewer acute knee injuries in female adolescent football players; secondary analysis of an RCT, Br J Sports Med 2013; 47: 974–979.
- Lindblom H, Walden M, Carlford S, et al. Implementation of a neuromuscular training programme in female adolescent football: 3-year follow-up study after a randomised controlled trial. Br J Sports Med 2014; 48: 1425-1430.
- Norcross MF, Johnson ST, Boybjerg VE, et al. Factors influencing high school coaches' adoption of injury prevention programs. J Sci Med Sport 2016, 19: 299–304.
- O^{*}Brien J and Finch CF. Injury prevention exercise programs for professional soccer: understanding the perceptions of the end-users. *Clin J Sport Med* 2017; 27: 1–9.
- Morgan EA, Johnson ST, Boybjerg VE, et al. Associations between player age and club soccer coaches' perceptions of injury risk and lower extremity injury prevention program use. Int J Sports Sci Coach 2018; 13: 122-128.
- Junge A, Lamprecht M, Stamm H, et al. Countrywide rampaign to prevent soccer injuries in swiss another players. Am J Sports Med 2011; 39: 57-63.
- Joy EA, Taylor JR, Novak MA, et al. Factors influencing the implementation of anterior cruciate ligament injury prevention strategies by girls soccer coaches. J Strength Cond Rev 2013; 27: 2263–2269.
- Lindblom H, Carlford S and Higglund M. Adoption and use of an injury prevention essercise program in female football: a qualitative study among coaches. Seand J Med Sci Sports 2018, 28: 1295-1303.

- O'Brien J, Young W and Finch CF. The delivery of injury prevention exercise programmes in professional youth soccer; comparison to the FIFA 11. J Sci Med Sport 2017; 20: 26-31.
- Donaldson A, Callaghan A, Bizzini M, et al. A concept mapping approach to identifying the barriers to implementing on evidence-based sports injury prevention programme. *Inj Prev* 2019; 25: 244-251.
- Saunders N, Otago L, Romiti M, et al. Coaches' perspectives on implementing an evidence-informed injury prevention programme in junior community netball. Br J Sports Med 2010: 44 1128-1132.
- Kane M and Trochim W. Concept mapping for planning and evaluation. Thousand oaks. CA: Sage Publications, 2007
- Trochim WM and McLinden D. Introduction to a special issue on concept mapping. Eval Program Plann 2017; 50: 166–175.
- Snider CE, Kirst M, Abubakar S, et al. Community-based participatory research: development of an emergency department-based youth violence intervention using concept mapping. *Acad Emerg Med* 2010; 17: 877–885.
- Ageberg E, Bunke S, Lucander K, et al. Facilitators to support the implementation of injury prevention training in youth handball: a concept mapping approach. Scand J Med Sci Sparse 2019; 29: 275-285.
- Visek AJ, Achrati SM, Mannix H, et al. The fun integration theory: ioward sustaining children and adolescents sport participation. J Phys. Act Health 2015, 12: 424–433.
- Concept Systems Incorporated: Concept systems global max Trhaca, NV: Author, 2019.
- Rosas SR and Kane M. Quality and rigor of the concept mapping methodology: a pooled atudy analysis. Eval Program Plann 2012; 35: 236-245
- Munnes M. Using concept mapping for planning the implementation of a social technology. Eval Program Plann 1989; 12: 67–74.
- Liamputtong P. Rewarch methods in health: foundations for evidence-based practice. 2nd ed. Melbourne: Oxford University Press, 2010.
- Fixsen DB, K and Van Dyke M. Implementation praction & science. 1st ed. North Carolina: Active Implementation Research Network, 2019.
- Bekker S and Clark AM. Bringing complexity to sports injury prevention research: from simplification to explanation. Br J Sports Med 2016; 50: 1489-1490.
- Frank BS, Register-Mihalik J and Padna DA. High levels of coach intent to integrate a ACL injury prevention program into training does not translate to effective implementation. J Sol Med Spart 2015, 18: 400-406.
- Donaldson A and Finch CF. Applying implementation science to sports injury prevention. Br J Sports Med 2013: 47: 473-475.
- Donaldson A, Newton J, McCrory P, et al. Translating guidelines for the diagnosis and monagement of sportsrelated concussion into practice. Am J Effective Med 2016; 10: 120-135.
- Blase K., Fissen DL and Van Dyke MK. Implementation practice and science. Independently Published, 2019.

- Mawson R. Creech MJ, Peterson DC et al. Lower limb injury prevention programs in youth succer: a survey of coach knowledge, usage, and barriers. J Exp Orthop 2018; 5: 43.
- 44 Forington LV, Donaldson A, Lathlean T, et al. When 'just doing it' is not enough assessing the fidelity of player performance of an injury prevention exercise program. J Sci Med Sport 2015; 18: 272-277.
- 45. Ljunggren G, Perera NKP and Hagglund M. Inter-rater reliability in assessing exercise fidelity for the injury prevention exercise programme knee control in youth football players. Sports Med Open 2019; 5: 55.
- 46 Whalan M, Lovell R, Steele JR, et al. Rescheduling part 2 of the 11+ reduces injury burden and increases compliance in semi-professional football. Scand J Med Sci Sports 2019, 29: 1941–1951.
- 47 Tucker AC and Singer S. The effectiveness of management-by-walking-around: a randomized field study. Prod Oper Manag 2015; 24: 253-271.
- 48: Ekstrand J, Lundqvist D, Dayison M, et al. Communication quality between the medical learn and the head coach/manager is associated with injury burden and player availability in clite football clubs. Br J Sports Med 2019; 53: 304–308.
- Vries M. Leadership group coaching in action: the Zen of greating high performance teams, Acad Manage Perspect 2005, 19: 61-76.
- 30. White P. Doualdson A and Finch DF. But can someone like me do it? The importance of appropriate role

modelling for safety behaviours in sports injury prevention. Br J Sports Med 2016: 50: 569-570.

- 51 Padua DA, Frank B, Donaldson A, et al. Seven steps for developing and implementing a preventive training program: lessons learned from JUMP-ACL and beyond. *Clin Sports Med* 2014; 33: 615–632.
- 52 Talpey S and Siesman E. Sports injury prevention: the role of the strength and conditioning coach. Strength Condition J 2017; 39: 14–19.
- Kok G. A practical guide to effective behavior change: how to apply theory-and evidence-based behavior change methods in an intervention. *Eur Health Psychol* 2014; 16: 156–170.
- Bartholomew LK, Parcel GS, Kok G, et al. Planning health promotion programs: an intervention mapping approach. 3rd ed. San Francisco, CA. Jossey-Bass, 2006.
- 55 Donaldson A, Lloyd DG, Gabbe BJ, et al. We have the programme, what next? Planning the implementation of an injury prevention programme. *Inf. Prev.* 2017; 25: 273-280.
- Domildson A and Poulos RG. Planning the diffusion of a neck-injury prevention programme among community rugby union coaches. Br J Sports Med 2014; 48: 151–159.
- 57 Bolling C, Delfino Barboza S, van Mechelen W, et al. Letting the cat out of the bag athletes, coaches and physiotherapists share their perspectives on injury prevention in elite sports. *Bt & Sports Med* 2020: 54: 871-877.

12

Chapter 6 Evaluating effectiveness of preventative measures in an implementation context (TRIPP 6)

Paper submitted for publication

This chapter is based on a paper that has been submitted for publication in the journal Translational Sports Medicine. Ankle sprains, low back pain and foot blisters were the most frequent injuries reported at the 2018 17/U & 19/U ANNC, while ACL ruptures and concussion resulted in the greatest sports incapacity. Prevention strategies such as the KNEE program are in place to address the prevalence of ankle sprains and ACL rupture. Netball Australia is addressing the incidence of concussions within its related policy and procedures. Strength and conditioning coaches and physiotherapists are addressing the incidence of lower back pain at an individual level. However, foot blister prevention has never been prioritised at this level. The current study assessed the effectiveness of a foot blister prevention strategy within the real-world setting of the 2019 17/U & 19/U ANNC. Additionally, medical attention and self-report injury surveillance conducted at the 2018 17/U & 19/U ANNC was repeated at the 2019 17/U & 19/U ANNC, resulting in improved understanding of the pre-elite netball athlete injury profile.

Co-authors' declaration

Declaration of Co-Authored Publications

CANBERRA

For use in theses which include co-authored publications. This declaration must be completed for each coauthored publication and to be placed at the start of the thesis chapter in which the publication appears, or as a preface to the thesis.

Declaration for Thesis Chapter Six

Publication: "Injury surveillance at the 17/U & 19/U Australian National Netball Championships and the effect of a foot blister prevention intervention"

DECLARATION BY CANDIDATE

In the case of Chapter Four, the nature and extent of my contribution to the work was the following:

 Nature of Contribution
 Extent of Contributions (%)

Conception and design of the research proposal	80%
Data collection	50%
Analysis and interpretation of the findings	70%
Writing the paper and critical appraisal of content	70%
Corresponding author for journal communication and publication peer-review process.	100%

The following co-authors contributed to the work:

Name	Nature of Contribution	Contributor is also a UC student (Yes/No)
Gordon Waddington	Conception and design of the research proposal Data collection Analysis and interpretation of the findings Writing the paper and critical appraisal of content	No
Alanna Antcliff	Conception and design of the research proposal Analysis and interpretation of the findings	No
Phil Newman	Writing the paper and critical appraisal of content	No
Juanita Weissensteiner	Writing the paper and critical appraisal of content	No
Liam Toohey	Analysis and interpretation of the findings	No
Ruth Fazakerley	Design of intervention	No
Laura Piromalli	Conception and design of the research proposal Analysis and interpretation of the findings	No
Theo Niyosenga	Analysis and interpretation of the findings	No
Michael Drew	Conception and design of the research proposal Analysis and interpretation of the findings Writing the paper and critical appraisal of content	No



___17/09/2021____

Candidate's Signature

_____Date
Declaration of Co-Authored Publications DECLARATION BY CO-AUTHORS



The undersigned hereby certify that:

- (1) the above declaration correctly reflects the nature and extent of the candidate's contribution to this work, and the nature of the contribution of each of the co-authors.
- (2) they meet the criteria for authorship in that they have participated in the conception, execution, or interpretation, of at least that part of the publication in their field of expertise;
- (3) they take public responsibility for their part of the publication, except for the responsible author who accepts overall responsibility for the publication;
- (4) there are no other authors of the publication according to these criteria;
- (5) potential conflicts of interest have been disclosed to (a) granting bodies, (b) the editor or publisher of journals or other publications, and (c) the head of the responsible academic unit; and
- (6) the original data are stored at the following location(s) and will be held for at least five years from the date indicated below:

[Please note that the location(s) must be institutional in nature, and should be indicated here as a department, centre or institute, with specific campus identification where relevant.]

Location(s):	Australian Institute of Sport						
	1						
Gordon Wadd	lington	Ruth Fazakerley					

Signed:

Date:

Signed:

Date:

Theo Niyonsenga

17 -

20

	17/00/2021
Date:	17/03/2021
Alanna An	teliff A

Signed: Date:

Phil Newman Signed: Date: 11/09/2021

Juanita	Weissensteiner	
Signed:	Juanita weissensteiner	_
Date:	30/08/2021	

Liam To	ohey A	
Signed:	0-	
Date:	3/09/2021	

Laura Piron	nam
Signed:	LPiromalli
Date:	31/8/2021
Date:	31/8/2021

Theo Niyonseng

07/09/2021

Michael Drew	
Signed:	Michael Drew
Date:	17/09/2021

Injury surveillance at the 17/U & 19/U Australian National Netball Championships and the effect of a foot blister prevention intervention

Abstract

Objective: Surveillance of injury/illness at the 2019 17/U & 19/U Australian Netball National Championships (ANNC) and evaluation of the effectiveness of a foot blister prevention strategy. **Design:** Prospective observational cohort study with intervention.

Methods: One hundred and ninety-two athletes were observed during a six-day tournament. Medical attention injuries were recorded prospectively by physiotherapists and athlete self-reported injury data was collected. Additionally, a foot blister prevention strategy was applied six weeks prior to tournament commencement.

Results: Ninety-five medical attention injuries were sustained by 73 athletes, at an incidence rate of 82.5 injuries/1000 player competition hours. Ankle sprains (n=16) and lumbar pain (n=12) had the highest incidence of medical attention injuries. Thirty per cent of athletes started the tournament with a self-reported injury/illness. Twelve sports incapacity injuries were recorded, with concussion (n=5) and ACL rupture (n=4) the most frequent. Team physiotherapists recorded eight foot blisters in 2019 and eleven foot blisters in 2018, this difference was not statistically significant.

Conclusions: Ankle sprains were the highest medical attention injury recorded in pre-elite netball athletes. This has not changed over the past 30 years, indicating current ankle sprain prevention strategies are not effective. Our foot blister prevention strategy was not effective for preventing foot blisters in pre-elite netball athletes.

Keywords: Athletic injuries; Surveillance; Sports medicine; Sprains and strains; Foot blisters; Pre-elite netball population

Introduction

Netball is the most popular team sport in Australia for women and girls, but participation decreases after the age of 15 with poor health or injury the second greatest reason for drop-out.¹¹⁸ Australia has remained on top of the world netball rankings over the last five years.¹¹⁹ For sustained success, Netball Australia seeks to continue junior player development, with a critical component focusing on maintaining the health of athletes through injury and illness prevention strategies.

Injury surveillance has previously been conducted at the 2018 17/U & 19/U Australian Netball National Championships (ANNC).¹¹⁴ These championships are an opportunity for each Australian state to compete against the others and for Australian national squads to be selected. Ankle sprains, low back pain and foot blisters were the most frequent injuries reported, with anterior cruciate ligament (ACL) ruptures the most severe, at the 2018 ANNC.¹¹⁴ Ankle sprains and knee injuries have consistently had the highest incidence rate in netball over the past 30 years.⁵⁶⁻⁵⁹ Historical prevention programs targeting ankle and knee injuries have previously been implemented;^{74,75} however, foot blisters have not yet been targeted.

Foot blisters are a common occurrence in netball.^{55,58,114} Despite this, there is no evidence of foot blister prevention strategy implementation in the netball population. Of the many netball injury surveillance studies,⁵⁴ only two have reported foot blister injury rates.^{55,58} Hopper, in 1986 recorded 45 foot blisters (28.5%) across a 14-week netball season,⁵⁵ and Hume & Steele in 2000 recorded 2 foot blisters (2%) during a three-day tournament.⁵⁸ At the 2018 17/U & 19/U ANNC, foot blisters continued to be problematic, causing 11 athletes to require medical attention and resulting in sports incapacity for two athletes.¹¹⁴ Foot blister prevention studies have previously been conducted with military personnel and endurance runners.^{120,121} It is generally accepted that foot blisters can be easily prevented with correct footwear and foot hygiene.^{122,123} We tested this hypothesis by implementing a prevention strategy at the 2019 17/U & 19/U ANNC.

The objectives of this study are to:

- Describe the type and mechanism of injuries that occurred across the six-day 2019 17/U & 19/U ANNC and during the four weeks following the tournament.
- 2) Compare injury rates between the 2018 and 2019 17/U and 19/U ANNC.
- Evaluate the effectiveness of a blister prevention strategy for reducing the number of foot blisters reported at the 2019 ANNC compared to the 2018 ANNC.

Methods

Participants & Design

All athletes (n=192; 17/U athletes = 96, 19/U athletes = 96) competing at the 17/U & 19/U ANNC were invited to participate in the present study. Sixteen female teams competed in the six-day tournament (11–16 April 2019). Each team played nine matches, with two matches played on days one, three and four, while one match was played on days two, five and six. A total of 144 matches were played, with each match consisting of ten-minute quarters. Athletes were not exposed to training during the competition due to the congested match schedule. The methods for the 2018 17/U & 19/U ANNC injury surveillance collection have previously been published¹¹⁴ and were repeated in 2019. The Strengthening the Reporting of Observational studies in Epidemiology in Sports Injury and Illness Surveillance (STROBE-SIIS) was followed.⁴⁷ Ethics approval was obtained from the Australian Institute of Sport Human Research Ethics Committee (Approval number: 20180404). All participants or their parents/guardians provided consent via their individual online Athlete Management System (AMS; Smartabase, Fusion Sport, Brisbane, Australia) account.

Intervention strategy

Six weeks prior to the ANNC, every athlete received a foot blister prevention and management pack, frequently used by the Australian Institute of Sport Medical Department, and an advice sheet (Appendix 17) developed by one of the authors (RF). Each team physiotherapist was asked to present the information in the advice sheet to their athletes while distributing the prevention packs. Athletes were also encouraged to not wear new netball shoes within four weeks of the tournament start.

Outcome measures

Injuries were defined in three ways: 1) self-reported, 2) medical attention and 3) sports incapacity. Athlete self-reported injuries were defined as sensations that are interpreted by an athlete as being indicators of abnormal body function.¹²⁴ All athletes were asked to complete the Oslo Sports Trauma Research Centre Questionnaire on Health Problems (OSTRC-H)^{20,125} via their AMS accounts at four time points: 1) the start of the tournament, 2) end of the tournament, 3) one week post-tournament and 4) four weeks post-tournament. This questionnaire asks the athlete to report any injuries or illnesses experienced in the past week in terms of participation, training volume, effect on performance and symptoms. The questionnaire allows for multiple injuries and illnesses to be reported.

A medical attention injury was defined as loss or abnormality of bodily structure or functioning that following examination is diagnosed by a clinical professional as a medically recognised injury.¹²⁴ Team physiotherapists were asked to record all medical attention injuries using a standardised injury record form within the AMS. A four-character Orchard Sports Injury Classification System 10 (OSICS-10.1)¹²⁶ injury diagnosis code was assigned prospectively by each team physiotherapist. The lead author (ES) met with all team physiotherapists prior to the tournament to explain the injury record system and clarify any areas of uncertainty. All physiotherapists were encouraged to complete each injury record on the day of injury. Sports incapacity injuries were defined as an injury that required an athlete to miss any duration of match play during the tournament or where the athlete reported a reduction in participation as assessed by the OSTRC-H.¹²⁴

Exposure was quantified according to the 'athlete participation' method, which defines participants as those that are on the game roster whether they played or not.¹²⁷ The number of athletes per team (n=12) was multiplied by the number of teams (total=16, 17/U=8, 19/U=8), number of matches (n=9) and the number of minutes per match (n=40) and then divided by the number of minutes per hour (n=60). This resulted in a total of 1,152 player competition hours, or 576 player competition hours for each of the 17/U or 19/U athlete cohorts.

Statistical analysis

Descriptive analysis was performed for the 17/U and 19/U athlete cohorts at the 2019 ANNC. The differences between age groups for medical attention injuries (including foot blisters), sports incapacity injuries and injury classification were measured by calculating incidence rate ratios (IRR) and Fisher's exact test to test for significance. ANNC competition years were also compared by calculating IRR for medical attention injuries and specific injury diagnoses. Injury incidence rates were calculated using the formula: number of new injuries/number of exposures per 1000h.⁴⁰ Incidence rate ratios were also calculated to compare age groups and ANNC competition years. Analysis of the OSTRC-H questionnaire outcomes across the four dates completed by the 2019 cohort utilised the McNemar chi-square test, as this involved comparison of non-independent groups (repeated sampling). Comparison of the OSTRC-H outcomes between the 2018 and 2019 ANNC cohorts was performed using the Pearson chi-square test.

Results

Medical attention and sports incapacity injuries (including foot blisters)

Ninety-five medical attention injuries were sustained by 73 athletes over a six-day period, at an incidence rate of 82.5 injuries/1000 player competition hours. Sports incapacity incidence rate was 10.4/1000 player competition hours. There was no significant difference between 17/U & 19/U medical attention injury rates (17/U, n=47; 19/U, n=48; IRR=0.98; 95% CI: 0.64–1.50; p=0.92). Lateral ankle ligament sprains (n=16, 16.8%), lumbar pain (n=12, 12.6%) and foot blisters (n=8, 8.4%) were the most frequently recorded injuries. There were 12 (12.6% of total injuries) sports incapacity injuries recorded, with concussion (n=5) and ACL rupture (n=4) the most common. None of the foot blisters resulted in sports incapacity. There were twice as many 17/U sports incapacity injuries compared to 19/U sports incapacity injuries; however, this was not statistically significant (17/U, n=8, 8.4%; 19/U, n=4, 4.2%; IRR=2.00; 95% CI: 0.54–9.08; p=0.27). Ninety-five per cent (n=90) of injuries were classified as new injuries and occurred 18 times more frequently than recurrent injuries (n=5, 5%; IRR=18.00; 95% CI: 7.43–56.78, p<0.01). Details of the mechanism of injury, injury onset and classification of injury according to each body area are outlined in Table 1.

2018 versus 2019 ANNC

Compared to the 2018 ANNC¹¹⁴ results, there was no difference in total medical attention injuries in 2019 (2018, n = 103; 2019, n = 95; IRR=1.08; 95% CI: 0.81-1.45; p=0.57). There were also similar number of 19/U injuries (2018 19/U, n=64; 2019 19/U, n=48; IRR=1.33; 95% CI: 0.90-1.98; p<0.13) and 17/U injuries across both years (2018 17/U, n=39; 2019 17/U, n=47; IRR=0.83, 95% CI: 0.53-1.30; p=0.39). No reduction in foot blisters was observed (2018, n=11; 2019, n=8; IRR=1.38; 95% CI: 0.50-3.94; p=0.50). There were also similar frequency of ankle sprains (2018, n=14; 2019, n=16; IRR=0.88; 95% CI: 0.40-1.91; p=0.72), lumbar pain (2018, n=10; 2019, n=12; IRR=0.84; 95% CI: 0.32-2.10; p=0.68), ACL ruptures (2018, n=3; 2019, n=4; IRR=0.75; 95% CI: 0.11-4.43, p=0.73) and concussion (2018, n=3; 2019, n=5; IRR 0.60; 95% CI: 0.09-3.08, p=0.51).

Health problems questionnaire

The response rate for the four OSTRC-H questionnaires (1: at the start of the tournament, 2: end of the tournament, 3: one-week post-tournament, and 4: four weeks post-tournament) was greater in 2019 (93%, 99%, 78% and 74% respectively) than in 2018 (90%, 76%, 41% and 34% respectively). Of the 175 athletes that completed the first and second OSTRC-H in 2019, 30% (n=53) started the tournament with a self-reported injury/illness. There was a significant increase in the number of athletes (n=74, 42%) who self-reported an injury/illness at the end of the tournament compared to the start (χ^2 =11.76, p<0.01; OR= 2.11; 95% CI: 1.19–3.85). Ten (7% of respondents) of these athletes could not participate at all four weeks following the 2019 ANNC, whereas three athletes (2% of respondents) reported not being able to participate in netball during the week prior to the 2019 ANNC.

Forty athletes (28% of respondents) in 2019 and 22 athletes (34% of respondents) in 2018 reported training at a reduced volume four weeks following the tournament due to self-reported injury/illness (χ^2 =0.69, p=0.41). Sixty-nine athletes (36% of respondents) in 2019 and 55 athletes (38% of respondents) in 2018 thought their performance was affected by injury/illness during the tournament (χ^2 =0.07, p=0.80).

Discussion

Injury surveillance has now been prospectively investigated across two years of the 17/U & 19/U ANNC. The combined 17/U and 19/U injury incidence rate was lower in 2019 (82.5/1000 player hours) compared to 2018 (89.4/1000 player hours), but this was not statistically significant.¹¹⁴ Lateral ankle ligament sprains, lumbar pain and foot blisters were the most frequently recorded injuries in 2018 and 2019.¹¹⁴ The number of sports incapacity injuries fell from 22 in 2018¹¹⁴ to 12 in 2019; however, the number of ACL ruptures increased by one (2018, n=3; 2019, n=4) and concussions slightly elevated from three to five.¹¹⁴ Further prevention strategies are indicated to address these significant ongoing health issues.

The International Olympic Committee (IOC) consensus statement on youth athletic development states: 'No youth athlete should compete – or train or practice in a way that loads the affected injured area, interfering with or delaying recovery – when in pain or not completely rehabilitated and recovered from an illness or injury'.⁴ Thirty-one per cent of athletes started the tournament with a self-reported injury/illness, an increase from 27% in 2018.¹¹⁴ Prior to these two studies, the OSTRC-H has not been utilised to record injury data for pre-elite athletes competing in a tournament, making it difficult to draw conclusions about whether these figures are common in this population. Nevertheless, our results indicate that a concerted effort needs to be made to heed the advice of the IOC⁴ and improve the health of athletes attending the tournament. Improved awareness of pre-elite injury prevention and management needs to occur across the system, including state bodies, clubs, coaches, managers, parents and athletes.

There was no meaningful reduction in foot blisters in 2019 (n=8, 8.4%) compared to 2018 (n=11, 10.7%). There was no incidence of sports incapacity due to foot blisters in 2019. A higher frequency of foot blisters was recorded at the 2018 & 2019 17/U & 19/U ANNC compared to previously reported injury data from a three-day New South Wales netball tournament in 1995 (n=2, 2% of total injuries).⁵⁸ It is difficult to claim a reduction of three blisters across six days and 192 athletes at the 2019 17/U & 19/U ANNC is of clinical significance; however, it is promising to observe a reduction in this injury rather than an increase. This strategy could be improved by involving the team physiotherapists in the research project earlier to increase their engagement and highlight the important role they play in

educating athletes about foot blister prevention. Continuing to implement this simple foot blister prevention strategy is advised.

There were two additional concussions at the 2019 17/U & 19/U ANNC (n=3) compared to the 2018 17/U & 19/U ANNC (n=5) but this difference was not statistically significant. Concussion incidence appears to have increased over the past 20 years. In 1998, Finch et al.¹¹⁵ reported a frequency of 1.6% (n=4) over two consecutive five-month seasons and in 2014, there was one concussion recorded from 154 netball injuries presenting to an emergency department across a 2-year period.¹²⁸ Finch et. al.¹²⁹ reported an 8% annual increase in concussions sustained by netball athletes from 2002–03 to 2010–11. Possible explanations for this apparent increase in concussions are that historically they were not recognised and diagnosed, or that changes in game style over time mean that more collisions now occur in netball,^{130,131} resulting in more concussions. Improving concussion injury surveillance and understanding mechanisms of injury are important steps towards preventing concussion and should be prioritised, as it is an injury that results in sports incapacity and can cause long-term sequelae.

Approximately one-third of athletes were training at a reduced volume due to an injury or illness following the 2018 and 2019 17/U & 19/U ANNC. In 2018 and 2019, five (8%) and ten athletes (7%) respectively could not participate at all, four weeks post tournament. At the start of the tournament, two (1%) athletes in 2018 and three (2%) athletes in 2019 could not participate in the week prior to the tournament. Athletes are required to play nine matches in six days at possibly a higher intensity than they would normally play. Preparation for this workload is very difficult when these athletes have other competing commitments such as school, university or work.^{14,132} A spike in training load can increase risk of injury or illness for the following four weeks.¹³³ This data warrants a review of the 17/U & 19/U ANNC format – perhaps a system where teams are divided into two pools to limit the number of fixtures could be considered to reduce health consequences. Additionally, awareness and utilisation of recovery modalities^{134,135} during this multiday tournament could be reviewed.

Two injury surveillance methods, athlete self-report and medical attention, were introduced in 2018 at the 17/U & 19/U ANNC and repeated in 2019. The athlete self-report response rate improved to be above 70% across all four OSTRC-H collection periods in 2019, compared to 2018¹¹⁴ when the response

rate was below 50% for the two post-tournament OSTRC-H. There were also fewer missing data points for medical attention injuries in the second year of injury surveillance. For example, in 2018, 32 (31%) injury records did not include mechanism of injury data; in 2019, this decreased to 10 (11%) incomplete injury records. Establishing quality injury surveillance takes time and commitment by all personnel working within high-performance sport and is a critical component of injury prevention. We were able to implement injury surveillance across two 17/U & 19/U ANNC due to the strong support received from Netball Australia, highlighting the critical role National Sporting Organisations (NSOs) play in injury prevention. Our method for obtaining injury data from a pre-elite level national tournament is a good example of how other NSOs may also employ injury surveillance in their pre-elite populations at similar events.

We conducted a prospective intervention cohort study and compared results to a study conducted one year earlier.¹¹⁴ Several limitations of this study are apparent. Firstly, there was no monitoring to ensure all athletes received their blister prevention packs, read their advice sheets, or followed the 'no new shoe' guidelines. This was due to the difficulties in measuring reach and adoption in a decentralised system and the inability of the research team to have direct contact with the athletes prior to the competition. In future, fidelity of the foot blister prevention strategy could be checked by researchers at the start of the tournament when athletes are asked to complete their first HPQ. Secondly, employing the OSTRC-H questionnaire potentially introduces self-reporting bias.

Conclusion

A high number of medical attention injuries (n=95) occurred at the 2019 17/U & 19/U ANNC. Ankle sprains are the most frequent medical attention injury in pre-elite netball athletes. This has not changed over the past 30 years, indicating current injury prevention interventions for ankle sprain are not effective. The foot blister prevention strategy trialled in this study did not result in a statistically significant reduction in foot blisters. Ongoing injury prevention work is required to not only reduce the number of foot blisters but also the overall number of injuries and number of significant injuries such as ankle sprain, ACL rupture and concussion.

Practical implications

- Introduction of a foot blister prevention advice sheet and pack did not result in fewer foot blisters at a six-day netball tournament. Greater attention needs to be focussed on the implementation process.
- Prevention of ankle sprains, ACL ruptures and concussion must remain a priority in pre-elite netball athletes.
- We must heed the advice of the IOC and stop our youth athletes from competing or training when injured or ill.

Acknowledgements

The authors would like to thank the team physiotherapists attending the 2019 ANNC for collection of the medical attention injury data. We would also like to thank the athletes who attended these championships for completing the questionnaires and we thank their coaches and managers as well as Netball Australia for supporting this research. We would like to acknowledge and thank the Australian Institute of Sport Medical Department for the assistance with the design of the foot blister pack provided to athletes.

We would also like to thank Beiersdorf Australia for supplying the blister pads, plasters and non-stick dressings used to make up the blister packs distributed to all 192 athletes. Beiersdorf Australia did not provide input, review or approval over this manuscripts design, analysis, interpretation, or reporting. Additionally, we would like to thank the 'blister pack team' for making the 192 packs.

References

- 1. SPORTAUS. AUSPLAY netball state of play report. 2019. Accessed January 23, 2020. https://www.clearinghouseforsport.gov.au/research/smi/ausplay/results/sport.
- 2. International Netball Federation. Current world rankings. 2019. Accessed January 23, 2020. https://netball.sport/events-and-results/current-world-rankings.
- 3. Smyth EA, Piromalli L, Antcliff A, et al. A prospective study of health problems at the 2018 17/U and 19/U Australian National Netball Championships with comparison of surveillance methodology. *J Sci Med Sport*. 2020;23(3):215-221. doi: 10.1016/j.jsams.2019.10.004

- 4. Hume PA, Steele JR. A preliminary investigation of injury prevention strategies in netball: are players heeding the advice? *J Sci Med Sport*. 2000;3(4):406-413. doi: 10.1016/s1440-2440(00)80007-9.
- 5. Langeveld E, Coetzee FF, Holtzhausen LJ. Epidemiology of injuries in elite South African netball players. *S Afr J Res Sport Phys Educ Recreat*. 2012;34(2):83-93. Accessed July 30, 2018. <u>https://www.ajol.info/index.php/sajrs/article/view/84547</u>.
- 6. Pickering Rodriguez EC, Watsford ML, Bower RG, et al. The relationship between lower body stiffness and injury incidence in female netballers. *Sports Biomech*. 2017;16(3):361-373. doi: 10.1080/14763141.2017.1319970
- 7. Hopper D, Elliott B. Lower limb and back injury patterns of elite netball players. *Sports Med.* 1993;16(2):148-162. doi: 10.2165/00007256-199316020-00006
- 8. de Villiers JE, Venter RE. Barefoot training improved ankle stability and agility in netball players. *Int J Sports Sci Coach*. 2014;9(3):485-495. doi: 10.1260/1747-9541.9.3.485
- 9. Zulkarnain J, Khairullina K. The pilot study on Down to Earth (D2E) injury prevention program among varsity netball players. *Gazz Med Ital Arch Sci Med.* 2019;178(4):188-194. doi: 10.23736/S0393-3660.18.03766-X
- 10. Hopper D. A survey of netball injuries and conditions related to these injuries. *Aust J Physiother*. 1986;32(4):231-239. doi: 10.1016/s0004-9514(14)60656-5
- 11. Downs C, Snodgrass SJ, Weerasekara I, et al. Injuries in netball: a systematic review. *Sports Med Open.* 2021;7(1):3. doi: 10.1186/s40798-020-00290-7
- 12. Knapik JJ, Hamlet MP, Thompson KJ, et al. Influence of boot-sock systems on frequency and severity of foot blisters. *Mil Med.* 1996;161(10):594-598. doi: 10.1093/milmed/161.10.594
- 13. Lipman GS, Sharp L, Christensen M, et al. Paper tape prevents foot blisters: a randomized prevention trial assessing paper tape in endurance distances II (Pre-TAPED II). *Clin J Sport Med.* 2016;26(5):362-368. doi: 10.1097/JSM.00000000000319.
- 14. Baussan E, Bueno MA, Rossi RM, et al. Analysis of current running sock structures with regard to blister prevention. *Text Res J.* 2012;83(8):836-848. doi: 10.1177/0040517512461698
- 15. Knapik JJ, Reynolds KL, Duplantis KL, et al. Friction blisters. Pathophysiology, prevention and treatment. *Sports Med.* 1995;20(3):136-147. doi: 10.2165/00007256-199520030-00002
- Bahr R, Clarsen B, Derman W, et al. International Olympic Committee consensus statement: methods for recording and reporting of epidemiological data on injury and illness in sport 2020 (including STROBE Extension for Sport Injury and Illness Surveillance (STROBE-SIIS)). Br J Sports Med. 2020;54(7):372-389. doi: 10.1136/bjsports-2019-101969
- 17. Timpka T, Jacobsson J, Bickenbach J, et al. What is a sports injury? *Sports Med.* 2014;44(4):423-428. doi: 10.1007/s40279-014-0143-4
- 18. Clarsen B, Rønsen O, Myklebust G, et al. The Oslo Sports Trauma Research Center questionnaire on health problems: a new approach to prospective monitoring of illness and injury in elite athletes. *Br J Sports Med.* 2014;48(9):754-760. doi: 10.1136/bjsports-2012-092087
- 19. Clarsen B, Bahr R, Myklebust G, et al. Improved reporting of overuse injuries and health problems in sport: an update of the Oslo Sport Trauma Research Center questionnaires. *Br J Sports Med.* 2020:bjsports-2019-101337. doi: 10.1136/bjsports-2019-101337
- 20. Rae K, Orchard J. The Orchard Sports Injury Classification System (OSICS) version 10. *Clin J Sport Med.* 2007;17(3):201-204. doi: 10.1097/JSM.0b013e318059b536
- 21. Stovitz SD, Shrier I. Injury rates in team sport events: tackling challenges in assessing exposure time. *Br J Sports Med.* 2012;46(14):960-963. doi: 10.1136/bjsports-2011-090693
- 22. Gordis L. *Epidemiology*. 5th ed. Elsevier Saunders; 2014.
- 23. Bergeron MF, Mountjoy M, Armstrong N, et al. International Olympic Committee consensus statement on youth athletic development. *Br J Sports Med.* 2015;49(13):843-851. doi: 10.1136/bjsports-2015-094962
- 24. Finch C, Costa AD, Stevenson M, et al. Sports injury experiences from the Western Australian sports injury cohort study. *Aust N Z J Public Health*. 2002;26(5):462-467. doi: 10.1111/j.1467-842X.2002.tb00348.x

- 25. Kirkwood G, Hughes TC, Pollock AM. Results on sports-related injuries in children from NHS emergency care dataset Oxfordshire pilot: an ecological study. *J R Soc Med.* 2019;112(3):109-118. doi: 10.1177/0141076818808430
- 26. Finch CF, Clapperton AJ, McCrory P. Increasing incidence of hospitalisation for sport-related concussion in Victoria, Australia. *Med J Aust.* 2013;198(8):427-430. doi: 10.5694/mja12.11217
- 27. Saunders N, Otago L. Elite netball injury surveillance: implications for injury prevention. *J Sci Med Sport*. 2009;12:S63. doi: 10.1016/j.jsams.2008.12.148
- 28. Smith MMF, Mendis MD, Parker A, et al. Injury surveillance of an Australian community netball club. *Phys Ther Sport*. 2020;44:41-46. doi: 10.1016/j.ptsp.2020.04.004
- 29. Harris BS, Watson JC. Developmental considerations in youth athlete burnout: a model for youth sport participants. *J Clin Sport Psychol.* 2014;8(1):1-18. doi: 10.1123/jcsp.2014-0009
- 30. DiFiori JP, Benjamin HJ, Brenner J, et al. Overuse injuries and burnout in youth sports: a position statement from the American Medical Society for Sports Medicine. *Clin J Sport Med.* 2014;24(1):3-20. doi: 10.1097/jsm.0000000000000000
- 31. Drew MK, Finch CF. The relationship between training load and injury, illness and soreness: a systematic and literature review. *Sports Med.* 2016;46(6):861-883. doi: 10.1007/s40279-015-0459-8
- 32. Juliff LE, Halson SL, Bonetti DL, et al. Influence of contrast shower and water immersion on recovery in elite netballers. *J Strength Cond Res.* 2014;28(8):2353-2358. doi: 10.1519/jsc.00000000000417
- Juliff LE, Halson SL, Hebert JJ, et al. Longer sleep durations are positively associated with finishing place during a national multiday netball competition. *J Strength Cond Res*. 2018;32(1):189-194. doi: 10.1519/jsc.000000000001793

Tables

	Body area										
				Lower	Head &		Wrist, hand		Upper		
	Ankle	Foot	Trunk †	leg	neck	Knee	& fingers	Hip & groin	limb ‡	Thigh	Total
Total (n=)	24	14	16	8	8	9	5	2	4	5	95
<u>Mechanism</u>											
Accident/other	0	0	1	0	0	0	1	0	0	0	2
Catching	0	0	0	0	0	0	4	0	0	0	4
Change of direction	2	0	0	0	0	0	0	0	0	0	2
Collision	1	0	0	2	5	1	0	0	2	4	15
Contested landing	4	0	1	0	1	2	0	0	0	0	8
Jumping	1	0	0	1	0	1	0	0	0	0	3
Land on another	7	0	0	0	0	0	0	0	0	0	7
Non-contested landing	0	0	0	0	0	4	0	0	0	0	2
No specific incident	5	13	10	5	0	0	0	2	1	1	37
Missing	3	0	3	0	2	1	0	0	1	0	10
Classification											
New injury	23	14	14	7	8	8	5	2	4	5	90
Recurrent injury	1	0	2	1	0	1	0	0	0	0	5
Onset											
Trauma	18	0	1	2	7	7	5	0	2	4	46
Overuse	5	9	12	5	0	2	0	2	2	1	38
Other	1	5	3	1	1	0	0	0	0	0	11

Table 1: Mechanism of injury, classification and onset for each body area

† Thoracic and Lumbar spine; ‡ Shoulder, Elbow & Forearm

Chapter 7 Discussion and conclusion

Thesis summary

The primary objective of this sequence of doctoral studies was to apply a holistic strategy to prevent injuries in pre-elite athletes, using netball as an exemplar sport. To achieve this, we collaborated with Netball Australia. There are many stakeholders that influence outcomes for pre-elite athletes, and there are many injury prevention components to consider. Most organisations will already have established injury prevention strategies; therefore, it is important for these strategies to be considered when planning for future injury prevention interventions. Netball Australia had already started collecting injury data at the 17/U & 19/U ANNC, developed the KNEE program and established an ankle taping policy. Therefore, any interventions must take into consideration the established work within any organisation.

The major contribution of this work was to demonstrate how the process of injury prevention can only be achieved when researchers, practitioners, coaches, athletes and administrative staff work collaboratively. This has been demonstrated in other sports such as Norwegian female handball,⁷³ New Zealand rugby union¹³⁶ and Swiss amateur football.¹³⁷ We have worked collaboratively with Netball Australia to establish their pre-elite injury profile and progress the development and implementation of injury prevention interventions to reduce injury incidence in the pre-elite population. The lessons gained in this process have influenced the research studies included in this thesis and the thesis discussion.

Firstly, a multidisciplinary approach was used to achieve good quality injury surveillance, aetiology and mechanisms of injury, as per TRIPP Stages 1 and 2. Across a six-day tournament, 103 injuries were sustained by 80 athletes. The most frequent injuries were lateral ankle sprain, foot blisters and lumbar pain. Following the injury surveillance, a thorough systematic review was conducted as part of TRIPP Stage 3 to inform potential injury prevention intervention strategies. Eleven articles met the inclusion criteria of a pre-elite population competing in an Olympic or professional sport investigating interventions for any injury at any body site. There is limited evidence from level 2 and 3 studies suggesting exercise and psychology interventions may prevent injury in pre-elite athletes. There were no studies meeting the criteria of low risk of bias.

These first two studies then directed the approach of the remaining studies. Regular prophylactic ankle taping was in place for the athletes competing at the 17/U & 19/U ANNC but there was limited recent evidence supporting this intervention. Discussion with Netball Australia and relevant ethics committees determined that removing a preventative action to assess efficacy was not ethical, noting it would be the best scientific approach. Therefore, ankle taping was scientifically evaluated under 'ideal conditions' according to TRIPP Stage 4 to demonstrate that proprioception improves immediately after application; And that it is maintained throughout a netball session regardless of whether it is applied by a Sport and Exercise Physiotherapist (SEP) or self-applied.

Concept mapping (CM) was used to fulfill TRIPP Stage 5 to improve implementation of Netball Australia's KNEE program. The coaches, physiotherapists, and strength and conditioning coaches attending the 2019 17/U & 19/U ANNC were consulted to gain a better understanding of the challenges to implementing the KNEE program. This will help guide implementation strategies in the future.

Finally, TRIPP Stage 6 was applied by assessing the effectiveness of a foot blister prevention strategy at the 2019 17/U & 19/U ANNC. Additionally, medical attention and self-report injury surveillance conducted at the 2018 17/U & 19/U ANNC were repeated at the 2019 17/U & 19/U ANNC.

We attempted to apply a generalisable approach to sport injury prevention. However, when working with established sports such as netball, who have established policy and procedure to prevent injuries, we had to be flexible with our study design. Working in collaboration with our end user, in this case Netball Australia, we prioritised the research questions they needed to be answered rather than what the research community thought should be answered. A strength of this thesis is the diversity of research methodology used to add value for the target population of netball. This is a parallel thesis rather than a series thesis to achieve a pragmatic approach to research.

Focus of thesis

The primary objective of this PhD was to develop and implement a system that will prevent injuries in pre-elite athletes, using netball as an exemplar sport. To achieve this, we collaborated with Netball Australia and attempted to apply the TRIPP framework with their pre-elite athletes. We established injury surveillance methods, reviewed current injury prevention strategies in the pre-elite setting, created supporting evidence and determined barriers for implementation of current injury prevention strategies and implemented a foot blister prevention strategy. This work will inform our strategies in the future for injury prevention in pre-elite athletic populations.

Summary of chapters and their key findings

Chapter 1

Chapter 1 presented background information on athletic development, injuries in pre-elite athletes, the sport of netball, the importance of injury prevention, types of injury prevention and models of injury prevention. One model of injury prevention is the six-stage TRIPP framework, which has been promoted as an approach that leads to real-world injury prevention gains. It has provided the foundation for the current investigation into injury prevention in pre-elite athletes. Chapter 1 outlines each of the TRIPP stages and any evidence of these stages being applied in the sport of netball.

Chapter 2

Chapter 2 reports the findings of a prospective study investigating health problems at the 2018 17/U & 19/U Australian National Netball Championships with comparison of two surveillance methodologies. One hundred and ninety-two athletes were observed, and injuries were recorded using medical attention and self-report data collection methods. There were 103 medical attention injuries sustained by 80 athletes, resulting in an incidence rate of 89.4 injuries per 1000 player hours. The most frequently recorded medical attention injury diagnoses across both age groups were lateral ankle ligament sprains (n=14, 13.6%), foot blisters (n=11, 10.7%) and lumbar pain (n=10, 9.7%). Ankle sprains (n=4), anterior cruciate ligament (ACL) ruptures (n=3) and concussion (n=3) were recorded as the highest sports

incapacity injuries. Foot blisters (n=2) and foot sprain (n=2) also accounted for multiple incidences of a sports incapacity injury diagnosis. Of the 103 medical attention injuries, the most frequent 'mechanism of injury' was 'unknown/missing' (n=32, 31%). The second most frequent was collision (n=17, 17%). The self-report data collection revealed that 46 (27.2%) athletes arrived at the tournament with a health problem and there was a significant increase in health problems during the tournament as there were 57 (39.3%) self-reported health problems at the conclusion of the tournament (RR 1.44 95%CI 1.05-1.99, p=0.030). Finally, combining both the medical attention and self-report health problem data collection methods identified more health problems than the use of one method alone.

Chapter 3

Chapter 3 reported the outcome of a systematic review on injury prevention strategies specific to preelite athletes competing in Olympic and professional sports. Studies were considered if their included data was specific to pre-elite athletes (determined by the T3/4 levels of the FTEM model), featured an injury prevention intervention, provided sufficient data related to injury so the effect could be analysed, and involved a randomised or non-randomised controlled trial or prospective cohort. Eleven studies satisfied the inclusion criteria, but none of these demonstrated low risk of bias. Four different types of intervention were identified: exercise, psychological, equipment and nutrition. There is limited evidence from level 2 and 3 studies suggesting exercise and psychology interventions may prevent injury in preelite athletes. However, the exercise interventions had high risk of bias, low levels of intervention reporting and minimal evidence of implementation planning. There continues to be a need for quality research designs confirming the clinical impact of existing injury prevention interventions for pre-elite athletes.

Chapter 4

A prophylactic ankle taping policy was in place by Netball Australia, requiring all athletes to tape their ankles for training and matches but there was limited recent evidence for this intervention. In Chapter 4, ankle taping was scientifically evaluated under 'ideal conditions' according to TRIPP Stage 4. An Active Movement Extent Discrimination Apparatus (AMEDA) was used to assess proprioception before and after tape was applied and after a netball session before and after tape was removed. Two different tape conditions were assessed, applied by a Sport and Exercise Physiotherapist and self-applied tape. This study found that proprioception improves immediately after application and is maintained throughout a netball session and that there is no difference between tape conditions.¹³⁸ Based on this evidence it appears that the taping procedure increases proprioception and may address the hypothesised risk for ankle injury. Previous basketball research has demonstrated ankle tape prevents injury.^{139,140} A barrier to performing research to assess the efficacy of ankle tape to prevent ankle sprain in pre-elite netball athletes is the need for a control group to not wear tape. Pre-elite athletes would not be willing to risk spraining their ankles by not wearing tape. Based on previous research,^{139,140} and our recent findings, Netball Australia should continue to encourage its pre-elite netball athletes to tape their ankles.

Chapter 5

TRIPP Stage 5 was applied to improve implementation of Netball Australia's KNEE program. Concept mapping was used to consult the State U/17 & U/19 coaches, strength and conditioning coaches, and physiotherapists as to what they identified as challenges to using the KNEE program. Athlete engagement, supervision/correction of technique, time constraints, athlete technique, education and support staff resourcing were deemed to be the most important challenges to address. Supervision/correction of technique, time constraints, athlete technique, support staff resourcing, and athlete engagement were identified as the most difficult challenges to overcome. This information will help guide future implementation planning.

Chapter 6

Injury surveillance was repeated at the 2019 17/U and 19/U ANNC and a foot blister prevention intervention was implemented. One hundred and ninety-two athletes were observed, and injuries were recorded using medical attention and self-report data collection methods. A foot blister prevention pack and advice sheet were provided to each athlete six weeks prior to the 2019 ANNC. There were 95 medical attention injuries reported across 73 athletes over a six-day period at the 2019 ANNC, at an incidence rate of 82.5 injuries/1000 player competition hours. Ankle sprains (n=16, 16.8%), lumbar

pain (n=12, 12.6%) and foot blisters (n=8, 8.4%) were the most frequently reported injuries. Thirty per cent of athletes started the tournament with a self-reported injury/illness. There were eight reported foot blisters in 2019 compared to 11 in 2018, resulting in an IRR of 1.38 (95% CI: 0.50–3.94, p=0.50). No foot blisters caused sports incapacity in 2019, compared to two cases of sports incapacity foot blisters in 2018.

Ankle sprains (n=14) were also the most frequent injury at the 2018 ANNC,¹¹⁴ indicating a need for more targeted injury prevention strategies. There were three fewer foot blisters in 2019 compared to 2018, which was not statistically significant. Foot blister discomfort can impact performance as blisters can be distracting and result in biomechanical changes.^{141,142} They can be easily prevented with correct footwear and foot hygiene, so this reduction is encouraging but it is not unreasonable to aim for eradication of foot blisters at future netball tournaments.

Discussion of the key thematic findings of the thesis

The implementation of injury prevention strategies in a pre-elite population is challenging

Understanding the context of the injury problem supersedes all other injury prevention requirements.^{143,144} Injury prevention in high-performance sport therefore needs to be led by sporting organisations rather than researchers trying to 'translate' their research findings for a sporting organisation with little regard for its context. The benefit of collaborating with researchers early has been recognised in a report released by the Australian Government, outlining that businesses collaborating with research organisations are three times more likely to experience productivity growth.¹⁴⁵ It is also recognised that research is more effective if it is driven by industry; hence, the Australian Government has developed a service to empower businesses to engage with research organisations.^{146,147} Injury surveillance is widely conducted in professional youth football academies but use of injury data by coaching staff appears to be poor.¹⁴⁸ Coaching staff engagement is essential to achieve injury prevention in sport.^{149,150} If injury prevention was prioritised by sporting organisations we may see improved coach engagement and, therefore, more meaningful research resulting in effective implementation of injury prevention strategies. Perhaps the only way to achieve this, as suggested by

Ekstrand,¹⁵¹ is to lift the level of importance of health professionals within a sporting organisation hierarchy – for example, to include a health professional on the board. Baugh et al.¹⁵² have demonstrated the impact of organisational structure on injury rates, showing that National Collegiate Athletic Association (NCAA) schools with sports medicine departments financed by and reporting to athletic departments rather than medical institutions had 31% higher injury incidences. If health professionals could be more autonomous within a sporting organisation, perhaps injury prevention strategies would be prioritised and organisationally endorsed, resulting in lower injury rates.

Injury prevention processes such as injury surveillance and implementation methods must be tailored to the pre-elite athlete. We cannot assume that injury prevention strategies found to be effective in the community or at elite levels will also be effective for pre-elite athlete groups. Further, pre-elite athletes, when compared to community-based athletes, are commonly more invested in their development and self-management but at the same time often have more competing demands, such as school and multiple teams/sports demands, than elite athletes.^{14,15} Current pre-elite athletes are predominantly 'Generation Z' athletes (i.e., born between 1996 and 2012) with differing values and priorities. Several surveys have found that Generation Z are heavily reliant on technology, self-educated and value treating people with respect, ethical behaviour, fair compensation and transparent communication.¹⁵³ Therefore injury prevention strategies need to be communicated well, participants need to be treated fairly and technology must be embraced. Because Generation Z athletes have different characteristics and values to other generations, injury prevention for pre-elite athletes warrants prioritisation by sporting organisations for specific strategies to be developed for this cohort.

Implementation of an injury prevention strategy requires an organisation to move from its present state to a desired future state. This is also known as planned organisational change,¹⁵⁴ for which many change management models exist. Yet these models have not been incorporated into any existing injury prevention frameworks.¹⁵⁵ Stouten et al.,¹⁵⁶ after reviewing the literature, proposed ten change management principles. The first step is to 'get facts regarding the nature of the problem', which could be interpreted as TRIPP Stage 1, and the second step is to 'assess and address the organisation's readiness for change'. A critical part of this step is ensuring senior leadership is ready and capable of

guiding and implementing change. Without the endorsement and guidance of senior leadership, implementation of injury prevention strategies will fail.¹⁵⁷ The third step consists of implementing an evidence-based change intervention and the fourth step focusses on developing effective change leadership throughout the organisation. The fifth step uses multiple communication channels to help change recipients (i.e., athletes and coaches) understand the reason for change and the sixth step recommends working with social networks to tap their influence. The seventh step encourages the use of tools to support implementation such as goal setting and optimising change recipient participation. The eighth step promotes small-scale change interventions to adjust for local requirements (i.e., available resources) and the ninth step assesses change progress over time. The final principle proposed by Stouten et al.¹⁵⁶ is 'institutionalise the change to sustain its effectiveness'. This means integrating the change into the larger systems of the organisation, including its culture and management systems. A key requirement for this to occur is having a committed board that can help make change sustainable by providing critical resources as needed and by recruiting senior management who will ensure the continuity of change.¹⁵⁸ Without incorporating change management principles into injury prevention strategy implementation, we will not see a reduction in injury rates.

Australia's federated structure also poses a challenge for implementing injury prevention strategies for pre-elite athletes.¹⁵⁹ For example, Netball Australia has eight state and territory member organisations. These state and territory member organisations provide support and services for all pre-elite athletes. Therefore, to implement injury prevention strategies for Australian pre-elite netball athletes, change management needs to occur across nine different organisations. This requires not only strong leadership but also excellent communication skills, coordination and commitment by all organisations. The federated structure creates layers of administration and management resulting in duplication of processes and inefficiencies.¹⁶⁰ A sports injury prevention framework, incorporating change management principles, developed in consultation and agreed upon by all member organisations could help address the challenges created by a federated structure.

Prevention frameworks are sequential and do not consider measures in place already. This should be considered when attempting to apply the TRIPP framework in a pre-elite population with established prevention programs in place.

The TRIPP framework focuses on translating research into prevention practice in a sequential manner but it may be necessary to rethink this approach, particularly when working in high-performance sport. Although all steps of the TRIPP framework are essential for injury prevention, it is not always practical and/or feasible to use this sequential format in the high-performance sport setting. This is evident throughout this thesis. The first task when researchers and high-performance sporting organisations collaborate is to understand the context,^{143,161} including what injury prevention strategies are currently in place in the sport, how the injury prevention strategies are assessed, how well the strategies are being adopted and the level of evidence supporting their use. With this additional information, researchers and high-performance staff can collaborate in a continual improvement process. The TRIPP steps can be used as a guide for planning meaningful injury prevention projects to improve upon what exists or remove existing ineffective strategies rather than needlessly initiating the TRIPP framework in a sequential manner. Importantly, any researcher attempting to undertake the TRIPP process without first understanding what measures, policy and processes are already in place is unlikely to be effective when working with sporting organisations.

Given the ever-changing nature of high-performance sport, a contingency approach often needs to be adopted where different situations require different methods of management. An example of this is how the TRIPP framework was used to support the collaboration with Netball Australia. Firstly, efficacy trials were not completed, and injury surveillance had not been implemented before introducing the KNEE program to the netball community. This limits Netball Australia's ability to measure effectiveness of the KNEE program, which can then negatively impact ongoing adoption and implementation of the program as well as jeopardising ongoing funding for the program. In response, injury surveillance was conducted¹¹⁴ (i.e., TRIPP Stages 1 and 2), identifying ankle and knee injuries as the most frequent. Netball Australia's staff, using research evidence and clinical experience, had already developed a prevention program for knee and ankle injuries, so the next logical progression was to apply TRIPP Stage 5 using concept mapping to investigate the challenges to implementing the KNEE

program.¹⁶² The next stage would be to review the KNEE program by revisiting TRIPP Stage 3 to address the challenges to implementation and then reapply TRIPP Stage 5, using Intervention Mapping (Step 5) to plan implementation. Finally, the injury surveillance (TRIPP Stage 6) would be repeated to test effectiveness in a real-world setting. Our investigation demonstrates that the TRIPP framework is useful but not in its intended sequential manner when working in a high-performance environment with existing programs.

An additional challenge for using this framework is that it is time-consuming to complete all six stages, possibly taking considerable time (multiple seasons) to complete. TRIPP Stages 1 and 2 involved a three-month process to prepare for data collection, collect data and analyse data.¹¹⁴ TRIPP Stage 3 was a two-month process of searching, screening and extracting data from the literature¹⁰⁴ and this was only Step 1 of a six-step intervention development process.⁶⁷ TRIPP Stage 4 was a two-month process for planning, executing and analysing the results of a randomised controlled trial.¹³⁸ In this stage we sought to understand the mechanism by which ankle tape prevents ankle sprain. Previous studies have shown efficacy of taping/bracing in other populations, but this has not been assessed specifically in pre-elite netball. TRIPP Stage 5 was a two-month process for planning, administering and analysing the concept mapping data to determine the barriers to implementing the KNEE program.¹⁶² This only partly fulfilled Stage 5; using IM (Step 5) to plan implementation would have involved an additional time cost. Finally, TRIPP Stage 6 incorporated an additional three-month process to prepare the intervention, prepare for data collection, collect data and analyse data. In total, completing all of the TRIPP stages was a 12month process of planning, enacting and analysing in a full-time capacity. In reality, it was an even longer process due to the negotiations required with key stakeholders for when each stage could be conducted and waiting for those opportunities to arise. Currently, within Australian National Sporting Organisations (NSOs), including Netball Australia, there are no dedicated personnel employed in a fulltime capacity to oversee and manage the processes outlined above. Further, high-performance sport is known for its high staff turnover¹⁶³ and it is therefore difficult to maintain the momentum and direction of injury prevention strategies with lengthy timeframes. Additionally, delays and changes in staff can also result in initial data collection not reflecting current injury profiles. This reinforces the need for

sporting organisation executives to set a culture and legacy of injury prevention. This could be achieved by embedding injury prevention strategies into policy and strategic plans and employing researchers with a multi-disciplinary and clinical focus within high-performance sport for sustained injury prevention success.¹⁶⁴

TRIPP Stage 1 (Injury surveillance)

Reflections and learnings from applying TRIPP Stage 1 in the field

TRIPP Step 1 requires accurate injury data collection to allow comparison between injury surveillance programmes. This requires consistent injury definition and recording methodology be used.⁴⁶ The Orchard Sports Injury Classification System (OSICS)¹⁶⁵ was developed in 1992 and has been consistently updated, with the most recent version (13.5) released in July 2021. It is a commonly used system that has been shown to result in high levels of agreement among data recorders.¹⁶⁶ However, more than a reliable classification system is required for accurate data collection. Wik et al.¹⁶⁷ identified that research-invested clinicians reported 8.8 times greater injury incidence of non-time-loss injuries than their colleagues not involved in research. At the commencement of this thesis, netball had implemented a paper-based medical attention surveillance system, but it was unclear whether these data had been analysed and reported. All clinicians entering the data were primarily engaged to provide physiotherapy services to their state teams and, therefore, were likely not engaged in the surveillance system. Extrapolating from Wik et al., it is likely that injuries were under-reported prior to the study reported in Chapter 2 of this thesis.¹¹⁴

Another area for improvement of TRIPP Stage 1 is expanding 'injury surveillance' to 'injury surveillance and its context', as presented by Bolling et al.¹⁴³ They argue that because of the complexity of sports injury, context needs to be determined from the outset; otherwise, context-free preventative solutions will be developed. Environmental, sociocultural and individual contextual aspects such as beliefs, attitudes, culture, social structures, governance, policy and budget should be identified when describing the sports injury problem. Verhagen et al.¹⁴⁴ also stress the importance of understanding context when describing Step 1 of their Knowledge Transfer Scheme (KTS). The first KTS step, labelled 'Problem statement', includes not only a description of a health problem but also a description

of the health problem's impact on the community and its specific behavioural and environmental causes. If additional contextual information is sought at the time of injury surveillance, injury prevention initiatives could be more effective.

Most academic sport injury prevention models^{43,45,168} assume there are no existing prevention strategies in the target population, but this would be rare in high-performance sport. O'Brien et al.¹⁶⁹ proposed the Team-sport Injury Prevention (TIP) cycle, stating that sport practitioners require a model that is directly applicable to their team's specific context. Their first phase of 'evaluation' includes not only an assessment of current injury prevalence but also injury prevention strategies in situ. This is part of understanding context, as recommended by Bolling et al.¹⁴³ and Verhagen et al.¹⁴⁴ This was undertaken during the initial strategising scope of studies with Netball Australia. We learnt that injury surveillance had been attempted at the 2017 17/U and 19/U ANNC, an ankle taping policy was in place and the KNEE program had been implemented.

Netball specific injury surveillance (TRIPP Stage 1)

Injury rates at the 2018 17/U & 19/U ANNC were high, with an incidence rate of 89.4/1000 player hours.¹¹⁴ We found that across a six-day tournament ankle sprains (n=14), foot blisters (n=11) and lumbar pain (n=10) were the most frequently recorded medical attention injuries.¹¹⁴ Ankle sprain (n=4), ACL rupture (n=3) and concussion (n=3) were the highest sports incapacity injuries, followed by foot blisters (n=2) and foot sprain (n=2).¹¹⁴ Ankle sprains and knee injuries have consistently had the highest incidence rate in netball over the past 25 years.^{56-58,131} However, of the 46 studies identified in a recent netball injury surveillance systematic review,⁵⁴ only three studies^{114,115,128} have specifically reported concussion rates and three studies^{55,58,114} reported foot blister rates, which does not provide us with an accurate understanding of netball injuries.

Concussion incidence in netball appears to have increased over the past 20 years. In 1998, Finch et al.¹¹⁵ reported a frequency of just 1.6% (n=4) over two consecutive five-month seasons. In 2014, there was one concussion recorded from 154 netball injuries presenting to an emergency department within a two-year period.¹²⁸ Finch et al.¹²⁹ reported an 8% annual increase in concussion sustained by netball athletes from 2002–03 to 2010–11. One possible explanation of this increase in concussions is potentially

multifactorial in nature. Historically, concussions have not been recognised and formally diagnosed. Another potential explanation is that there are now more collisions occurring in netball,^{130,131} resulting in more concussions.

Foot blisters have not been frequently reported in the netball literature. However, in 1983 Hopper⁵⁵ recorded 45 foot blisters across a 14-week season for 3108 netball athletes, translating into 1.4% of athletes sustaining a foot blister. A similar frequency of 2% (n=2) was reported from a three-day netball tournament in 1995.⁵⁸ Foot blisters can severely impair function and concentration,^{141,142} and future netball injury surveillance initiatives should ensure foot blisters are recorded to obtain an accurate netball injury profile. The type of playing surface, sock, shoe and training load should also be recorded to enhance our aetiological knowledge.

Two injury surveillance methods were implemented and compared at the 2018 17/U & 19/U ANNC. The HPQ identified 26 injuries/illnesses that were not recorded in the Athlete Management System (AMS), 12 of which were overuse injuries and ten illnesses. Conversely, the AMS recorded 33 injuries not identified in HPQ. This demonstrates that using multiple injury surveillance tools results in more accurate data collection. It also confirms the usefulness of the HPQ to capture overuse injuries and illnesses and is not reliant on a health professional to capture data. It is also possible that self-report data can capture injuries when athletes fail to report their injury to a health professional due to pressure from coaches, parents, and peers.²⁹ The HPQ can be used for injury surveillance in many development athlete settings where surveillance may have been previously considered not possible.

The HPQ also provides us with additional information about extent of symptoms, impact on training/competition and affect on performance.¹⁰² It's important to recognise that athlete responses may not be a true indication of injury, rather a reflection of the athlete's coping skills. It is possible for an athlete to use injury as an excuse for poor performance or as a coping mechanism to manage the stress of high-performance sport.^{170,171} To gain an accurate understanding of self-report injury surveillance, perhaps a questionnaire such as the Athletic Coping Skills Inventory¹⁷¹ can be employed and interpreted by a psychologist.

TRIPP Stage 2 (Establish aetiology and mechanisms of injury)

Reflections and learnings from applying TRIPP Stage 2 in the field

Our findings revealed the importance of recording aetiology and mechanism of injury to help guide future injury prevention interventions. Additional injury/athlete characteristics such as previous injury, sex, age, height, weight, dominant limb, level of competition, tournament or in-season, match or training, time of year, playing position, position on court, time of injury, menstrual cycle information etc. can provide valuable information when developing injury prevention strategies. A challenge for collecting these data is that it is more time consuming for the clinician to record and often the required information is unknown at the time of data entry.¹⁷² Therefore, it is not uncommon to have a high level of missing data or non-descriptive data (i.e., 'other'), as demonstrated in our study¹¹⁴ and other studies.^{55,58} If clinicians – or, better, researcher/clinician partnerships – were more invested in research outcomes, more accurate aetiological and mechanism of injury data might be obtained. Similarly, these results could be enhanced if epidemiology were included in undergraduate and postgraduate clinician education and recording of this information was mandated in policy, job descriptions or funding agreements.

The literature suggests there is an association between hormonal fluctuations during a menstrual cycle and ACL injury.¹⁷³ There is also evidence for Relative Energy Deficiency in Sport (RED-S) causing impairment of menstrual function, bone health and immunity.¹⁷⁴ To improve our understanding of these physiological functions, there is a need to continue to collect these data.¹⁷⁵ However, despite the evidence for collecting menstrual cycle information, team physiotherapists did not record this information for over half of the injuries sustained at the 2018 17/U & 19/U ANNC (Chapter 2). This was a missed opportunity to identify and manage athletes who may be experiencing RED-S and to gain a better understanding of the degree of association between hormone levels and ligament injury.

Netball specific mechanisms of injury and aetiology (TRIPP Stage 2)

We found that new injuries occurred nine times more frequently than recurrent injuries.¹¹⁴ Langeveld et al. also reported new injuries to be more frequent.⁵⁷ This finding suggests that either tertiary prevention has been effective for preventing injuries from recurring in this pre-elite population or there

has been previous under-reporting of injuries in this pre-elite population. Regardless, it indicates a need for a greater focus on primary prevention to prevent new injuries from occurring.

Collisions and landings have consistently been the most frequent mechanisms of injury in netball over the past 30 years. The second most frequent mechanism of injury after 'unknown/missing' at the 2018 17/U & 19/U ANNC was 'collision'.¹¹⁴ Hume and Steele⁵⁸ also found collision to be the second most frequent mechanism of injury behind 'other'. This is problematic and improving collection of mechanism of injury data needs to become a priority in future surveillance studies. This can be resolved by data collectors being involved in the research team and by reviewing video footage for accurate injury analysis. Collision was the third-highest mechanism of injury at the 1988 Australian netball championships and 'incorrect landing' was the most frequent,⁵⁹ whereas 'contested landing' was the fifth most frequent mechanism of injury at the 2018 17/U & 19/U ANNC¹¹⁴ and fourth most frequent mechanism of injury at the 1995 New South Wales State Netball Championships.⁵⁸ 'Incorrect landing' was less frequent than 'collision' in 2018 compared to 1988, suggesting coaching of landing technique may have improved over the past 30 years and perhaps that more attention should be given to reducing the number of collisions.

TRIPP Stage 3 (Develop preventive measures)

Reflections and learnings from applying TRIPP Stage 3 in the field

Developing an injury prevention strategy relies on a combination of scientific evidence, practitioner expertise and end-user values. Donaldson et al.⁶⁷ proposed the six-step IDP to fulfil the requirements of Stage 3 of the TRIPP⁴³ and van Mechelen's 'sequence of prevention' model.⁴² There is limited evidence of organisations applying this process for injury prevention development.^{67,176} It was first applied within community Australian football to develop FootyFirst, a lower-limb injury prevention exercise training program.⁶⁷ Subsequently the IDP has been used to plan for intervention development in youth handball but has not yet been implemented.¹⁷⁶ We completed Step 1 of the six-step process (i.e., use the research evidence and clinical experience) by undertaking a systematic review¹⁰⁴ but could not complete the six-step process due to time constraints. The IDP is a systematic and pragmatic guide but requires a significant time commitment, as demonstrated by the 14 months required to develop FootyFirst.⁶⁷ The

'Intervene' stage of the TIP cycle¹⁶⁹ is less time-consuming as it consists of three steps, as follows: obtain administrative support,¹⁷⁷ engage key partners to design a strategy¹⁷⁸ and, finally, plan for implementation in a timely manner.¹⁶⁹ Perhaps the 'Intervene' process is a more realistic and practical fit than the IDP for developing an effective and timely intervention in high-performance sport.

Netball specific development of preventive measure (TRIPP Stage 3)

Several netball-specific exercise injury prevention interventions have been developed, such as 'Down to Earth',¹⁷⁹ the KNEE program,¹⁸⁰ 'Knees up'⁵⁸ and 'Netball Smart'.¹⁸¹ The 'Knees up' program, developed in 1995 by Sports Medicine Australia, provides advice on netball injury prevention and management.⁵⁸ The 'Netball Smart' program was developed in 2004 by New Zealand's Accident Compensation Corporation (ACC) and Netball New Zealand based on 'SportSmart', a 10-point action plan for sports injury prevention developed by the ACC.¹⁸² The 'Down to Earth' program was developed in 2006 by researchers at the School of Human Movement and Sport Sciences, University of Ballarat.¹⁷⁹ It was based on the best available research at the time, specifically adapted for netball from sports such as basketball, soccer and European handball. The KNEE program was developed in 2015 by Netball Australia using the principles of existing injury prevention programs.¹⁸⁰ There is no published evidence describing how these programs were developed but we must acknowledge that implementation science was still in its infancy within sports injury prevention when these programs were developed.

Step 1 of the IDP process undertaken in this thesis was completed by conducting a systematic review¹⁰⁴ of injury prevention strategies specifically trialled with pre-elite athletes. Limited evidence from level 2 and 3 studies was identified, suggesting exercise¹⁸³⁻¹⁸⁹ and psychological^{190,191} interventions may assist in preventing injury within the pre-elite athlete cohort. It is noteworthy that netball did not feature in any of the included studies. The other types of interventions included in the systematic review were equipment¹⁹² and nutrition¹⁹³ strategies. All published netball injury prevention strategies to date have focused on exercise interventions, but these have not been trialled in the pre-elite population. There would be value in assessing these interventions in subsequent pre-elite populations in addition to considering other types of intervention, such as psychological support or policy change as demonstrated within 13–14-year-old ice hockey leagues,¹⁹⁴ New Zealand Rugby¹³⁶ and baseball.¹⁹⁵

TRIPP Stage 4 ('Ideal conditions'/scientific evaluation)

Reflections and learnings from applying TRIPP Stage 4 in the field

Randomised controlled trials are the foundation of evidence-based practice.¹⁹⁶ Unfortunately, there are very few randomised controlled trials inclusive of high-performance athletes.¹⁹⁷ It is often difficult to access elite sporting populations¹⁶⁴ as the experimental-control requirements will often interfere with typical training and competitive preparations and the chance of not receiving the experimental intervention is at odds with the high-performance mind set of 'searching for the winning edge'.¹⁹⁷ Additionally, it is critical that researchers take into consideration the current safety/injury culture of the sport,^{43,198} which means athletes, coaches and support staff are not unduly concerned about athlete safety and welfare if participating in research. Most high-performance sports already have injury prevention strategies in place. Therefore, perhaps TRIPP Stage 4 needs to focus on developing evidence for a strategy, not just on proving efficacy because it is unethical to remove a prevention strategy that has already been implemented. Another consideration is the sample size in high-performance sport because of the small number of high-performance athletes when compared to the community. Nevertheless, despite the difficulty in obtaining significant results leading to generalisable inference, research in highperformance sport can result in improved decision making for the sport. Recommendations have been made for statistical techniques which can improve insights from suboptimal sample sizes.¹⁹⁹ Perhaps in high-performance sport, injury prevention strategies need to be developed based on context and existing evidence.¹⁴⁴ It may be more realistic to evaluate effectiveness of injury prevention strategies in the realworld using a cohort study rather than attempting to assess efficacy in a more controlled environment which requires the need of a control group.

Research utilising approaches other than randomised controlled trials, such as a recent observational study²⁰⁰ and nonrandomised intervention study,²⁰¹ has demonstrated how partnerships between researchers and high-performance sport can be mutually beneficial, supporting both high-performance outcomes and improved professional practice.¹⁶⁴ Research in high-performance sport must remain relevant for those at the 'coal-face'²⁰² and must maintain rigour to ensure studies are not poorly interpreted or misleading.²⁰³ To conduct sound scientific evaluation, it is important to monitor potential

athlete recruitment bias, the compliance and fidelity of the intervention, drop-out rates and adverse effects of the intervention. Establishing efficacy of an injury prevention strategy remains critical for the injury prevention implementation process.⁴³

Netball specific scientific evaluation of injury prevention strategies (TRIPP Stage 4)

We found proprioception improves and is maintained during a netball session with either SEP or selfapplied taping.¹³⁸ There have been limited scientific evaluations of injury prevention strategies in the sport of netball. Two biomechanical studies have examined the effect of an ankle brace or tape on lower limb electromyographic, kinematic and kinetic variables during a netball-specific landing task.^{204,205} Another study reported an improvement in lower-limb biomechanics when 11–13-year-old netballers underwent neuromuscular training.²⁰⁶ Additional research studies have been conducted within netball to investigate the effect of specific training on performance but not on injury prevention.²⁰⁷⁻²⁰⁹ Given netball's injury profile has not changed over the past 20 years,^{58,114} it is a concern that limited research has been conducted to support the efficacy of enduring netball-specific injury prevention strategies.

TRIPP Stage 5 (Describe intervention context to inform implementation strategies)

Reflections and learnings from applying TRIPP Stage 5 in the field

Implementation science is an emerging area within sports medicine. Applying implementation science within sports injury prevention was first described in 2006.⁴³ It is widely accepted that it takes an estimated average of 17 years for research evidence to reach clinical practice;²¹⁰⁻²¹² perhaps the same will hold true for implementation science to become more mainstream within injury prevention initiatives. The body of knowledge in sports medicine implementation is growing^{76,79,83,84,213} but utilising implementation science would not yet be regarded as common practice,^{76,98} even though international sporting federations highly value this type of research.⁹⁷ Several tools can be utilised to help fulfill TRIPP Stage 5, such as RE-AIM SSM (Sports Setting Matrix),⁷⁶ concept mapping,^{88,213} intervention mapping Step 5⁷⁹ and the knowledge and attitudes questionnaire.⁹¹ These tools can be used to either help plan implementation or evaluate implementation of already established injury prevention strategies. The challenge is to select the most appropriate tool for the setting in which the strategy has been applied.

Coach education has been recognised as an effective implementation strategy for achieving high compliance rates for injury prevention initiatives. In a South African rugby study, it was demonstrated that coach education was critical for improving athlete injury prevention behaviours.²¹⁴ In netball and soccer, coach education has also been shown to change the way coaches coach and a vehicle for indirectly educating athletes.^{85,92} The principles of injury prevention need be included in all coaching accreditation courses for implementation of injury prevention strategies to be successful.

All socioecological levels of an organisation need to be involved and committed to implementation planning for an injury prevention strategy to be effective.^{76,215} Coaching staff alone cannot deliver a successful program, as they rely on the athletes to attend training, execute the program with fidelity and integrate the program into their other training sessions. The coach needs to be supported by managers and the executive above them to ensure they receive the necessary training and education as well as resources and support materials that are regularly reviewed and updated. Often national and State Sporting Organisation (SSO) executive endorsement is critical for adoption, implementation and maintenance of an injury prevention strategy.^{76,213} For greater implementation success, TRIPP Stage 5 should emphasise the importance of involving all socioecological levels of operation within sport.

Netball specific implementation planning (TRIPP Stage 5)

Concept mapping was used to help improve the implementation of the KNEE program.¹⁶² Netball Australia implemented the KNEE program in 2015 but high lower-limb injury rates were still a feature of the 2018 17/U & 19/U ANNC.¹¹⁴ Using concept mapping, the following challenges were determined to be the most important and difficult to overcome when implementing the KNEE program in pre-elite netball: 'athlete engagement', 'supervision and correction of technique', 'time constraints', athlete technique', 'education' and 'support staff resourcing'. With this new knowledge, intervention mapping Step 5 can now be used to improve implementation by effectively attending to each of these recognised barriers.

Five netball injury prevention implementation studies have been identified within the literature.^{85,92-95} Areas for improvement include program modifications,⁸⁵ coach education,^{85,92} emphasis placed on injury prevention benefits to achieve attitudinal change, and the involvement of significant others and role models to promote injury prevention programs.⁹³ Additionally, coach and player attitudes to injury need further investigation, as a high prevalence of players playing-on despite being injured was identified.⁹⁵ These studies provide valuable insight for improving implementation of current netball injury prevention strategies and inform future implementation planning.

TRIPP Stage 6 (Evaluate effectiveness of preventive measures to implementation context)

Reflections and learnings from applying TRIPP Stage 6 in the field

This stage involves implementing an intervention in a real-world context and evaluating its effectiveness. While ankle sprains and ACL ruptures are recognised as the most frequent sports incapacity injuries within pre-elite netball athletes,¹¹⁴ a joint decision with Netball Australia was made to implement a foot blister prevention strategy at the 2019 17/U & 19/U ANNC. This decision was due in part to Netball Australia's KNEE program being already implemented to reduce lower limb injuries.¹⁸⁰ Barriers to implementation of the KNEE program were identified in Chapter 5;¹⁶² however, there was insufficient time to address these barriers and then assess the effectiveness of the KNEE program at the 2019 17/U & 19/U ANNC. Instead, a proven foot blister prevention strategy was implemented as blisters can be easily prevented with correct footwear and foot hygiene.^{120,121}

We repeated the 2018 17/U & 19/U ANNC medical attention and self-report injury surveillance in 2019 (Chapter 6). Netball Australia initiated injury surveillance at the 2017 17/U & 19/U ANNC, asking team physiotherapists to complete paper forms for every injury their athletes sustained. Team physiotherapists were required to complete this task in addition to maintaining their own injury records. In 2018, injury surveillance was advanced by introducing electronic injury records for the team physiotherapists to reduce their workload and simplify analysis.¹¹⁴ Additionally, athlete self-report injury/illness data were collected to compare injury surveillance methods and potentially improve accuracy of data collection.¹¹⁴ In 2019, team physiotherapists and athletes were provided with 2018 results and were asked to repeat the same injury surveillance. The athlete response rate improved to above 70% across all four health problem questionnaires in 2019, compared to 2018 when the response rate was below 50% for the two post-tournament health problem questionnaires. There were also fewer missing data for medical attention injuries; for example, in 2018, 32 (31%) injury records did not

include mechanism of injury data, which decreased to 10 (11%) incomplete injury records in 2019. This improved HPQ response rate and medical attention data collection was possibly due to athletes and physiotherapists being more familiar with the AMS in 2019 and results from 2018 had been shared which possibly provided more context. Also, coaches and high performance managers actively endorsed the research in 2019 compared to 2018. Establishing quality injury surveillance takes time and commitment by all when working within high-performance sport, where surveillance is an ongoing process enduring beyond any research project timeframes. Injury surveillance needs to be a continuous, evolving process, rather than just the first and last step of the TRIPP framework. It is advocated that prevention frameworks should be updated to reflect this ongoing need for the organisations involved.

Netball specific evaluation of effectiveness of injury prevention strategy

One previous netball study has examined the effectiveness of an injury prevention strategy in the 'real world'.⁹⁹ Elphinston and Hardman⁹⁹ provide a realistic view of implementing an injury prevention strategy for an international netball team. Using a multidisciplinary team (i.e., coach, physiologist, physiotherapist and a strength and conditioning coach), multiple injury prevention strategies were implemented including: 1) multidisciplinary athlete profiling, 2) athlete performance self-assessments, 3) functional stability program, 4) strength and conditioning programs, 5) individual prophylactic programs, 6) regular multidisciplinary meetings and 7) a Head Physiotherapist being appointed to the program. Injury incidence rates and statistical analysis were not reported, making it difficult to ascertain the relative effectiveness of this approach. Nevertheless, this case study provided a realistic example of the high-performance sport environment in an operational and multidisciplinary sense, illustrating the need for researchers to adapt their approach in order to contribute to injury prevention.

Foot blisters are a common occurrence in netball but there is no evidence of foot blister prevention strategy implementation within the netball population. Of the many netball injury surveillance studies,⁵⁴ only two have reported foot blister injury rates, once in 1986⁵⁵ and then again in 2000.⁵⁸ In 2018, foot blisters continued to be problematic at the 17/U & 19/U ANNC, causing player discomfort and resulting in sports incapacity for two athletes.¹¹⁴ We found implementation of a foot blister pack and advice sheet reduced the number of sports incapacity foot blisters from two in 2018 to zero in 2019 and all foot

blisters from eleven at the 2018 17/U & 19/U ANNC to eight at the 2019 17/U & 19/U ANNC, but this was not statistically significant. This strategy could be improved by involving the team physiotherapists in the research project earlier to increase their engagement and highlight the important role they play in educating athletes about foot blister prevention. Given foot blisters have been an issue for netball athletes for over 35 years and can lead to sports incapacity, and should be easily prevented, it is recommended that foot blister prevention continue to be prioritised.

Thesis strengths

This thesis has applied all stages of the TRIPP framework within a sport, which has not been previously demonstrated within the relevant literature. A feature and strength of this body of work is its significant breadth of mixed methodological approaches, including observational studies, a systematic review, randomised controlled trial, concept mapping and an interventional cohort study.

In both observational cohort studies¹¹⁴ (and Chapter 6) we applied the STROBE statement⁷² to ensure high-quality reporting. The OSICS¹²⁶ was used for coding injury diagnosis and the IDCF¹²⁴ guided our definitions of clinical examination and sports incapacity injuries. A valid and reliable questionnaire, the Oslo Sports Trauma Research Centre (OSTRC) Health Problems Questionnaire (HPQ),²⁰ was also utilised to collect athlete self-report data. Adopting these measures has resulted in two valid and effective injury surveillance studies of pre-elite netball athletes. While published after the studies were completed, our overall approach and methods align to the recommendations within the IOC's statement on methods for recording and reporting epidemiological data on injury and illness in sport.⁴⁷

The systematic review¹⁰⁴ protocol was registered on the International Prospective Register of Systematic Reviews (PROSPERO) (http://www.crd.york.ac.uk/PROSPERO) and the Preferred Reporting Items for Systematic Reviews and Meta-analyses (PRISMA) guidelines²¹⁶ were followed. Covidence systematic review software (http://www.covidence.org) was employed to ensure accurate article screening and data extraction. The FTEM framework⁵ was used to classify athlete developmental level and the Oxford Centre of Evidence-based Medicine (OCEBM) Levels of Evidence²¹⁷ was used to determine the hierarchical levels of evidence. To establish transparency of evidence, risk of bias was
assessed using Cochrane Collaboration's tool for assessing the risk of bias in randomised trials.¹⁰⁵ These provisions resulted in a high-quality systematic review identifying level 2 and 3 studies suggesting exercise and psychology interventions may prevent injury in pre-elite athletes.

We produced level 1 evidence by conducting a randomised controlled trial,¹³⁸ demonstrating proprioception improves and is maintained during a netball session with either SEP or self-applied ankle tape. We used the AMEDA, which has been shown to be a valid^{218,219} and reliable instrument, to test proprioception.²²⁰ To mitigate a learning effect which has been previously reported,^{220,221} we randomised the order of tape condition. Double-blinding of our participants would reduce risk of bias; however, because of the nature of the applied intervention (participants clearly knew if they were strapped and who strapped them) we were unable to employ this strategy. Advanced statistical analysis was utilised, such as a mixed effects generalised linear model to test the effect of the taping condition on proprioception over time and a three-way interaction model to assess learning effect. Employing these strategies minimised risk of bias, resulting in a trustworthy study design.

We used concept mapping, which is a real-world pragmatic consultative method to understand the barriers to implementing Netball Australia's KNEE program by creating and organising the ideas of our netball coaches, physiotherapists, and strength and conditioning coaches.¹⁶² Analysis of 69 concept mapping studies revealed strong internal representational validity and very strong sorting and rating reliability for this type of research.²²² We also succeeded in recruiting and retaining a high percentage of invited participants, with 39 of 47 (83%) contributing CM data, thus providing good representation of pre-elite netball coaching and support staff.

Another strength of this body of work is that it focused on the pre-elite population, whose vulnerability to injury, due to a range of risk factors is well recognised. To ensure efficacy and effectiveness, injury prevention strategies must be specific to their target population. Because the pre-elite stage of an athlete's development is critical to their overall athletic development and a critical precedent to future high-performance, we targeted this population.

Finally, we demonstrated how effective organisational collaboration and linkage (i.e., University of Canberra, Australian Institute of Sport and Netball Australia) is conducive to meaningful and impactful research with mutual and system level benefit. The Australian Institute of Sport and University of Canberra recognised the need to develop knowledge around injury/illness prevention in pre-elite athletes and Netball Australia prioritised injury/illness prevention for their pre-elite athletes. This collaboration was effective because the research organisations were guided by Netball Australia's research needs.

Thesis limitations

We attempted to apply the entire TRIPP framework with Netball Australia's pre-elite athletes who had already adopted some injury prevention strategies. This resulted in a wide range of studies but a reduction in depth. Additionally, the timing of the 17/U & 19/U ANNC and national camps also dictated what could be accomplished within those time constraints. And as previously discussed, the concept mapping project could have been improved by the inclusion of athletes. Further, our understanding of the foot blister intervention would have been enhanced by collecting fidelity data. Also, adding an additional AMEDA training session prior to data collection may have reduced the initial learning effect.

We did not follow a specific implementation framework when implementing our foot blister prevention strategy at the 2019 17/U & 19/U ANNC. However, the strategy itself was a proven intervention that has been utilised for many years within the Australian Institute of Sport's Medical Department. Pragmatically, we did not undertake formal implementation planning due to time constraints prior to the commencement of the 2019 17/U & 19/U ANNC. This is not optimal for effective implementation of an injury prevention strategy; however, based on clinical experience and evidence we proceeded with the intervention to determine the effectiveness of an existing tool as a separate question to the development and implementation of a tool. If a framework such as REAIM was used, team physiotherapists, athletes and coaches would have been engaged earlier and invested in the strategy, possibly leading to greater compliance.

Directions for further research

Through the course of this research, we have established annual injury surveillance tools and processes at the 17/U & 19/U ANNC. Further research opportunities include ongoing injury surveillance at these championships, where injury surveillance can be further developed and refined.

Barriers to implementation of the KNEE program have been identified by coaches and support staff within this body of work. Further research is required to tap into and understand the perspectives of the athlete in implementing this program and identify potential barriers to compliance and effective implementation. It is recommended that IM (Step 5) then be applied to address these barriers and inform and support effective implementation planning.

There were three concussions reported at the 2018 17/U & 19/U ANNC and five concussions at the 2019 17/U & 19/U ANNC. We have not addressed this issue within this thesis but preventing concussion should be prioritised in future research endeavours as it is an injury that results in sports incapacity and may contribute to long-term sequelae. Historically these injuries have not routinely been included in netball injury surveillance studies. Future surveillance studies and systems should include concussion within their data capture.

Lumbar pain was the second most frequent medical attention injury at the 2018 (n=10) and 2019 (n=12) 17/U & 19/U ANNC. There is no evidence for a specific prevention strategy for lumbar pain in preelite netball athletes, however a functional stability program for an international netball squad has been shown to reduce the incidence of lumbar pain.⁹⁹ With Lumbar pain being such a frequent injury in this population, a prevention strategy should be developed and implemented.

One study has previously reported the effect of research-invested clinicians on quality of injury data collection. This can potentially have a significant impact on research outcomes and therefore warrants further research into potential factors that influence clinician data collection behaviours, such as education, experience and the job descriptions within the sports medicine support roles.

Injury has been found to be a major contributing factor to pre-elite athlete dropout.³ Further research should be implemented with a focus on this vulnerable population to further ascertain the extent of this

issue. This should be of primary importance for sporting organisations at all levels to better support sustainable pathways to high performance into the future.

Conclusion

The aim of this thesis was to develop a holistic strategy to reduce the burden of injuries in pre-elite athletes, using netball as an exemplar sport. To achieve this, we applied the TRIPP framework to a pre-elite netball athlete population using a mixed-methods approach employing established approaches and protocols.

This thesis reconfirmed that ankle sprains are the most frequent injury and ACL ruptures the most severe injury sustained by pre-elite netball athletes. A novel finding was that concussions are a relatively frequent sports incapacity injury in this sport. Our findings support the ongoing need for injury surveillance and the need for injury prevention strategies for pre-elite netball athletes to continue to focus on the most frequent and incapacitating injuries: ankle sprain, lumbar pain, ACL rupture, concussion, and foot blisters.

The practical application of this body of work revealed the importance of researchers working with sporting organisations that oversee talent pathways (i.e., State Sporting Organisations and their partner National Sporting Organisations). Without the collaboration between the Australian Institute of Sport, University of Canberra, Netball Australia and its State member organisations, this research would not have been possible. A flexible contingency approach towards injury prevention assists organisations to improve existing practices rather than following a sports injury prevention framework like a recipe. Although the TRIPP framework is intended to be followed in a sequential order, a learning from this thesis is that it cannot be applied in such a manner without significant challenges to relationships. For example, Netball Australia had already invested a lot of time and resources into developing and implementing the KNEE program. Instead of developing a new program, we worked with Netball Australia to learn how implementation of the KNEE program could be improved. While all steps of the TRIPP framework are essential for injury prevention, the outcomes of this thesis, beyond its individual studies, indicate they should be used as a guide to continual improvement of what the sport has already

implemented. In doing so, researchers embedded within sport can add value to the organisation processes, reducing the need for translation of findings at the end of the process.

Injury prevention processes such as injury surveillance and implementation methods must be tailored to the pre-elite athlete. These processes must be cognisant of the resources available to the organisation. Coaches, physiotherapists, and strength and conditioning coaches are pivotal for injury prevention success, but at the pre-elite level they are not available to the same extent as at the elite level due to resourcing constraints. Sporting organisations wishing to improve injury rates should be aware of the practical barriers to implementing any systems or programs. For this to occur, injury prevention must be prioritised by sporting organisations and be reflected within related policy, strategy, organisational structure and resourcing.

References

- 1. Boston Consulting Group. *Intergenerational Review of Australian Sport 2017*. Canberra: Australian Sports Commission, 2017.
- 2. Sport Australia. *Sport 2030: Participation, performance, integrity, industry*. Canberra: Department of Health, 2018.
- 3. Huxley DJ, O'Connor D, Healey PA. An examination of the training profiles and injuries in elite youth track and field athletes. *Eur J of Sport Sci.* 2014;14(2):185-192. doi: 10.1080/17461391.2013.809153
- 4. Bergeron MF, Mountjoy M, Armstrong N, et al. International Olympic Committee consensus statement on youth athletic development. *British Journal of Sports Medicine*. 2015;49(13):843-851. doi: 10.1136/bjsports-2015-094962
- 5. Gulbin JP, Croser MJ, Morley EJ, et al. An integrated framework for the optimisation of sport and athlete development: a practitioner approach. *Journal of sports sciences*. 2013;31(12):1319-1331. doi: 10.1080/02640414.2013.781661
- 6. Drew MK, Raysmith BP, Charlton PC. Injuries impair the chance of successful performance by sportspeople: a systematic review. *Br J Sports Med.* 2017;51(16):1209-1214. doi: 10.1136/bjsports-2016-096731
- Visnes H, Aandahl HÅ, Bahr R. Jumper's knee paradox—jumping ability is a risk factor for developing jumper's knee: a 5-year prospective study. *British Journal of Sports Medicine*. 2013;47(8):503-507. doi: 10.1136/bjsports-2012-091385
- 8. von Rosen P, Floström F, Frohm A, et al. Injury patterns in adolescent elite endurance athletes participating in running, orienteering, and cross-country skiing. *International Journal of Sports Physical Therapy*. 2017;12(5):822-832. Accessed August 8, 2018. http://www.ncbi.nlm.nih.gov/pmc/articles/PMC5685405/.
- 9. Loose O, Fellner B, Lehmann J, et al. Injury incidence in semi-professional football claims for increased need of injury prevention in elite junior football. *Knee Surg Sports Traumatol Arthrosc.* 2019;27(3):978-984. doi: 10.1007/s00167-018-5119-8
- 10. Toohey LA, Drew MK, Cook JL, et al. Is subsequent lower limb injury associated with previous injury? A systematic review and meta-analysis. *British Journal of Sports Medicine*. 2017;51(23):1670. doi:
- 11. Toohey LA, Drew MK, Finch CF, et al. A 2-Year Prospective Study of Injury Epidemiology in Elite Australian Rugby Sevens: Exploration of Incidence Rates, Severity, Injury Type, and Subsequent Injury in Men and Women. *The American Journal of Sports Medicine*. 2019;47(6):1302-1311. doi: 10.1177/0363546518825380
- Gulbin J, Oldenziel KE, Weissensteiner J, et al. A look through the rear view mirror: developmental experiences and insights of high performance athletes. *Talent Dev Excell*. 2010;2:149-164. Accessed 6/16/2021. <u>https://www.researchgate.net/publication/285864365</u>.
- Hollings S, Hume P. Progression of elite New Zealand and Australian junior athletes to senior representation. *New Stud Athl.* 2011;26(3/4):127-135. Accessed June 16, 2021. <u>https://www.worldathletics.org/nsa</u>.
- 14. DiFiori JP, Benjamin HJ, Brenner J, et al. Overuse injuries and burnout in youth sports: a position statement from the American Medical Society for Sports Medicine. *Clin J Sport Med.* 2014;24(1):3-20. doi: 10.1097/jsm.0000000000000000
- 15. Bahr R. Demise of the fittest: are we destroying our biggest talents? *British Journal of Sports Medicine*. 2014;48(17):1265-1267. doi: 10.1136/bjsports-2014-093832
- 16. Moseid CH, Myklebust G, Fagerland MW, et al. The prevalence and severity of health problems in youth elite sports: a 6-month prospective cohort study of 320 athletes. *Scandinavian Journal of Medicine & Science in Sports*. 2017;28(4):1412-1423. doi: 10.1111/sms.13047
- 17. von Rosen P, Heijne A, Frohm A, et al. High injury burden in elite adolescent athletes: a 52week prospective study. *J Athl Train*. 2018;53(3):262-270. doi: 10.4085/1062-6050-251-16

- Le Gall F, Carling C, Reilly T. Injuries in young elite female soccer players: an 8-season prospective study. *American Journal of Sports Medicine*. 2008;36(2):276-284. doi: 10.1177/0363546507307866
- Rejeb A, Johnson A, Vaeyens R, et al. Compelling overuse injury incidence in youth multisport athletes. *Eur J Sport Sci.* 2017;17(4):495-502. doi: 10.1080/17461391.2016.1275820
- 20. Clarsen B, Rønsen O, Myklebust G, et al. The Oslo Sports Trauma Research Center questionnaire on health problems: a new approach to prospective monitoring of illness and injury in elite athletes. *Br J Sports Med.* 2014;48(9):754-760. doi: 10.1136/bjsports-2012-092087
- 21. Jacobsson J, Timpka T, Kowalski J, et al. Injury patterns in Swedish elite athletics: annual incidence, injury types and risk factors. *Br J Sports Med.* 2013;47(15):941-952. doi: 10.1136/bjsports-2012-091651
- 22. Giza E, Mithofer K, Farrell L, et al. Injuries in women's professional soccer. *British Journal* of Sports Medicine. 2005;39(4):212-216. doi: 10.1136/bjsm.2004.011973
- 23. Gagné F. Building gifts into talents: Detailed overview of the DMGT 2.0. In: MacFarlane B, Stambaugh T, eds. *Leading change in gifted education: The festschrift of Dr. Joyce VanTassel-Baska*. Waco, TX: Prufrock Press; 2009:61-80.
- 24. International Netball Federation. Regions & members. International Netball Federation. 2021. Accessed May 19, 2021. <u>https://netball.sport/inside-inf/regions-members</u>.
- 25. International Netball Federation. International rules of netball. International Netball Federation. 2018. Accessed July 30, 2018. <u>http://netball.org/wp-content/uploads/2018/06/INF-Rules-of-Netball-2018-Edition-text-correction.pdf</u>.
- 26. McManus A, Stevenson MR, Finch CF. Incidence and risk factors for injury in non-elite netball. *Journal Of Science And Medicine In Sport*. 2006;9(1-2):119-124. doi: 10.1016/j.jsams.2006.03.005
- 27. Steele JR. Biomechanical factors affecting performance in netball. *Sports Medicine*. 1990;10(2):88-102. doi: 10.2165/00007256-199010020-00003
- 28. Rom N, Aitken O, Sanchez G. *Change the Routine: Report on the Independent Review into Gymnastics in Australia (2021).* Sydney: Australian Human Rights Commission, 2021.
- 29. Howman D, Nicol L, Vickery R. *Independent Review of Gymnastics New Zealand*. Gymnastics New Zealand, 2021. Accessed May 16, 2021. <u>https://www.gymnasticsnz.com/wp-content/uploads/2021/02/Gymnastics-New-Zealand-Independent-Report-10-February-2021.pdf</u>.
- 30. Human Rights Watch. "I was hit so many times I can't count" abuse of child athletes in Japan. Human Rights Watch, 2020. Accessed May 16, 2021. https://www.hrw.org/report/2020/07/20/i-was-hit-so-many-times-i-cant-count/abuse-child-athletes-japan.
- 31. Whyte A, Cave M, Couch T. *The Whyte Review*. UK Sport & Sport England, 2021. Accessed May 16, 2021. <u>https://www.whytereview.org</u>.
- 32. Soltanzadeh S, Mooney M. Systems thinking and team performance analysis. *Int J Sports Sci Coach.* 2016;3(2):184-191. doi: 10.1123/iscj.2015-0120
- 33. Raysmith BP, Drew MK. Performance success or failure is influenced by weeks lost to injury and illness in elite Australian track and field athletes: a 5-year prospective study. *J Sci Med Sport*. 2016;19(10):778-783. doi: 10.1016/j.jsams.2015.12.515
- 34. Hägglund M, Waldén M, Magnusson H, et al. Injuries affect team performance negatively in professional football: an 11-year follow-up of the UEFA Champions League injury study. *British Journal of Sports Medicine*. 2013;47(12):738-742. doi: 10.1136/bjsports-2013-092215
- 35. Podlog L, Buhler CF, Pollack H, et al. Time trends for injuries and illness, and their relation to performance in the National Basketball Association. *Journal of science and medicine in sport*. 2015;18(3):278-282. doi: 10.1016/j.jsams.2014.05.005
- 36. Verrall GM, Kalairajah Y, Slavotinek J, et al. Assessment of player performance following return to sport after hamstring muscle strain injury. *Journal of Science and Medicine in Sport*. 2006;9(1-2):87-90. doi: 10.1016/j.jsams.2006.03.007

- 37. Alonso J-M, Jacobsson J, Timpka T, et al. Preparticipation injury complaint is a risk factor for injury: a prospective study of the Moscow 2013 IAAF Championships. *Br J Sports Med.* 2015;49(17):1118-1124. doi: 10.1136/bjsports-2014-094359
- 38. von Rosen P, Heijne A. Substantial injuries influence ranking position in young elite athletes of athletics, cross-country skiing and orienteering. *Scand J Med Sci Sports*. 2018;28(4):1435-1442. doi: 10.1111/sms.13032
- 39. Larruskain J, Lekue JA, Martin-Garetxana I, et al. Injuries are negatively associated with player progression in an elite football academy. *Sci Med Footb*. 2021:1-10. doi: 10.1080/24733938.2021.1943756
- 40. Gordis L. *Epidemiology*. 5th ed. Elsevier Saunders; 2014.
- 41. Jacobsson J, Timpka T. Classification of prevention in sports medicine and epidemiology. *Sports Medicine* 2015;45(11):1483-1487. doi: 10.1007/s40279-015-0368-x
- 42. van Mechelen W, Hlobil H, Kemper HC. Incidence, severity, aetiology and prevention of sports injuries: a review of concepts. *Sports Med.* 1992;14(2):82-99. doi: 10.2165/00007256-199214020-00002
- 43. Finch C. A new framework for research leading to sports injury prevention. *J Sci Med Sport*. 2006;9(1):3-9. doi: 10.1016/j.jsams.2006.02.009
- 44. Chalmers DJ. Injury prevention in sport: not yet part of the game? *Inj Prev.* 2002;8(Suppl 4):IV22-IV25. doi: 10.1136/ip.8.suppl_4.iv22
- 45. van Mechelen W. The severity of sports injuries. *Sports Medicine*. 1997;24(3):176-180. doi: 10.2165/00007256-199724030-00006
- 46. Finch CF. An overview of some definitional issues for sports injury surveillance. *Sports Medicine*. 1997;24(3):157-163. doi: 10.2165/00007256-199724030-00002
- 47. Bahr R, Clarsen B, Derman W, et al. International Olympic Committee consensus statement: methods for recording and reporting of epidemiological data on injury and illness in sport 2020 (including STROBE Extension for Sport Injury and Illness Surveillance (STROBE-SIIS)). *Br J Sports Med.* 2020;54(7):372-389. doi: 10.1136/bjsports-2019-101969
- 48. Fuller CW, Ekstrand J, Junge A, et al. Consensus statement on injury definitions and data collection procedures in studies of football (soccer) injuries. *Clin J Sport Med.* 2006;16(2):97-106. doi: 10.1136/bjsm.2005.025270
- 49. Bahr R. No injuries, but plenty of pain? On the methodology for recording overuse symptoms in sports. *British Journal of Sports Medicine*. 2009;43(13):966-972. doi: 10.1136/bjsm.2009.066936.
- 50. Whalan M, Lovell R, Sampson JA. Do Niggles Matter? Increased injury risk following physical complaints in football (soccer). *Sci Med Footb*. 2020;4(3):216-224. doi: 10.1080/24733938.2019.1705996
- 51. Giesecke J. *Modern infectious disease epidemiology*. 2nd ed. Taylor & Francis; 2001.
- 52. Timpka T, Jacobsson J, Ekberg J, et al. Meta-narrative analysis of sports injury reporting practices based on the Injury Definitions Concept Framework (IDCF): A review of consensus statements and epidemiological studies in athletics (track and field). *Journal of Science and Medicine in Sport*. 2015;18(6):643-650. doi: 10.1016/j.jsams.2014.11.393
- 53. Nielsen RO, Debes-Kristensen K, Hulme A, et al. Are prevalence measures better than incidence measures in sports injury research? *British Journal of Sports Medicine*. 2019;53(7):396. doi: 10.1136/bjsports-2017-098205
- 54. Downs C, Snodgrass SJ, Weerasekara I, et al. Injuries in netball: a systematic review. *Sports Med Open.* 2021;7(1):3. doi: 10.1186/s40798-020-00290-7
- 55. Hopper D. A survey of netball injuries and conditions related to these injuries. *Aust J Physiother*. 1986;32(4):231-239. doi: 10.1016/s0004-9514(14)60656-5
- 56. Pickering Rodriguez EC, Watsford ML, Bower RG, et al. The relationship between lower body stiffness and injury incidence in female netballers. *Sports Biomech.* 2017;16(3):361-373. doi: 10.1080/14763141.2017.1319970
- 57. Langeveld E, Coetzee FF, Holtzhausen LJ. Epidemiology of injuries in elite South African netball players. *S Afr J Res Sport Phys Educ Recreat*. 2012;34(2):83-93. Accessed July 30, 2018. <u>https://www.ajol.info/index.php/sajrs/article/view/84547</u>.

- 58. Hume PA, Steele JR. A preliminary investigation of injury prevention strategies in Netball: are players heeding the advice? *Journal Of Science And Medicine In Sport*. 2000;3(4):406-413. doi: 10.1016/s1440-2440(00)80007-9.
- 59. Hopper D, Elliott B. Lower limb and back injury patterns of elite netball players. *Sports Med.* 1993;16(2):148-162. doi: 10.2165/00007256-199316020-00006
- Meeuwisse W. Assessing causation in sport Injury: a multifactorial model. *Clinical Journal of Sport Medicine*. 1994;4(3):166-170. Accessed May 22, 2021. https://journals.lww.com/cjsportsmed/abstract/1994/07000.
- 61. Bahr R, Krosshaug T. Understanding injury mechanisms: a key component of preventing injuries in sport. *British Journal of Sports Medicine*. 2005;39(6):324-329. doi: 10.1136/bjsm.2005.018341
- 62. Hulme A, Salmon PM, Nielsen RO, et al. Closing Pandora's Box: adapting a systems ergonomics methodology for better understanding the ecological complexity underpinning the development and prevention of running-related injury. *Theor Issues Ergon Sci.* 2017;18(4):338-359. doi: 10.1080/1463922X.2016.1274455
- 63. Coetzee D, Langeveld E, Holtzhausen L. Training habits, training surface and injuries among South African netball players. *S Afr J Res Sport Phys Educ Recreat*. 2014;36(3):39-49. doi: doi:10.10520/EJC163548
- 64. Otago L. Kinetic analysis of landings in netball: is a footwork rule change required to decrease ACL injuries? *Journal Of Science And Medicine In Sport*. 2004;7(1):85-95. doi: 10.1016/s1440-2440(04)80047-1
- 65. Stuelcken MC, Mellifont DB, Gorman AD, et al. Mechanisms of anterior cruciate ligament injuries in elite women's netball: a systematic video analysis. *Journal of Sports Sciences*. 2016;34(16):1516-1522. doi: 10.1080/02640414.2015.1121285
- 66. Maulder PS. Dominant limb asymmetry associated with prospective injury occurrence. *S Afr J Res Sport Phys Educ Recreat*. 2013;35(1):121-131. doi: 10520/EJC134231
- 67. Donaldson A, Lloyd DG, Gabbe BJ, et al. Scientific evidence is just the starting point: A generalizable process for developing sports injury prevention interventions. *J Sport Health Sci.* 2016;5(3):334-341. doi: 10.1016/j.jshs.2016.08.003
- 68. Hanson D, Allegrante JP, Sleet DA, et al. Research alone is not sufficient to prevent sports injury. *Br J Sports Med.* 2014;48(8):682-684. doi: 10.1136/bjsports-2012-091434
- 69. Fox AS, Bonacci J, Saunders N. The relationship between performance of a single-leg squat and leap landing task: moving towards a netball-specific anterior cruciate ligament (ACL) injury risk screening method. *Sports biomech.* 2020;19(4):493-509. doi: 10.1080/14763141.2018.1498535
- Sinclair JK, Vincent H, Richards JD. Effects of prophylactic knee bracing on knee joint kinetics and kinematics during netball specific movements. *Physical Therapy in Sport*. 2017;23:93-98. doi: 10.1016/j.ptsp.2016.08.005
- Schulz KF, Altman DG, Moher D, et al. CONSORT 2010 Statement: updated guidelines for reporting parallel group randomised trials. *BMC Medicine*. 2010;8(1):18. doi: 10.1186/1741-7015-8-18
- 72. von Elm E, Altman DG, Egger M, et al. The Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) statement: guidelines for reporting observational studies. *Lancet.* 2007;370(9596):1453-1457. doi: 10.1016/s0140-6736(07)61602-x
- 73. Myklebust G, Skjølberg A, Bahr R. ACL injury incidence in female handball 10 years after the Norwegian ACL prevention study: important lessons learned. *British Journal of Sports Medicine*. 2013;47(8):476-479. doi: 10.1136/bjsports-2012-091862
- 74. de Villiers JE, Venter RE. Barefoot training improved ankle stability and agility in netball players. *Int J Sports Sci Coa.* 2014;9(3):485-495. doi: 10.1260/1747-9541.9.3.485
- 75. Zulkarnain J, Khairullina K. The pilot study on Down to Earth (D2E) injury prevention program among varsity netball players. *Gazz Med Ital Arch Sci Med.* 2019;178(4):188-194. doi: 10.23736/S0393-3660.18.03766-X
- 76. Finch C, Donaldson A. A sports setting matrix for understanding the implementation context for community sport. *Br J Sports Med.* 2010;44(13):973-978. doi: 10.1136/bjsm.2008.056069

- 77. Kok G. A practical guide to effective behavior change: how to apply theory-and evidencebased behavior change methods in an intervention. *Eur Health Psychol.* 2014;16(5):156-170. doi: 10.31234/osf.io/r78wh
- 78. Glasgow RE, Vogt T, Boles SM. Evaluating the public health impact of health promotion interventions: the RE-AIM framework. *Am J Public Health*. 1999;89(9):1322-1327. doi: 10.2105/ajph.89.9.1322.
- 79. Donaldson A, Lloyd DG, Gabbe BJ, et al. We have the programme, what next? Planning the implementation of an injury prevention programme. *Injury Prevention*. 2017;23(4):273-280. doi: 10.1136/injuryprev-2015-041737
- 80. Bartholomew L, Parcel G, Kok G, et al. *Planning Health Promotion Programs: An Intervention Mapping Approach.* 3rd ed. Jossey-Bass; 2006.
- 81. Trochim WM, McLinden D. Introduction to a special issue on concept mapping. *Eval Program Plann.* 2017;60:166-175. doi: 10.1016/j.evalprogplan.2016.10.006
- 82. Eime R, Owen N, Finch C. Protective Eyewear Promotion. *Sports Medicine*. 2004;34(10):629-638. doi: 10.2165/00007256-200434100-00001
- 83. Donaldson A, Finch CF. Applying implementation science to sports injury prevention. *Br J Sports Med.* 2013;47(8):473-475. doi: 10.1136/bjsports-2013-092323
- 84. Donaldson A, Poulos RG. Planning the diffusion of a neck-injury prevention programme among community rugby union coaches. *Br J Sports Med.* 2014;48(2):151-159. doi: 10.1136/bjsports-2012-091551
- 85. Saunders N, Otago L, Romiti M, et al. Coaches' perspectives on implementing an evidenceinformed injury prevention programme in junior community netball. *Br J Sports Med.* 2010;44(15):1128-1132. doi: 10.1136/bjsm.2009.069039
- 86. Finch C, Gabbe B, Lloyd D, et al. Towards a national sports safety strategy: addressing facilitators and barriers towards safety guideline uptake. *Injury Prevention*. 2011;17(3):e4. doi: 10.1136/ip.2010.031385
- 87. Snider CE, Kirst M, Abubakar S, et al. Community-based participatory research: development of an emergency department–based youth violence intervention using concept mapping. *Academic Emergency Medicine*. 2010;17(8):877-885. doi: 10.1111/j.1553-2712.2010.00810.x
- 88. Ageberg E, Bunke S, Lucander K, et al. Facilitators to support the implementation of injury prevention training in youth handball: a concept mapping approach. *Scand J Med Sci Sports*. 2019;29:275-285. doi: 10.1111/sms.13323
- 89. Trochim WM, McLinden D. Introduction to a special issue on concept mapping. *Evaluation and Program Planning*. 2017;60:166-175. doi: 10.1016/j.evalprogplan.2016.10.006
- 90. Concept Systems Incorporated. Concept systems global max. Concept Systems Incorporated. 2019. Accessed January 14, 2019. <u>https://conceptsystemsglobal.com/</u>.
- 91. Shamlaye J, Tomšovský L, Fulcher ML. Attitudes, beliefs and factors influencing football coaches' adherence to the 11+ injury prevention programme. *BMJ Open Sport Exerc Med.* 2020;6(1):e000830. doi: 10.1136/bmjsem-2020-000830
- 92. Gianotti S, Hume PA, Tunstall H. Efficacy of injury prevention related coach education within netball and soccer. *Journal of Science and Medicine in Sport*. 2010;13(1):32-35. doi: 10.1016/j.jsams.2008.07.010
- White PE, Ullah S, Donaldson A, et al. Encouraging junior community netball players to learn correct safe landing technique. *Journal of Science and Medicine in Sport*. 2012;15(1):19-24. doi: 10.1016/j.jsams.2011.08.004
- 94. Saad T, Davies L, Smith M. Implementation of an injury prevention programme in community netball: an observational study. *J Sports Sci.* 2021:1-9. doi: 10.1080/02640414.2021.1923932
- 95. Whatman C, Walters S, Schluter P. Coach and player attitudes to injury in youth sport. *Phys Ther Sport.* 2018;32:1-6. doi: 10.1016/j.ptsp.2018.01.011
- 96. Eime R, Finch C, Wolfe R, et al. The effectiveness of a squash eyewear promotion strategy. *British journal of sports medicine*. 2005;39(9):681-685. doi: 10.1136/bjsm.2005.018366
- 97. Finch CF, Talpey S, Bradshaw A, et al. Research priorities of international sporting federations and the IOC research centres. *BMJ Open Sport Exerc Med.* 2016;2(1):e000168. doi: 10.1136/bmjsem-2016-000168

- 98. Klugl M, Shrier I, McBain K, et al. The prevention of sport injury: an analysis of 12,000 published manuscripts. *Clin J Sport Med.* 2010;20(6):407-412. doi: 10.1097/JSM.0b013e3181f4a99c.
- 99. Elphinston J, Hardman SL. Effect of an integrated functional stability program on injury rates in an international netball squad. *Journal of Science and Medicine in Sport*. 2006;9(1):169-176. doi: 10.1016/j.jsams.2005.11.002
- 100. International Netball Federation. History of netball. International Netball Federation. 2021. Accessed June 2, 2021. <u>https://netball.sport/game/history-of-netball</u>.
- 101. Emmons KM. Behavioral and social science contributions to the health of adults in the United States. In: Smedley B, Syme S, eds. *Promoting Health: Intervention Strategies from Social and Behavioral Research*. Washington, DC: National Academy Press; 2000:254-321.
- 102. Clarsen B, Bahr R, Myklebust G, et al. Improved reporting of overuse injuries and health problems in sport: an update of the Oslo Sport Trauma Research Center questionnaires. *Br J Sports Med.* 2020;54(7):390-396. doi: 10.1136/bjsports-2019-101337
- 103. Clarsen B, Myklebust G, Bahr R. Development and validation of a new method for the registration of overuse injuries in sports injury epidemiology: the Oslo Sports Trauma Research Centre (OSTRC) Overuse Injury Questionnaire. *British Journal of Sports Medicine*. 2013;47(8):495-502. doi: 10.1136/bjsports-2012-091524
- 104. Smyth EA, Newman P, Waddington G, et al. Injury prevention strategies specific to pre-elite athletes competing in Olympic and professional sports a systematic review. *Journal of science and medicine in sport*. 2019;22(8):887-901. doi: 10.1016/j.jsams.2019.03.002
- 105. Higgins JPT, Altman DG, Gøtzsche PC, et al. The Cochrane Collaboration's tool for assessing risk of bias in randomised trials. *BMJ*. 2011;343. doi: 10.1136/bmj.d5928
- 106. Dalen-Lorentsen T, Bjørneboe J, Clarsen B, et al. Does load management using the acute:chronic workload ratio prevent health problems? A cluster randomised trial of 482 elite youth footballers of both sexes. *British journal of sports medicine*. 2021;55(2):108-114. doi: 10.1136/bjsports-2020-103003
- 107. Reneker JC, Babl R, Pannell WC, et al. Sensorimotor training for injury prevention in collegiate soccer players: an experimental study. *Physical Ther Sport.* 2019;40:184-192. doi: 10.1016/j.ptsp.2019.09.012
- 108. Omi Y, Sugimoto D, Kuriyama S, et al. Effect of hip-focused injury prevention training for anterior cruciate ligament injury reduction in female basketball players: a 12-year prospective intervention study. *Am J Sports Med.* 2018;46(4):852-861. doi: 10.1177/0363546517749474
- 109. Reurink G, Goudswaard GJ, Moen MH, et al. Myotoxicity of injections for acute muscle injuries: A systematic review. *Sports Medicine*. 2014;44(7):943-956. doi: 10.1007/s40279-014-0186-6
- 110. Charlton PC, Drew MK, Mentiplay BF, et al. Exercise interventions for the prevention and treatment of groin pain and injury in athletes: a critical and systematic review. *Sports Med.* 2017;47(10):2011-2026. doi: 10.1007/s40279-017-0742-y
- 111. Vuurberg G, Hoorntje A, Wink LM, et al. Diagnosis, treatment and prevention of ankle sprains: update of an evidence-based clinical guideline. *British Journal of Sports Medicine*. 2018;52(15):956. doi: 10.1136/bjsports-2017-098106
- 112. Trojian TH, McKeag DB. Single leg balance test to identify risk of ankle sprains. *Br J Sports Med.* 2006;40(7):610-613. doi: 10.1136/bjsm.2005.024356
- 113. Han J, Anson J, Waddington G, et al. The role of ankle proprioception for balance control in relation to sports performance and injury. *Biomed Res Int.* 2015;2015:842804-842808. doi: 10.1155/2015/842804
- 114. Smyth EA, Piromalli L, Antcliff A, et al. A prospective study of health problems at the 2018 17/U and 19/U Australian National Netball Championships with comparison of surveillance methodology. *Journal of Science and Medicine in Sport*. 2020;23(3):215-221. doi: 10.1016/j.jsams.2019.10.004
- Finch C, Costa AD, Stevenson M, et al. Sports injury experiences from the Western Australian sports injury cohort study. *Australian and New Zealand Journal of Public Health*. 2002;26(5):462-467. doi: 10.1111/j.1467-842X.2002.tb00348.x

- 116. Kreisfeld R, Harrison J, Pointer S. *Australian sports injury hospitalisations 2011-12*. Canberra: Australian Institute of Health and Welfare, 2014.
- 117. Soligard T, Nilstad A, Steffen K, et al. Compliance with a comprehensive warm-up programme to prevent injuries in youth football. *British Journal of Sports Medicine*. 2010;44(11):787-793. doi: 10.1136/bjsm.2009.070672
- 118. SPORTAUS. AUSPLAY netball state of play report. 2019. Accessed January 23, 2020. https://www.clearinghouseforsport.gov.au/research/smi/ausplay/results/sport.
- 119. International Netball Federation. Current world rankings. 2019. Accessed January 23, 2020. https://netball.sport/events-and-results/current-world-rankings.
- 120. Knapik JJ, Hamlet MP, Thompson KJ, et al. Influence of boot-sock systems on frequency and severity of foot blisters. *Mil Med.* 1996;161(10):594-598. doi: 10.1093/milmed/161.10.594
- 121. Lipman GS, Sharp L, Christensen M, et al. Paper tape prevents foot blisters: a randomized prevention trial assessing paper tape in endurance distances II (Pre-TAPED II). *Clin J Sport Med.* 2016;26(5):362-368. doi: 10.1097/JSM.00000000000319.
- 122. Baussan E, Bueno MA, Rossi RM, et al. Analysis of current running sock structures with regard to blister prevention. *Text Res J.* 2012;83(8):836-848. doi: 10.1177/0040517512461698
- 123. Knapik JJ, Reynolds KL, Duplantis KL, et al. Friction blisters. Pathophysiology, prevention and treatment. *Sports Medicine*. 1995;20(3):136-147. doi: 10.2165/00007256-199520030-00002
- 124. Timpka T, Jacobsson J, Bickenbach J, et al. What is a sports injury? *Sports Medicine*. 2014;44(4):423-428. doi: 10.1007/s40279-014-0143-4
- 125. Clarsen B, Bahr R, Myklebust G, et al. Improved reporting of overuse injuries and health problems in sport: an update of the Oslo Sport Trauma Research Center questionnaires. *British Journal of Sports Medicine*. 2020:bjsports-2019-101337. doi: 10.1136/bjsports-2019-101337
- 126. Rae K, Orchard J. The Orchard Sports Injury Classification System (OSICS) version 10. *Clin J Sport Med.* 2007;17(3):201-204. doi: 10.1097/JSM.0b013e318059b536
- 127. Stovitz SD, Shrier I. Injury rates in team sport events: tackling challenges in assessing exposure time. *British Journal of Sports Medicine*. 2012;46(14):960-963. doi: 10.1136/bjsports-2011-090693
- 128. Kirkwood G, Hughes TC, Pollock AM. Results on sports-related injuries in children from NHS emergency care dataset Oxfordshire pilot: an ecological study. *Journal of the Royal Society of Medicine*. 2019;112(3):109-118. doi: 10.1177/0141076818808430
- 129. Finch CF, Clapperton AJ, McCrory P. Increasing incidence of hospitalisation for sport-related concussion in Victoria, Australia. *Med J Aust.* 2013;198(8):427-430. doi: 10.5694/mja12.11217
- 130. Saunders N, Otago L. Elite netball injury surveillance: implications for injury prevention. *Journal of Science and Medicine in Sport*. 2009;12:S63. doi: 10.1016/j.jsams.2008.12.148
- 131. Smith MMF, Mendis MD, Parker A, et al. Injury surveillance of an Australian community netball club. *Physical therapy in sport*. 2020;44:41-46. doi: 10.1016/j.ptsp.2020.04.004
- 132. Harris BS, Watson JC. Developmental considerations in youth athlete burnout: a model for youth sport participants. *Journal of clinical sport psychology*. 2014;8(1):1-18. doi: 10.1123/jcsp.2014-0009
- 133. Drew MK, Finch CF. The relationship between training load and injury, illness and soreness: a systematic and literature review. *Sports Medicine*. 2016;46(6):861-883. doi: 10.1007/s40279-015-0459-8
- Juliff LE, Halson SL, Bonetti DL, et al. Influence of contrast shower and water immersion on recovery in elite netballers. *J Strength Cond Res.* 2014;28(8):2353-2358. doi: 10.1519/jsc.00000000000417
- 135. Juliff LE, Halson SL, Hebert JJ, et al. Longer sleep durations are positively associated with finishing place during a national multiday netball competition. *J Strength Cond Res.* 2018;32(1):189-194. doi: 10.1519/jsc.000000000001793

- 136. Quarrie KL, Gianotti SM, Hopkins WG, et al. Effect of nationwide injury prevention programme on serious spinal injuries in New Zealand rugby union: ecological study. *Br Med J*. 2007;334(7604):1150. doi: 10.1136/bmj.39185.605914.AE
- 137. Junge A, Lamprecht M, Stamm H, et al. Countrywide campaign to prevent soccer injuries in Swiss amateur players. *Am J Sports Med.* 2011;39(1):57-63. doi: 10.1177/0363546510377424
- 138. Smyth E, Waddington G, Witchalls J, et al. Does ankle tape improve proprioception acuity immediately after application and following a netball session? A randomised controlled trial. *Physical Therapy in Sport.* 2021;48:20-25. doi: 10.1016/j.ptsp.2020.12.010
- 139. Garrick JG, Requa RK. Role of external support in the prevention of ankle sprains. *Med Sci Sports.* 1973;5(3):200-203. doi:
- 140. Moiler K, Hall T, Robinson K. The role of fibular tape in the prevention of ankle injury in basketball: a pilot study. *Journal of Orthopaedic & Sports Physical Therapy*. 2006;36(9):661-668. doi:
- 141. Brennan F, Jr., Jackson CR, Olsen C, et al. Blisters on the battlefield: the prevalence of and factors associated with foot friction blisters during Operation Iraqi Freedom I. *Military Medicine*. 2012;177(2):157-162. doi: 10.7205/milmed-d-11-00325
- 142. Hashmi F, Richards B, Forghany S, et al. The formation of friction blisters on the foot: the development of a laboratory-based blister creation model. *Skin Research and Technology*. 2012;19(1):e479-e489. doi: 10.1111/j.1600-0846.2012.00669.x
- 143. Bolling C, van Mechelen W, Pasman HR, et al. Context matters: revisiting the first step of the 'sequence of prevention' of sports injuries. *Sports Medicine*. 2018;48(10):2227-2234. doi: 10.1007/s40279-018-0953-x
- 144. Verhagen E, Voogt N, Bruinsma A, et al. A knowledge transfer scheme to bridge the gap between science and practice: an integration of existing research frameworks into a tool for practice. *Br J Sports Med.* 2014;48(8):698-701. doi: 10.1136/bjsports-2013-092241
- 145. Australian Government. *National Innovation and Science Agenda Report*. Canberra: Department of Industry, Science, Energy and Resources, 2015.
- 146. Gurzawska A, Mäkinen M, Brey P. Implementation of Responsible Research and Innovation (RRI) practices in industry: providing the right incentives. *Sustainability*. 2017;9(10):1759. doi: 10.3390/su9101759
- 147. Australian Government. Get help with your research needs to take forward an idea or innovation. Australian Government. 2021. Accessed August 13, 2021. https://business.gov.au/Grants-and-Programs/Innovation-Connections.
- 148. McCunn R, Gibson NV, Fullagar HHK, et al. Professional youth football academy injury data: collection procedures, perceived value, and use. *Sci Med Footb*. 2018;2(2):141-148. doi: 10.1080/24733938.2017.1410564
- 149. Linton L, Valentin S. Running coaches and running group leaders' engagement with, and beliefs and perceived barriers to prehabilitation and injury prevention strategies for runners. *Physical Therapy in Sport.* 2020;46:54-62. doi: 10.1016/j.ptsp.2020.08.004
- 150. Mooney M, Charlton PC, Soltanzadeh S, et al. Who 'owns' the injury or illness? Who 'owns' performance? Applying systems thinking to integrate health and performance in elite sport. *British Journal of Sports Medicine*. 2017;51(14):1054. doi: 10.1136/bjsports-2016-096649
- 151. Ekstrand J. Keeping your top players on the pitch: the key to football medicine at a professional level. *British Journal of Sports Medicine*. 2013;47(12):723. doi: 10.1136/bjsports-2013-092771
- 152. Baugh CM, Meehan WP, McGuire TG, et al. Staffing, financial, and administrative oversight models and rates of injury in collegiate athletes. *J Athl Train*. 2020;55(6):580-586. doi: 10.4085/1062-6050-0517.19
- 153. Schwieger D, Ladwig C. Reaching and retaining the next generation: Adapting to the expectations of Gen Z in the classroom. *Information Systems Education Journal*. 2018;16(3):45-54. <u>https://isedj.org/2018-16/n3/ISEDJv16n3p45.html</u>.
- 154. Harigopal K. Management of Organizational Change: Leveraging Transformation. Sage; 2006.

- 155. Heward S, Hutchins C, Keleher H. Organizational change—key to capacity building and effective health promotion. *Health Promotion International*. 2007;22(2):170-178. doi: 10.1093/heapro/dam011
- 156. Stouten J, Rousseau D, Cremer D. Successful organizational change: integrating the management practice and scholarly literatures. *Acad Manag Ann.* 2018;12(2). doi: 10.5465/annals.2016.0095
- 157. Casey MM, Payne WR, Eime RM. Organisational readiness and capacity building strategies of sporting organisations to promote health. *Sport Manage Rev.* 2012;15(1):109-124. doi: 10.1016/j.smr.2011.01.001
- 158. Alänge S, Steiber A. The board's role in sustaining major organizational change: An empirical analysis of three change programs. *Int J Qual Serv Sci.* 2009;1:280-293. doi: 10.1108/17566690911004212
- 159. May C. Structure of Australian Sport. Clearinghouse for Sport. 2021. Accessed August 19, 2021. <u>https://www.clearinghouseforsport.gov.au/kb/structure-of-australian-sport.</u>
- 160. Australian Government. *Governance Reform in Sport: Discussion Paper*. Bruce, ACT: Australian Sports Commission, 2016.
- 161. Hanson DW, Finch CF, Allegrante JP, et al. Closing the gap between injury prevention research and community safety promotion practice: revisiting the public health model. *Public Health Reports*. 2012;127(2):147-155. doi: 10.1177/003335491212700203
- 162. Smyth E, Appaneal R, Drew M, et al. Identifying the challenges to implementing a neuromuscular warm-up in pre-elite netball. *Int J Sports Sci Coach.* 2021;16(4):913-924. doi: 10.1177/1747954121998462
- 163. Wagstaff CRD, Gilmore S, Thelwell RC. Sport medicine and sport science practitioners' experiences of organizational change. *Scandinavian Journal of Medicine & Science in Sports*. 2015;25(5):685-698. doi: 10.1111/sms.12340
- 164. Coutts AJ. Working fast and working slow: the benefits of embedding research in high performance sport. *International journal of sports physiology and performance*. 2016;11(1):1-2. doi: 10.1123/IJSPP.2015-0781
- 165. Orchard J. Dr John Orchard AM: Sports and Exercise Physician. 2021. Accessed July 30, 2021. <u>www.johnorchard.com</u>.
- 166. Finch CF, Orchard JW, Twomey DM, et al. Coding OSICS sports injury diagnoses in epidemiological studies: does the background of the coder matter? *British Journal of Sports Medicine*. 2014;48(7):552-556. doi: 10.1136/bjsports-2012-091219
- 167. Wik EH, Materne O, Chamari K, et al. Involving research-invested clinicians in data collection affects injury incidence in youth football. *Scandinavian Journal of Medicine & Science in Sports*. 2019;29(7):1031-1039. doi: 10.1111/sms.13427
- 168. Van Tiggelen D, Wickes S, Stevens V, et al. Effective prevention of sports injuries: a model integrating efficacy, efficiency, compliance and risk-taking behaviour. *British journal of sports medicine*. 2008;42(8):648-652. doi: 10.1136/bjsm.2008.046441
- 169. O'Brien J, Finch CF, Pruna R, et al. A new model for injury prevention in team sports: the Team-sport Injury Prevention (TIP) cycle. *Sci Med Footb*. 2019;3(1):77-80. doi: 10.1080/24733938.2018.1512752
- 170. Fry MD, Hogue CM, Iwasaki S, et al. The Relationship Between the Perceived Motivational Climate in Elite Collegiate Sport and Athlete Psychological Coping Skills. *Journal of Clinical Sport Psychology*. 2021;15(4):334-350. doi: 10.1123/jcsp.2020-0002
- 171. Meyers MC, Stewart CC, Laurent CM, et al. Coping skills of olympic developmental soccer athletes. *Int J Sports Med.* 2008;29(12):987-993. doi: 10.1055/s-2008-1038679
- 172. Ekegren CL, Donaldson A, Gabbe BJ, et al. Implementing injury surveillance systems alongside injury prevention programs: evaluation of an online surveillance system in a community setting. *Inj Epidemiol*. 2014;1(1):19. doi: 10.1186/s40621-014-0019-y
- 173. Herzberg SD, Motu'apuaka ML, Lambert W, et al. The effect of menstrual cycle and contraceptives on ACL injuries and laxity: a systematic review and meta-analysis. *Orthop J Sports Med.* 2017;5(7):2325967117718781. doi: 10.1177/2325967117718781

- 174. Mountjoy M, Sundgot-Borgen JK, Burke LM, et al. IOC consensus statement on relative energy deficiency in sport (RED-S): 2018 update. *British Journal of Sports Medicine*. 2018;52(11):687. doi: 10.1136/bjsports-2018-099193
- 175. Martin D, Sale C, Cooper SB, et al. Period prevalence and perceived side effects of hormonal contraceptive use and the menstrual cycle in elite athletes. *Int J Sports Physiol Perform*. 2018;13(7):926-932. doi: 10.1123/ijspp.2017-0330
- 176. Ageberg E, Bunke S, Nilsen P, et al. Planning injury prevention training for youth handball players: application of the generalisable six-step intervention development process. *Injury Prevention.* 2020;26(2):164. doi: 10.1136/injuryprev-2019-043468
- 177. Padua DA, Frank B, Donaldson A, et al. Seven steps for developing and implementing a preventive training program: lessons learned from JUMP-ACL and beyond. *Clin Sports Med*. 2014;33(4):615-632. doi: 10.1016/j.csm.2014.06.012
- 178. Haynes RB, Devereaux PJ, Guyatt GH. Clinical expertise in the era of evidence-based medicine and patient choice. *BMJ Evid Based Med.* 2002;7(2):36. doi: 10.1136/ebm.7.2.36
- 179. Saunders N. *Characteristics of the female landing pattern*. [PhD Thesis]. Ballarat: University of Ballarat; 2006.
- 180. Netball Australia. The KNEE program. Netball Australia. 2019. Accessed January 30, 2019. https://knee.netball.com.au/.
- 181. Netball New Zealand. NetballSmart. Netball New Zealand. 2021. Accessed July 28, 2021. www.netballsmart.co.nz.
- 182. Accident Compensation Corporation. SportSmart. Accident Compensation Corporation. 2021. Accessed July 29, 2021. <u>www.accsportsmart.co.nz</u>.
- 183. Gilchrist J, Mandelbaum BR, Melancon H, et al. A randomized controlled trial to prevent non contact anterior cruciate ligament injury in female collegiate soccer players. *American Journal of Sports Medicine*. 2008;36(8):1476-1483. doi: 10.1177/0363546508318188
- 184. Junge A, Rosch D, Peterson L, et al. Prevention of soccer injuries: a prospective intervention study in youth amateur players. Am J Sports Med. 2002;30(5):652-659. doi: 10.1177/03635465020300050401
- 185. Owoeye OB, Akinbo SR, Tella BA, et al. Efficacy of the FIFA 11+ warm-up programme in male youth football: a cluster randomised controlled trial. *J Sports Sci Med.* 2014;13(2):321-328. Accessed May 22, 2017. <u>https://pubmed.ncbi.nlm.nih.gov/24790486/</u>.
- 186. Scase E, Cook J, Makdissi M, et al. Teaching landing skills in elite junior Australian football: evaluation of an injury prevention strategy. *British Journal of Sports Medicine*. 2006;40(10):834-838. doi: 10.1136/bjsm.2006.025692
- 187. Silvers-Granelli HJ, Bizzini M, Arundale A, et al. Does the FIFA 11+ injury prevention program reduce the incidence of ACL injury in male soccer players? *Clin Orthop Relat Res.* 2017;475(10):2447-2455. doi: 10.1007/s11999-017-5342-5
- 188. Silvers-Granelli H, Mandelbaum B, Adeniji O, et al. Efficacy of the FIFA 11+ injury prevention program in the collegiate male soccer player. *Am J Sports Med.* 2015;43(11):2628-2637. doi: 10.1177/0363546515602009
- 189. Zouita S, Zouita AB, Kebsi W, et al. Strength training reduces injury rate in elite young soccer players during one season. J Strength Cond Res. 2016;30(5):1295-1307. doi: 10.1519/jsc.000000000000920
- 190. Ivarsson A, Johnson U, Andersen MB, et al. It pays to pay attention: a mindfulness-based program for injury prevention with soccer players. *Journal of Applied Sport Psychology*. 2015;27(3):319-334. doi: 10.1080/10413200.2015.1008072
- 191. Johnson U, Ekengren J, Andersen MB. Injury prevention in Sweden: helping soccer players at risk. *Journal of Sport & Exercise Psychology*. 2005;27(1):32. doi: 10.1123/jsep.27.1.32
- 192. Barbic D, Pater J, Brison RJ. Comparison of mouth guard designs and concussion prevention in contact sports: a multicenter randomized controlled trial. *Clinical Journal of Sport Medicine*. 2005;15(5):294-298. doi: 10.1097/01.jsm.0000171883.74056.21
- 193. Lewis RM, Redzic M, Thomas DT. The effects of season-long vitamin D supplementation on collegiate swimmers and divers. *Int J Sport Nutr Exerc Metab.* 2013;23(5):431-440. doi: 10.1123/ijsnem.23.5.431

- 194. Emery C, Palacios-Derflingher L, Black AM, et al. Does disallowing body checking in nonelite 13- to 14-year-old ice hockey leagues reduce rates of injury and concussion? A cohort study in two Canadian provinces. *British Journal of Sports Medicine*. 2020;54(7):414. doi: 10.1136/bjsports-2019-101092
- 195. Janda DH, Bir C, Kedroske B. A comparison of standard vs. breakaway bases: an analysis of a preventative intervention for softball and baseball foot and ankle injuries. *Foot & Ankle International*. 2001;22(10):810-816. doi: 10.1177/107110070102201006
- 196. Bhide A, Shah PS, Acharya G. A simplified guide to randomized controlled trials. *Acta Obstet Gynecol Scand.* 2018;97(4):380-387. doi: 10.1111/aogs.13309
- 197. Coutts AJ. Challenges in developing evidence-based practice in high-performance sport. *International journal of sports physiology and performance*. 2017;12(6):717-718. doi: 10.1123/IJSPP.2017-0455
- 198. Finch C, Braham R, McIntosh A, et al. Should football players wear custom fitted mouthguards? Results from a group randomised controlled trial. *Injury Prevention* (1353-8047). 2005;11(4):242-246. doi: 10.1136/ip.2004.006882
- 199. Skorski S, Hecksteden A. Coping With the "Small Sample–Small Relevant Effects" Dilemma in Elite Sport Research. *International Journal of Sports Physiology and Performance*. 2021;16(11):1559-1560. doi: 10.1123/jjspp.2021-0467
- 200. Drew MK, Vlahovich N, Hughes D, et al. A multifactorial evaluation of illness risk factors in athletes preparing for the Summer Olympic Games. *J Sci Med Sport*. 2017;20(8):745-750. doi: 10.1016/j.jsams.2017.02.010
- 201. Wollin M, Thorborg K, Drew M, et al. A novel hamstring strain injury prevention system: post-match strength testing for secondary prevention in football. *British journal of sports medicine*. 2020;54(9):498-499. doi: 10.1136/bjsports-2019-100707
- 202. Bishop D. An applied research model for the sport sciences. *Sports Med.* 2008;38(3):253-263. doi: 10.2165/00007256-200838030-00005
- 203. Buchheit M. Houston, we still have a problem. *International journal of sports physiology and performance*. 2017;12(8):1111-1114. doi: 10.1123/ijspp.2017-0422
- 204. Vanwanseele B, Stuelcken M, Greene A, et al. The effect of external ankle support on knee and ankle joint movement and loading in netball players. *Journal of Science and Medicine in Sport.* 2014;17(5):511-515. doi: 10.1016/j.jsams.2013.07.009
- 205. Hopper DM, McNair P, Elliott BC. Landing in netball: effects of taping and bracing the ankle. *British Journal of Sports Medicine*. 1999;33(6):409. doi: 10.1136/bjsm.33.6.409
- 206. Hopper AJ, Haff EE, Joyce C, et al. Neuromuscular training improves lower extremity biomechanics associated with knee injury during landing in 11-13 year old female netball athletes: a randomized control study. *Front Physiol.* 2017;8:883. doi: 10.3389/fphys.2017.00883
- 207. Terblanche E, Venter Ranel E. The effect of backward training on the speed, agility and power of netball players. *S Afr J Res Sport Phys Educ Recreation*. 2009;31(2):135-145. doi: 10.10520/EJC108898
- 208. Thomas C, Comfort P, Jones PA, et al. A comparison of isometric midthigh-pull strength, vertical jump, sprint speed, and change-of-direction speed in academy netball players. *International Journal of Sports Physiology and Performance*. 2017;12(7):916-921. doi: 10.1123/ijspp.2016-0317
- 209. McKenzie CR, Whatman C, Brughelli M, et al. The effect of the NetballSmart Dynamic Warm-up on physical performance in youth netball players. *Physical Therapy in Sport*. 2019;37:91-98. doi: 10.1016/j.ptsp.2019.03.005
- 210. Westfall JM, Mold J, Fagnan L. Practice-based research—"Blue Highways" on the NIH roadmap. *JAMA*. 2007;297(4):403-406. doi: 10.1001/jama.297.4.403
- 211. Trochim W. Translation won't happen without dissemination and implementation: Some measurement and evaluation issues. Paper presented at: 3rd Annual Conference on the Science of Dissemination and Implementation; 2010; Bethesda, USA.
- 212. Green LW, Ottoson JM, Garcia Č, et al. Diffusion theory and knowledge dissemination, utilization, and integration in public health. *Annual review of public health*. 2009;30:151-174. doi: 10.1146/annurev.publhealth.031308.100049

- 213. Donaldson A, Callaghan A, Bizzini M, et al. A concept mapping approach to identifying the barriers to implementing an evidence-based sports injury prevention programme. *Injury Prevention.* 2018;0:1-8. doi: 10.1136/injuryprev-2017-042639
- 214. Brown JC, Gardner-Lubbe S, Lambert MI, et al. Coach-directed education is associated with injury-prevention behaviour in players: an ecological cross-sectional study. *British Journal of Sports Medicine*. 2018;52(15):989. doi: 10.1136/bjsports-2016-096757
- 215. Hulme A, Salmon PM, Nielsen RO, et al. From control to causation: Validating a 'complex systems model' of running-related injury development and prevention. *Appl Ergon*. 2017;65:345-354. doi: 10.1016/j.apergo.2017.07.005
- 216. Moher D, Liberati A, Tetzlaff J, et al. Preferred reporting items for systematic reviews and meta-analyses: the PRISMA statement. *Int J Surg.* 2010;8(5):336-341. doi: 10.1016/j.ijsu.2010.02.007
- 217. OCEBM Levels of Evidence Working Group. The Oxford 2011 levels of evidence. Oxford Centre for Evidence-Based Medicine. 2011. Accessed August 31, 2020. http://www.cebm.net/index.aspx?o=5653.
- 218. Han J, Waddington G, Adams R, et al. Assessing proprioception: a critical review of methods. *J Sport Health Sci.* 2016;5(1):80-90. doi: 10.1016/j.jshs.2014.10.004
- 219. Lewkowicz DJ. The concept of ecological validity: what are its limitations and is it bad to be invalid? *Infancy*. 2001;2(4):437-450. doi: 10.1207/S15327078IN0204_03
- 220. Witchalls JB, Waddington G, Adams R, et al. Chronic ankle instability affects learning rate during repeated proprioception testing. *Physical Therapy in Sport*. 2014;15(2):106-111. doi: 10.1016/j.ptsp.2013.04.002
- 221. Steinberg N, Waddington G, Adams R, et al. Can textured insoles improve ankle proprioception and performance in dancers? *J Sports Sci.* 2016;34(15):1430-1437. doi: 10.1080/02640414.2015.1117120
- 222. Rosas SR, Kane M. Quality and rigor of the concept mapping methodology: a pooled study analysis. *Eval Program Plann.* 2012;35(2):236-245. doi: 10.1016/j.evalprogplan.2011.10.003

Appendices

Appendix 1: Human Ethics Committee approval for Chapter 2

Anstralian Sports Commission						
	The superior	-	and man			
0.	MS ENIT SHIYEN	SL: DF	MICK DOWN			
FRIDM:	Ns Helene Rushby		Sector Sector			
SUBJECT:	Approval from AIS Ethics Committee	DATE	16" April 2013			
On the 10° lited "Inju National C should not	April 2018, the AIS Ethics Committee ga any profile of 17/U and 19/U netball a Championships" The Committee saw ho proceed.	ve considerat thietes duri ethical reason	lori to your submissio ng and after 1 why your project			
The approv	al number for this project is; 20160404					
It is a required to a service of the	irement of the AIS Ethics Committee that esearchers involved in the study of Ethics of that approval. You are also required to y (via the Secretary) of:	the Principal Committee a advise the Et	Researcher (you) pproval and ony hics Committee			
	Any proposed changes t Any adverse events that	o the research may occur,	h design,			
Researcher secretary o contained i	s are required to submit annual status of If the AIS Ethics Committee, Details of sta n the "Guidelines" for ethics submissions.	eports and t	inal reports to the guinements are			
Please not	the approval for this submission expires an extension will need to be sought.	on the 31 th D	ecember 2020 after			
(0.2) 6214	any questions regarding this matter, ple 1577.	ase don't hes	itate to contact me or			
Sincensiy						
Helene Rus	Por					
ALS Ethics	Committee					

Appendix 2: Information to Participants for Chapter 2

INFORMATION TO PARTICIPANTS

Research Title: Injury profile of 17/U and 19/U netball athletes during and after National Championships

Principal Researcher:

Erin Smyth (0420512039, erin.smyth@ausport.gov.au)

We would like to invite you to participate in this research project. Choosing not to take part will not disadvantage you in any way. Before you decide whether you want to take part, it is important for you to understand why the research is being done and what your participation will involve. Please take time to read the following information carefully and discuss it with others if you wish. Ask us if there is anything that is not clear or if you would like more information.

Aim:

The aim of this research project is to investigate injury and illness incidence in 17/U and 19/U Netball athletes during and one month following the 2018 National Championships.

Benefits:

This study will provide valuable insights into the injury profile of 17/U and 19/U pre-elite Netball athletes to help guide future injury prevention initiatives.

What is involved?

The study will be conducted over a 5 week period (16 April 2018 - 19 May 2018) and will consist of the following:

Online survey

A survey will be accessible through the Athlete Management System (AMS) and consists of four questions that ask whether you have experienced any health problems in the last week. You are asked to provide the most accurate responses that you can.

AMS data entry

Your injury records and netball participation will be recorded on AMS by your team Physiotherapist and Netball Australia selectors. This data will be de-identified to protect your privacy.

Who are we recruiting?

We are recruiting all netball athletes participating in the 2018 17/U and 19/U National Championships.

Adverse effects and withdrawal:

We do not anticipate any adverse effects and you may withdraw at any time without penalty.

Confidentiality:

The confidentiality of your data will be respected at all times. Only researchers directly involved in this project will see your AMS data which will be kept for minimum of 5 years. Data is stored electronically in the AMS and any hard copies will be stored in a locked filing cabinet. Participants' names will not be associated with any aspect of this study; this will ensure anonymity during research and publication. Data will be collated so that an individual's information will not be able to be identified. The AMS data utilised in this study may also be used in future studies. This research will be used to inform the design of injury and illness prevention programs for netball athletes in the future. Results will be disseminated in the form of publications and be included in Erin Smyth's PhD thesis. Follow up support will be provided by the student researcher Erin Smyth between Monday-Thursday, between the hours of 9am and 5pm.

Ethics approval:

This study has been approved by the Australian Institute of Sport Ethics Committee (Approval Number: 20180404). If you, the participant, have any concerns you can contact the secretary of the AIS Ethics Committee on 02 6214 1577.

Further information:

If you have any questions, or you would like further information regarding the project titled: *Injury profile of 17/U and 19/U netball athletes during and after National Championships*, please contact the Principal Researcher, **Erin Smyth** of the **Australian Institute of Sport** and **University of Canberra: PH: +61 420512039 EMAIL: erin.smyth@ausport.gov.au**

Appendix 3: Information for Parent/Guardian of Minor for Chapter 2

INFORMATION FOR PARENT/GUARDIAN OF MINOR (UNDER 18 YEARS) PARTICIPANTS

Research Title: Injury profile of 17/U and 19/U netball athletes during and after National Championships

Principal Researcher:

Erin Smyth (0420512039, erin.smyth@ausport.gov.au)

We would like to invite your child to participate in this research project. Choosing not to take part will not disadvantage your child in any way. Before you decide whether you want your child to take part, it is important for you to understand why the research is being done and what their participation will involve. Please take time to read the following information carefully and discuss it with others if you wish. Ask us if there is anything that is not clear or if you would like more information.

Aim:

The aim of this research project is to investigate injury and illness incidence in 17/U and 19/U Netball athletes during and one month following the 2018 National Championships.

Benefits:

This study will provide valuable insights into the injury profile of 17/U and 19/U pre-elite Netball athletes to help guide future injury prevention initiatives.

What is involved?

The study will be conducted over a 5 week period (16 April 2018 – 19 May 2018) and will consist of the following:

Online survey

A survey will be accessible through the Athlete Management System (AMS) and consists of four questions that ask whether your child has experienced any health problems in the last week. Your child will be asked to provide the most accurate responses that they can. They will be asked to complete the questionnaire four times: 16/4/18, 22/4/18, 29/4/18 and 20/5/18.

AMS data entry

Your child's injury records and netball participation will be recorded on AMS by their team Physiotherapist and Netball Australia selectors. This data will be de-identified to protect their privacy.

Who are we recruiting?

We are recruiting all netball athletes participating in the 2018 17/U and 19/U National Championships.

Adverse effects and withdrawal:

We do not anticipate any adverse effects and your child may withdraw at any time without penalty.

Confidentiality:

The confidentiality of your child's data will be respected at all times. Only researchers directly involved in this project will see their AMS data which will be kept for minimum of 5 years. Data is stored electronically in the AMS and any hard copies will be stored in a locked filing cabinet. Participants' names will not be associated with any aspect of this study; this will ensure anonymity during research and publication. Data will be collated so that an individual's information will not be able to be identified. The AMS data utilised in this study may also be used in future studies. This research will be used to inform the design of injury and illness prevention programs for netball athletes in the future. Results will be disseminated in the form of publications and be included in Erin Smyth's PhD thesis. Follow up support will be provided by the student researcher Erin Smyth between Monday-Thursday, between the hours of 9am and 5pm.

Consent:

Your child will be asked to complete a questionnaire on their smartphone. Prior to this they will be asked to provide your consent, name and contact email. Can you please communicate with your child prior to 16th April 2018 whether you give your consent for them to participate in this research project.

Ethics approval:

This study has been approved by the Australian Institute of Sport Ethics Committee (Approval Number: 20180404). If you, the participant, have any concerns you can contact the secretary of the AIS Ethics Committee on 02 6214 1577.

Further information:

If you have any questions, or you would like further information regarding the project titled: *Injury profile of 17/U and 19/U netball athletes during and after National Championships*, please contact the Principal Researcher, **Erin Smyth** of the **Australian Institute of Sport** and **University of Canberra**:

PH: +61 420512039

EMAIL: erin.smyth@ausport.gov.au

Appendix 4: Informed Consent Form for Chapter 2

Injury profile of 17/U and 19/U netball athletes during and after 2018 National Championships

Principal Investigators: Ms Erin Smyth (PhD Scholar, AIS/UC), Dr Michael Drew (AIS), Prof Gordon Waddington (AIS/UC), Dr Juanita Weissensteiner (AIS), Dr Philip Newman (UC), Dr Laura Juliff (Netball Australia), Ms Alanna Antcliff (Netball Australia)

You are invited to take part in this research project.

Please read the **Participant Information Form** carefully as this will tell you about the research project and explain what is involved. This will help you decide if you want to continue and take part.

Ask questions about anything that you don't understand or want to know more about. Before deciding whether or not to take part, you might want to talk about it with a relative, friend or healthcare worker.

Participation

- Participation in this study is entirely voluntary.
- You're not obliged to participate and if you do, you can withdraw at any time without penalty or prejudice.
- To participate, we would like you to complete the online questionnaire: 'Questionnaire on Health Problems'.
- Your participation, personal details and results will be strictly confidential and only the Principal Investigators above will have access to the information.

By ticking the 'I ACCEPT' or 'My Parent or Guardian ACCEPTS' option below you are telling us that you or your Parent or Guardian:

- Understand what you have read.
- Consent to take part in the research project.
- Consent to participate in the research processes that are described.
- Consent to use your personal and health information as described.
- Understand that you are free to not answer specific items or questions.
- Understand that any data or answers to questions will remain confidential with regard to your identity.
- Certify to the best of your knowledge and belief, you have no physical or mental illness that would increase the risk of participating in this project.
- Are participating in this project of your own free will and have not been coerced in any way to participate
- □ I ACCEPT (If you do not accept, please exit the browser)

Minors (Please complete if under 18 years)

□ My Parent or Guardian ACCEPTS (If they do not accept, please exit the browser) Parent/Guardian full name:

Parent/Guardian email address: _____

Appendix 5: Human Ethics Committee approval for Chapter 4

Dear Erin

The Human Research Ethics Committee has considered your application to conduct research with human subjects for the project "20181527 - Does ankle tape improve proprioception acuity immediately after application and following netball training?".

The Committee made the following evaluation: Approved

The approval is valid until: 05/10/2018

The following general conditions apply to your approval. These requirements are determined by University policy and the *National Statement on Ethical Conduct in Human Research* (National Health and Medical Research Council, 2007).

Monitoring

You must assist the Committee to monitor the conduct of approved research by completing project review forms, and in the case of extended research, at least annually during the approval period.

Reporting Adverse Events

You must report any unexpected adverse events or complications that occur anytime during the conduct of the research study or during the follow up period after the research. Please refer these matters promptly to the HREC. Failure to do so may result in the withdrawal of the Ethics approval.

Discontinuation of Research

You must inform the Committee, giving reasons, if the research is not conducted or is discontinued before the expected date of completion.

Extension of Approval

If your project will not be complete by the expiry date stated above, you must apply for extension of approval. This must be done before current approval expires.

Retention and Storage of Data

University policy states that all research data must be stored securely, on University premises, for a minimum of five years. You must ensure that all records are transferred to the University when the project is complete.

Contact Details and Notification of Changes

All email contact should use the UC email address. You should advise the Committee of any change of address during or soon after the approval period including, if appropriate, email address(es).

Please do not hesitate to contact us via email <u>humanethicscommittee@canberra.edu.au</u> if you require any further information.

All the best,

Hendryk Flaegel Research Ethics & Integrity Research Services University of Canberra 29/10/2018

Appendix 6: Information to Participants for Chapter 4







PARTICIPANT INFORMATION SHEET

Research Title: Does ankle tape improve proprioception acuity immediately after application and following netball training?

Principal Researcher:

Erin Smyth (0420512039, u3098942@uni.canberra.edu.au)

We would like to invite you to participate in this research project. Choosing not to take part will not disadvantage you in any way. Before you decide whether you want to take part, it is important for you to understand why the research is being done and what your participation will involve. Please take time to read the following information carefully and discuss it with others if you wish. Ask us if there is anything that is not clear or if you would like more information.

Aim:

The aim of this research project is to investigate whether ankle tape improves proprioception (position and movement sense) of the ankle.

Benefits:

The benefit will be the creation of current evidence to support or refute the common practice of taping/bracing to prevent ankle injuries.

What is involved?

- 1. You will undertake movement sensitivity (proprioception) testing at the ankle in a standing position using an AMEDA device. The AMEDA device consists of a footplate on a platform that can tilt to 5 possible positions resulting in ankle inversion movements between 10.5 and 14.5 degrees from the horizontal. You will be familiarised with the 5 positions in an introductory sequence where the different positions are set in order from smallest (position 1) to largest (position 5). This sequence is repeated 3 times. Following familiarization, the footplate, from its horizontal start position, will be moved to one of the 5 positions which you will then be asked to identify. This will be repeated over 50 trials, each time returning to the start position, with each position presented 10 times in random order. The test is semi-automated and takes approximately 6 minutes to complete.
- 2. Then depending on which group you've been assigned to, you will then have your ankles taped by a Sport and Exercise Physiotherapist or you will self-tape if that is your normal practice.
- 3. Proprioception will then be tested again, repeating the 50 trials
- 4. A Catapult local monitoring system tracker will then be fitted and you will participate in a normal training session or match.
- 5. Immediately after the training session you will repeat the 50 trials regardless of whether you tape, brace or neither.
- 6. The tape will then be removed and you will repeat the 50 trials again.

Who are we recruiting?

Pre-elite netball players (17/U & 19/U state representatives and higher)

Withdrawal:

You may withdraw at any time without penalty.

Risks:

The risks to you associated with participation in this study are minimal and are consistent with normal activities undertaken in everyday life in a clinical setting. For example, tripping on a carpet edge. To minimise risk we will brief you on standard procedures for conduct and safety within the research laboratory space.

Confidentiality:

Your results will be kept as confidential as is possible by law. All data will be kept in the possession of the investigators. If the results of the study are published in a scientific journal, your identity will not be revealed. If you wish to withdraw from the study prior to publication of results your data will be removed from the study data. Participants will not be referred to by name during research reports or study discussions. All hard copy records will be stored in a locked filing cabinet in a private office. All computer records are restricted by password. All study data will be deleted at five years after the publication of results. Results will be disseminated in the form of publications and be included in Erin Smyth's PhD thesis.

Ethics approval:

This study has been approved by the University of Canberra Committee for Ethics in Human Research (Approval Number:20181527). If you, the participant, have any concerns you can contact the Human Ethics Committee: **E:** humanethicscommittee@canberra.edu.au; **T:** 02 6201 5220

Privacy statement

The conduct of this research involves the collection, access and/or use of your identified personal information. The information collected is confidential and will not be disclosed to third parties without your consent, except to meet government, legal or other regulatory authority requirements. A de-identified copy of this data may be used for other research purposes. However, your anonymity will at all times be safeguarded.

Further information:

If you have any questions, or you would like further information regarding the project titled: *Does ankle tape improve proprioception acuity immediately after application and following netball training*? please contact the Principal Researcher, **Erin Smyth** of the **Australian Institute of Sport** and **University of Canberra: PH:** +61 420512039

EMAIL: u3098942@uni.canberra.edu.au

Appendix 7: Information for Parent/Guardian of Minor for Chapter 4





PARENT/GUARDIAN OF MINOR (UNDER 18 YEARS) PARTICIPANTS INFORMATION SHEET

Research Title: Does ankle tape improve proprioception acuity immediately after application and following netball training?

Principal Researcher:

Erin Smyth (0420512039, u3098942@uni.canberra.edu.au)

We would like to invite your child to participate in this research project. Choosing not to take part will not disadvantage your child in any way. Before you decide whether you want your child to take part, it is important for you to understand why the research is being done and what their participation will involve. Please take time to read the following information carefully and discuss it with others if you wish. Ask us if there is anything that is not clear or if you would like more information.

Aim:

The aim of this research project is to investigate whether ankle tape improves proprioception (position and movement sense) of the ankle.

Benefits:

The benefit will be the creation of current evidence to support or refute the common practice of taping/bracing to prevent ankle injuries.

What is involved?

Your child will undertake movement sensitivity (proprioception) testing at the ankle in a standing position using an AMEDA device. The AMEDA device consists of a footplate on a platform that can tilt to 5 possible positions resulting in ankle inversion movements between 10.5 and 14.5 degrees from the horizontal. Your child will be familiarised with the 5 positions in an introductory sequence where the different positions are set in order from smallest (position 1) to largest (position 5). This sequence is repeated 3 times. Following familiarization, the footplate, from its horizontal start position, will be moved to one of the 5 positions which your child will then be asked to identify. This will be repeated over 50 trials, each time returning to the start position, with each position presented 10 times in random order. The test is semi-automated and takes approximately 6 minutes to complete.

Then depending on which group your child has been assigned to, they will then have their ankles taped by a Sport and Exercise Physiotherapist or they will self-tape if that is their normal practice.

Proprioception will then be tested again, repeating the 50 trials

A Catapult local monitoring system tracker will then be fitted, and your child will participate in a normal training session or match.

Immediately after the training session, your child will repeat the 50 trials regardless of whether they tape, brace or neither.

The tape will then be removed, and your child will repeat the 50 trials.

Who are we recruiting?

Pre-elite netball players (17/U & 19/U state representatives and higher)

Withdrawal:

Your child may withdraw at any time without penalty.

Risks:

The risks to your child associated with participation in this study are minimal and are consistent with normal activities undertaken in everyday life in a clinical setting. For example, tripping on a carpet edge. To minimise risk, we will brief your child on standard procedures for conduct and safety within the research laboratory space.

Confidentiality:

Your child's results will be kept as confidential as is possible by law. All data will be kept in possession of the investigators. If the results of the study are published in a scientific journal, your child's identity will not be revealed. If your child wishes to withdraw from the study prior to publication of results their data will be removed from the study data. Participants will not be referred to by name during research reports or study discussions. All hard copy records will be stored in a locked filing cabinet in a private office. All computer records are restricted by password. All study data will be deleted at five years after the publication of results. Results will be disseminated in the form of publications and be included in Erin Smyth's PhD thesis.

Ethics approval:

This study has been approved by the University of Canberra Committee for Ethics in Human Research (Approval Number: 20181527). If you, the participant, have any concerns you can contact the Human Ethics Committee: **E:** <u>humanethicscommittee@canberra.edu.au</u>; **T:** 02 6201 5220

Privacy statement

The conduct of this research involves the collection, access and/or use of your child's identified personal information. The information collected is confidential and will not be disclosed to third parties without your consent, except to meet government, legal or other regulatory authority requirements. A de-identified copy of this data may be used for other research purposes. However, your child's anonymity will at all times be safeguarded.

Further information:

If you have any questions, or you would like further information regarding the project titled: '*Does ankle tape improve proprioception acuity immediately after application and following netball training*?' please contact the Principal Researcher, Erin Smyth of the Australian Institute of Sport and University of Canberra: PH: +61 420512039

EMAIL: <u>u3098942@uni.canberra.edu.au</u>

Appendix 8: Informed Consent Form for Chapter 4

CONSENT FORM FOR PARTICIPANTS INVOLVED IN RESEARCH

Project:	Does ankle tape improve proprioception acuity immediately after application and following netball training?
Investigators:	Erin Smyth, Prof Gordon Waddington, Dr Phillip Newman, University of Canberra Research Institute for Sport and Exercise. Dr Michael Drew, Athlete Availability, Australian Institute of Sport. Dr Juanita Weissensteiner, NSW Office for Sport.

This is to certify that I, ______ am at least 18 years old and that I am voluntarily giving my consent to participate in the research project named above.

I certify that the investigation and my part in the investigation have been defined and explained to me by a member of the research team and I understand the explanation. A copy of the procedures of this investigation and a description of any risks has been provided to me and discussed with me.

- I have been given an opportunity to ask whatever questions I may have had and all such questions have been answered to my satisfaction.
- I understand that I am free to withdraw consent and to discontinue participation in the project or activity at any time, without disadvantage to myself.
- I understand that I am free to withdraw my data from analysis without disadvantage to myself.
- I understand that any data or answers to questions will remain confidential with regard to my identity.
- I am participating in this project of my own free will and I have not been coerced in any way to participate.

I consent to participating in this research

Signature of participant: _____

Date: ___/___/

Appendix 9: Informed Consent Form for Minors for Chapter 4

CONSENT FORM FOR PARTICIPANTS (MINOR) INVOLVED IN RESEARCH

Project:	Does ankle tape improve proprioception acuity immediately after application and following netball training?
Investigators:	Erin Smyth, Prof Gordon Waddington, Dr Phillip Newman, University of Canberra Research Institute for Sport and Exercise. Dr Michael Drew, Athlete Availability, Australian Institute of Sport. Dr Juanita Weissensteiner, NSW Office for Sport.

This is to certify that I, _______ hereby agree to give permission for my child _______ to participate as a volunteer in the research project named above.

I certify that the investigation and my child's part in the investigation have been defined and explained to me by a member of the research team and I understand the explanation. A copy of the procedures of this investigation and a description of any risks has been provided to me and discussed with me.

- I have been given an opportunity to ask whatever questions my child or myself may have had and all such questions have been answered to my satisfaction.
- I understand that my child is free to withdraw consent and to discontinue participation in the project or activity at any time, without disadvantage.
- I understand that my child is free to withdraw their data from analysis without disadvantage.
- I understand that any data or answers to questions will remain confidential with regard to my child's identity.
- My child is participating in this project of her own free will and has not been coerced in any way to participate.



I consent to my child participating in this research

Signature of Participant:	Date: _	/	/
Signature of Parent or			
Guardian of minor: (under 18 years)	Date:	/	_/
Appendix 10: Human Ethics Committee approval for Chapter 5

Dear Erin

The Human Research Ethics Committee has considered your application to conduct research with human subjects for the project "20181551 - Injury prevention intervention development and implementation.".

The Committee made the following evaluation: Approved

The approval is valid until: 28/02/2019

The following general conditions apply to your approval. These requirements are determined by University policy and the *National Statement on Ethical Conduct in Human Research* (National Health and Medical Research Council, 2007).

Monitoring

You must assist the Committee to monitor the conduct of approved research by completing project review forms, and in the case of extended research, at least annually during the approval period.

Reporting Adverse Events

You must report any unexpected adverse events or complications that occur anytime during the conduct of the research study or during the follow up period after the research. Please refer these matters promptly to the HREC. Failure to do so may result in the withdrawal of the Ethics approval.

Discontinuation of Research

You must inform the Committee, giving reasons, if the research is not conducted or is discontinued before the expected date of completion.

Extension of Approval

If your project will not be complete by the expiry date stated above, you must apply for extension of approval. This must be done before current approval expires.

Retention and Storage of Data

University policy states that all research data must be stored securely, on University premises, for a minimum of five years. You must ensure that all records are transferred to the University when the project is complete.

Contact Details and Notification of Changes

All email contact should use the UC email address. You should advise the Committee of any change of address during or soon after the approval period including, if appropriate, email address(es).

Please do not hesitate to contact us via email <u>humanethicscommittee@canberra.edu.au</u> if you require any further information.

All the best,

Hendryk Flaegel Research Ethics & Integrity Research Services University of Canberra 29/10/2018

Appendix 11: Information to Participants for Chapter 5



PARTICIPANT INFORMATION STATEMENT

Research Title: How are injury prevention interventions developed and implemented successfully to reduce injury rates in pre-elite netball athletes?

Principal Researcher: Erin Smyth (0420512039, u3098942@uni.canberra.edu.au)

We would like to invite you to participate in this research project. Choosing not to take part will not disadvantage you in any way. Before you decide whether you want to take part, it is important for you to understand why the research is being done and what your participation will involve. Please take time to read the following information carefully and discuss it with others if you wish. Ask us if there is anything that is not clear or if you would like more information.

Aim: The aim of this research project is to evaluate the implementation of the KNEE (Knee injury prevention for Netballers to Enhance performance and Extend play) program, among coaches and physiotherapists of pre-elite netball athletes. The KNEE program is an injury prevention and performance enhancing program developed by Netball Australia.

Benefits: This study will make a valuable contribution towards developing and preventing injuries in pre-elite netball athletes.

What is involved?: You are being asked to participate in a research process called concept mapping.

The study will be conducted over 3-4 weeks and consist of three online activities.

<u>Activity 1</u>: brainstorm ideas about what the challenges are for coaches and physiotherapists of pre-elite netball athletes implementing the KNEE program. (30mins)

Activity 2: sort these ideas into groups of related concepts (30mins)

Activity 3: rate the relative importance of these ideas and the feasibility of netball Australia being able to successfully implement strategies to address them (30 mins)

Your participation will be spread over several weeks. After completing Activity 1 it will be two weeks before you are invited to participate in Activities 2 and 3 which you will be asked to do at the same time.

You will be able to participate as an individual at a time convenient to you. You will not need to meet with others somewhere in order to participate.

Participation is voluntary and anonymous. You will not be required to provide your name when you participate in the research. Other research participants will not be able to identify you or link any information or answers to you.

How do I give my consent to participate in this study?: You will be asked to consent to participate in this study at the beginning of Activity 1. This will require you to respond 'accept' on the first page of the online Concept Mapping system indicating you have read and understood this Participant Information Statement and you are happy to participate in the Concept Mapping process.

Adverse effects and withdrawal: We do not anticipate any adverse effects and you may withdraw at any time without penalty.

Confidentiality: The confidentiality of your data will be respected at all times. Only researchers directly involved in this project will see your data which will be kept for minimum of 5 years. Data is stored electronically on a secure site and any hard copies will be stored in a locked filing cabinet. Participants' names will not be associated with any aspect of this study; this will ensure anonymity during research and publication. Data will be collated so that an individual's information will not be able to be identified. Results will be disseminated in the form of publications and be included in Erin Smyth's PhD thesis.

Ethics approval: This study has been approved by the University of Canberra Committee for Ethics in Human Research (Approval Number: 20181551). If you, the participant, have any concerns you can contact the Human Ethics Committee: **E:** humanethicscommittee@canberra.edu.au; **T:** 02 6201 5220.

Further information: If you have any questions, or you would like further information regarding the project titled: '*How are injury prevention interventions developed and implemented successfully to reduce injury rates in pre-elite netball athletes?*', please contact the Principal Researcher, Erin Smyth of the Australian Institute of Sport and University of Canberra: PH: +61 420512039

EMAIL: u3098942@uni.canberra.edu.au

Appendix 12: Informed Consent Form for Chapter 5

Concept mapping consent form

Thank you for taking the time to participate in this Concept Mapping study to identify the barriers to coaches/physiotherapists of 17/U and 19/U state netball teams implementing the KNEE (Knee injury prevention for Netballers to Enhance performance and Extend play) program with the teams they work with.

If you are not familiar with the KNEE program, please go to the <u>KNEE website</u> and familiarise yourself with the program before participating in this project.

You may be asked to offer your input in a variety of ways:

- by providing your ideas
- rating the ideas or sorting them into groups of similar themes
- by providing non-identifying information about yourself

Before participating in this Concept Mapping study, please read the Participant Information Statement (PIS) that was attached to the email you received inviting you to join the project. If you have not already read the PIS you can find a copy <u>here</u>.

It will be assumed that you are over the age of 18 and have read the PIS when you 'accept' to participate in this research below and start Activity 1.

Appendix 13: Human Ethics Committee approval for Chapter 6



TO: Erin Smyth DATE: 27st March 2019

FROM: Tim Kelly (AJS Ethics Committee Secretary)

SUBJECT: Minor Variation to 'Injury profile of 17/U and 19/U netball athletes during and after National Championships' (20180404)

As Secretary of the AIS Ethics Committee, I acknowledge receipt of the Minor Variation Form for the above project on 27/03/19. While submission of the Minor Variation is received before the Ethics due date of 22^{ed} March 2019, it cannot be considered by the AIS Ethics Committee before its scheduled meeting on 9th April 2019.

It is my understanding that due to circumstances beyond your control, the study is being conducted before 9th April, and is therefore not able to be considered by the AIS Ethics Committee. I have forwarded your submission on to the Chair of the Committee who has approved your amendment on the basis that the amendments are minor, and the study was granted full Ethics approval on 10th April 2018.

Please note, that this is not my preferred option, and that in future, the preference would be to have the AIS Ethics Committee review all amendments during scheduled meetings, no matter how minor. However, I accept the unusual circumstances, and the 'low risk' nature of the amendments.

The approval number for this project remains as: 20180404.

It is a requirement of the AIS Ethics Committee that the Principal Researcher (you) advise all researchers involved in the study of Ethics Committee approval and any conditions of that approval. You are also required to advise the Ethics Committee immediately (via the Secretary) of:

Any proposed changes to the research design, Any adverse events that may occur.

Researchers are required to submit annual status report and final reports to the secretary of the AIS Ethics Committee. Details of status report requirements are contained in the 'Guidelines' for ethics submissions.

Please continue to note that the approval for this submission expires on 31st December 2020, after which time an extension will need to be sought.

If you have any questions regarding this matter, please contact me on (02) 6214 1791.

Yours sincerely,

J Ked

Tim Kelly Secretary, AIS Ethics Committee

Appendix 14: Information to Participants for Chapter 6





INFORMATION TO PARTICIPANTS

Research Title: Injury profile of 17/U and 19/U netball athletes during and after National Championships

Principal Researcher:

Erin Smyth (0420512039, erin.smyth@ausport.gov.au)

We would like to invite you to participate in this research project. Choosing not to take part will not disadvantage you in any way. Before you decide whether you want to take part, it is important for you to understand why the research is being done and what your participation will involve. Please take time to read the following information carefully and discuss it with others if you wish. Ask us if there is anything that is not clear or if you would like more information.

Aim:

The aim of this research project is to investigate injury and illness incidence in 17/U and 19/U Netball athletes during and one month following the 2019 National Championships.

Benefits:

This study will provide valuable insights into the injury profile of 17/U and 19/U pre-elite Netball athletes to help guide future injury prevention initiatives.

What is involved?

The study will be conducted over a 7 week period (28 March 2019 - 14 May 2019) and will consist of the following:

Online survey

You will be asked to complete two surveys (Low Energy Availability in Females Questionnaire & Pittsburgh Sleep Quality Index) on the 28th March. You will also be asked to complete the Health Problems Questionnaire at the start and end of the tournament as well as one and four weeks post tournament. It consists of four questions that ask whether you have experienced any health problems in the last week. You will be asked to provide the most accurate responses you can. These surveys will be accessible through the Athlete Management System (AMS).

AMS data entry

Your injury records and netball participation will be recorded on AMS by your team Physiotherapist and Netball Australia selectors. This data will be de-identified to protect your privacy.

Medical Care

Should you become unwell while in Brisbane for the tournament you will have access to the tournament doctor for a free consultation and will receive appropriate medical care. If the presenting complaint is suggestive of a possible infection. The doctor will conduct a history and physical exam and discuss the clinical findings, any relevant pathology and treatment that may be indicated. Your consent will also be requested for the outcome of the medical assessment to be included in the AMS and subsequent research project.

The care provided may include bulk billed pathology, radiological investigations or prescription of appropriate medications under the pharmaceutical benefits scheme. The pathology services will not cost you anything providing you have a Medicare card. However, medications on the pharmaceutical benefits schedule can attract a small fee from a local pharmacy. There will be no fee from the tournament doctor.

Common investigations for infective symptoms include a swab from the back of the nose, collection of sputum) sample, (mucus from a cough), faecal sample, or blood test. You will be free to accept or decline all or part of any investigations offered. The tests if indicated will help to inform what pathogen is causing the infection and direct individual treatment. When collated with other athletes' pathology results can contribute to our understanding of infection transmission within high performing sporting teams in competition.

At the end of the tournament, this will conclude the episode of medical care and a summary of your presentation will be sent to your nominated general practitioner who will then be responsible for continuing care. Your general practitioner will be able to contact the tournament doctor should they have any further questions.

Who are we recruiting?

We are recruiting all netball athletes participating in the 2019 17/U and 19/U National Netball Championships.

Adverse effects and withdrawal:

We do not anticipate any adverse effects and you may withdraw at any time without penalty.

Confidentiality:

The confidentiality of your data will be respected at all times. Only researchers directly involved in this project will see your AMS data which will be kept for minimum of 5 years. Data is stored electronically in the AMS and any hard copies will be stored in a locked filing cabinet. Participants' names will not be associated with any aspect of this study; this will ensure anonymity during research and publication. Data will be collated so that an individual's information will not be able to be identified. The AMS data utilised in this study may also be used in future studies. This research will be used to inform the design of injury and illness prevention programs for netball athletes in the future. Results will be disseminated in the form of publications and be included in Erin Smyth's PhD thesis. Follow up support will be provided by the student researcher Erin Smyth between Monday-Thursday, between the hours of 9am and 5pm.

Ethics approval:

This study has been approved by the Australian Institute of Sport Ethics Committee (Approval Number: 20180404). If you, the participant, have any concerns you can contact the secretary of the AIS Ethics Committee on 02 6214 1577.

Further information:

If you have any questions, or you would like further information regarding the project titled: *Injury profile of 17/U and 19/U netball athletes during and after National Championships*, please contact the Principal Researcher, **Erin Smyth** of the **Australian Institute of Sport** and **University of Canberra: PH: +61 420512039 EMAIL: erin.smyth@ausport.gov.au**

Appendix 15: Information for Parent/Guardian of Minor for Chapter 6





INFORMATION TO PARENT/GUARDIAN OF MINOR (UNDER 18 YEARS) PARTICIPANTS

Research Title: Injury profile of 17/U and 19/U netball athletes during and after National Championships

Principal Researcher:

Erin Smyth (0420512039, erin.smyth@ausport.gov.au)

We would like to invite your child to participate in this research project. Choosing not to take part will not disadvantage your child in any way. Before you decide whether you want your child to take part, it is important for you to understand why the research is being done and what their participation will involve. Please take time to read the following information carefully and discuss it with others if you wish. Ask us if there is anything that is not clear or if you would like more information.

Aim:

The aim of this research project is to investigate injury and illness incidence in 17/U and 19/U Netball athletes during and one month following the 2019 National Championships.

Benefits:

This study will provide valuable insights into the injury profile of 17/U and 19/U pre-elite Netball athletes to help guide future injury prevention initiatives.

What is involved?

The study will be conducted over a 7 week period (28 March 2019 - 14 May 2019) and will consist of the following:

Online surveys

Your child will be asked to complete two surveys (Low Energy Availability in Females Questionnaire & Pittsburgh Sleep Quality Index) on the 28th March. They will also be asked to complete the Health Problems Questionnaire at the start and end of the tournament as well as one and four weeks post tournament. It consists of four questions that ask whether your child has experienced any health problems in the last week. Your child will be asked to provide the most accurate responses that they can. These surveys will be accessible through the Athlete Management System (AMS).

AMS data entry

Your child's injury records and netball participation will be recorded on AMS by their team Physiotherapist and Netball Australia selectors. This data will be de-identified to protect their privacy.

Who are we recruiting?

We are recruiting all netball athletes participating in the 2019 17/U and 19/U National Championships.

Adverse Effects and Withdrawal:

We do not anticipate any adverse effects and your child may withdraw at any time without penalty.

Confidentiality:

The confidentiality of your child's data will be respected at all times. Only researchers directly involved in this project will see their AMS data which will be kept for minimum of 5 years. Data is stored electronically in the AMS and any hard copies will be stored in a locked filing cabinet. Participants' names will not be associated with any aspect of this study; this will ensure anonymity during research and publication. Data will be collated so that an individual's information will not be able to be identified. The AMS data utilised in this study may also be used in future studies. This research will be used to inform the design of injury and illness prevention programs for netball athletes in the future. Results will be disseminated in the form of publications and be included in Erin Smyth's PhD thesis. Follow up support will be provided by the student researcher Erin Smyth between Monday-Thursday, between the hours of 9am and 5pm.

Consent:

Your child will be asked to complete a questionnaire on their smartphone. Prior to this they will be asked to provide your consent, name and contact email. Can you please communicate with your child prior to 28th March 2019 whether you give your consent for them to participate in this research project.

Ethics Approval:

This study has been approved by the Australian Institute of Sport Ethics Committee (Approval Number: 20180404). If you, the participant, have any concerns you can contact the secretary of the AIS Ethics Committee on 02 6214 1577.

Further information:

If you have any questions, or you would like further information regarding the project titled: *Injury profile of 17/U and 19/U netball athletes during and after National Championships*, please contact the Principal Researcher, **Erin Smyth** of the **Australian Institute of Sport** and **University of Canberra: PH: +61 420512039 EMAIL: erin.smyth@ausport.gov.au**

Appendix 16: Informed Consent Form for Chapter 6

Injury profile of 17/U and 19/U netball athletes during and after 2019 National Championships

Principal Investigators: Ms Erin Smyth (PhD Scholar, AIS/UC), Dr Michael Drew (AIS), Prof Gordon Waddington (AIS/UC), Dr Juanita Weissensteiner (AIS), Dr Philip Newman (UC), Dr Laura Juliff (Netball Australia), Ms Alanna Antcliff (Netball Australia)

You are invited to take part in this research project.

Please read the **Participant Information Form** carefully as this will tell you about the research project and explain what is involved. This will help you decide if you want to continue and take part.

Ask questions about anything that you don't understand or want to know more about. Before deciding whether or not to take part, you might want to talk about it with a relative, friend or healthcare worker.

Participation

- Participation in this study is entirely voluntary.
- You're not obliged to participate and if you do, you can withdraw at any time without penalty or prejudice.
- To participate, we would like you to complete the online questionnaires: 'Low Energy Availability in Females Questionnaire', Pittsburgh Sleep Quality Index' and 'Health Problems Questionnaire'.
- Your participation, personal details and results will be strictly confidential and only the Principal Investigators above will have access to the information.

By ticking the 'I ACCEPT' or 'My Parent or Guardian ACCEPTS' option below you are telling us that you or your Parent or Guardian:

- Understand what you/they have read.
- Consent to take part in the research project.
- Consent to participate in the research processes that are described.
- Consent to use your personal and health information as described.
- Understand that you are free to not answer specific items or questions.
- Understand that any data or answers to questions will remain confidential with regard to your identity.
- Certify to the best of your knowledge and belief, you have no physical or mental illness that would increase the risk of participating in this project.
- Are participating in this project of your own free will and have not been coerced in any way to participate

□ I ACCEPT (If you do not accept, please exit the browser)

Minors (Please complete if under 18 years)

□ My Parent or Guardian ACCEPTS (If they do not accept, please exit the browser)

Parent/Guardian full name:

Parent/Guardian email address: _____

Appendix 17: Foot blister advice sheet and pack





Blister Prevention & Care for Athletes

Introduction

One of the most common injuries that athletes sustain that has the potential to lead to loss of time from training and competition are blisters.

They can range from being an uncomfortable nuisance to requiring hospitalisation and the administration of intravenous antibiotics for spreading infection which can cause significant time out of play.

Being aware of how to avoid blisters and being prepared to safely self-mange any blisters immediately they occur can help prevent any complications from affecting your performance.

What are blisters?

Blisters occur from friction or rubbing of the skin causing one layer of skin to separate from another. This creates a pocket which fills with fluid, causing pain and inflammation.

The common types of blisters you will see are:

- Closed
- Open or torn
- Blood blister
- Deep blisters under a callus
- Infected blister

Common reasons that blisters occur in athletes:

- Returning to training or competition after a break
- Increased frequency and intensity of training loads or games such as camps or nationals
- New orthotics or shoes
- Poorly fitting shoes or orthotics
- A build-up of unhelpful callus

Preventing Blisters includes:

- Recognising when there is an increased likelihood of blisters occurring and take preventative action.
- Wearing in new shoes or a change of orthotics slowly by not wearing at every training or game.
- Wearing the right socks, choose socks that will wick moisture away from the foot, keeping it dry.
- Avoiding seams on your socks that can rub.
- Keeping your feet dry; by avoiding wearing damp socks, changing into clean dry socks if you
 have multiple games or training sessions in one day.
- Wash your feet, dry well and air in between sessions.
- Use a powder to promote dryness, Curash powder is a good one it contains cornstarch and zinc and absorbs excess perspiration. Apply to clean and dry skin.

Author: Ruth Fazakerley (AIS Registered Nurse)





- Stop and readjust when you feel a hot spot, this may be readjusting socks/ shoes/orthotic or applying a protective cover such as bandaid or fixomull.
- Check that your shoes are the correct fit, not too tight (causing pressure) or too loose aiding slippage (causing friction).
- Manage callused areas by preventing the build-up of unhelpful callus (see below).
- Cover the problem spots on your feet by protecting those areas that are vulnerable to friction with a protective covering such as Fixomull or Hypafix, adhesive moleskin. *Tip: when using Fixomuli/Hypafix curve the corners to reduce the risk of the material catching and rolling causing another area of pressure*

Callus management

Callus build up occurs on those area of the body that are regularly subjected to friction. These thickened areas of skin protect us against surface friction and help to reduce the incidence of blisters. However there are times when the build-up of callus can do more harm.

A thick, dry and cracked or abnormally shaped callus (one that does not conform to the normal anatomical contour of that part of the body) can transfer pressure and friction to the deeper skin layers causing a deep fluid filled blister. If left untreated the callus itself becomes the blister hood and if torn can result in a deep painful and open blister.

The management of your callus includes the recognition of unhelpful callus formation and its removal.

For netballers the most common areas of callus formation are:

- Along the middle edge of your large toe.
- At the base of your large toe over the ball of your foot.
- Over your heels.

Indications that a callus requires trimming can include:

- A blister is visible under the callus.
- The callus has developed an abnormal shape and hardness sometimes displayed as a palpable ridge.
- The callus is dry and cracking.
- Small dark spots appearing under the callused area which may be petechial haemorrhages, these are tiny areas of bleeding deep in the tissue caused by pressure from the thick callus.
- The callus is thick and yellowing.
- Localised pain at the site of the callus after exercising which may be a burning sensation.

Be responsible for your foot care and manage callused feet before problems occur.

How to removal excess callus:

For the effective safe removal of thickened problematic callus seek the help of a podiatrist or other health professional experienced in callus removal.

They will use a special tool to shave the excess callus to a more normal coverage of thickened skin. Athletes can then assist callus maintenance by the regular use of a foot file to shave the area and prevent reoccurrence of excess callus.





<u>Foot files</u> are like large emery boards although much coarser, they can be purchased for approximately \$8 to \$12 from most chemists and supermarkets that have a foot care section. The Scholl brand of foot file can be purchased from Priceline, but there are several good foot files you can choose from.

<u>Eulactol Scholl heel balm</u> is one of many products on the market that help to soften calluses. It is particularly helpful in preventing the cracking that occurs from the callus becoming dry. Applying the balm at night and wearing socks for several consecutive nights is very effective.

Closed blister management

The usual management of a closed fluid filled blister is to leave it alone and allow the reabsorption of the fluid to occur naturally.

Normally there is no need to puncture the fluid filled blister unless it is large, painful or likely to be further irritated. For elite athletes unless a blister occurs at the same time as a planned break, this option is not considered, the risk of leaving the blister closed and continuing to exercise can lead to a traumatic tearing of the blister hood causing more discomfort and potential for infection.

The preferred option for athletes is to remove the fluid under aseptic conditions in order to prevent further trauma and reduce the incidence of infection.

It is important to have the correct equipment do this.

Open blister management

If a blister hood has torn and the fluid has escaped, the remaining skin from the blister hood should be removed carefully and betadine or another antiseptic applied to prevent infection. The open blister should be covered with a suitable protective dressing to provide comfort and avoid contamination of the wound.

- An **adhesive foam dressing** (eg Allevyn gentle border, smith&nephew) will provide comfort from the padding and absorb discharge.
- A blister plaster will aid healing by providing a moist wound environment and protect against infection.



Blister Plasters are available in Pharmacy's and some supermarkets

A donut either of silicone gel or foam can be placed over the dressing to remove direct pressure on the healing wound and reduce discomfort while continuing activity.

Analgesia can be taken to relieve pain, and ice applied to reduce swelling and discomfort.





You should **not use shared pools** such as recovery or swimming pools when you have a healing open blister.

Signs & Symptoms of infection

The most common complication of blisters is infection, and for athletes this can develop quickly and requires prompt attention. Continuing to exercise with an untreated infected blister, can lead to the infection spreading rapidly beyond the blister.

If you experience any of the following when you have a blister seek medical advice immediately.

- Pain which can be throbbing and keep you awake at night.
- Pus or discharge from the wound.
- Spreading redness around the blister that may track away from the blisters edge.
- Area around the blister feels warm or hot to touch.
- Feeling unwell.
- Having a raised temperature.
- An elevated resting pulse.
- · Discomfort in the groin on the affected side.

By seeking early diagnosis and treatment you can avoid unnecessary removal from play or training. Treatment may include the application of local antimicrobials and sometimes oral antibiotics to eliminate the bacteria, in addition a wound swab may be taken to discover the particular organism causing the infection. Dressings to keep the area protected and covered will be used.



Blisters commonly occur at the edge of a thickened callus. This open blister has spreading erythema (redness) which is tracking away from the blisters edge, this indicates infection which requires medical intervention.

Marking the edge of the redness can help in assessing whether the infection is responding to treatment.





If you self-manage a blister while on camp always discuss this with your team support, who can provide any further assistance if needed.

These items should be included in a Blister Care Pack to help you to pierce a closed blister safely.

- Alcohol wipes
- Betadine wipes
- Skin prep wipes
- Gauze swabs
- Splinter probes (used instead of sterile needles to comply with no needle policy for athletes)
- ✓ Assorted Band-Aids
- Plastic waste bag

How to use the Blister Pack to pierce a closed blister:

To pierce a closed blister safely to reduce the incidence of infection follow these instructions:

- Wash your hands.
- Clean the area well with the alcohol swab.
- ✓ Using the splinter probe pierce the blister hood several times to allow the evacuation of fluid from the blister this will aid continued drainage.
- Gently squeeze out the blister fluid and use the gauze remove the fluid.
- Thoroughly apply betadine (if no allergy to iodine)
- ✓ Cover with Band-Aid, secure by using skin prep to surrounding intact skin to aid adherence and cover with fixomull tape
- Carefully recap the splinter probe and dispose of safely.
- Wash hands & dispose of used supplies in plastic bag provided

Use each splinter probe for a single blister treatment only

